Examining the impacts of projected precipitation changes on sugar beet yield in Eastern England

A thesis submitted for the degree of Doctor of Philosophy

By

Stanley Ob Joseph

College of Health and Life Sciences Brunel University London

February 2018

Abstract

Projected increasing temperatures and reduced summer precipitation in the UK raises questions about the sustainability of aspects of the agriculture industry and food security. This study investigates the potential impact of precipitation changes on sugar beet yield in Eastern England. Observations of precipitation was examined for the period 1971-2000 and the expected changes in precipitation were investigated using seven CMIP5 climate models for the historical phase (1971-2000) and RCP45 and RCP85 future scenarios (2021-2050). Three out of the seven models were found to show good agreement with observations but the MOHC ensemble mean was the closest to the observed means and was used for further precipitation analyses. Statistical analysis of the future precipitation changes were performed using the Met Office Hadley Centre (MOHC) model focused on changes between the historical phase and RCP45 and RCP85. Results showed a consistent and significant reduction in annual May-October precipitation under future scenarios.

The study then investigated the impact of reduced future precipitation changes on sugar beet yield by applying controlled watering regimes informed by the CMIP5 projections to sugar beet plants in a greenhouse experiment over two seasons – the use of CMIP5 projections in this way is a first. In the first experiment carried out in 2014, a climatological watering regime (i.e. where the total seasonal rainfall for the different scenarios was applied in equal and regular watering events) was applied to the plants, which meant a 16% reduction in precipitation in the "future" category relative to a "control" category. Analysis of the yields indicated a statistically significant reduction in mean tuber wet mass: mean of 360g for the control and 319g for the future (p-value 0.03). This implies a potential yield reduction of 11% by 2050. In the second experiment carried out in 2015, a "realistic distribution" watering regime (i.e. where the total seasonal rainfall events in the different categories), this meant a reduction in precipitation in the months of June (-15.6%), July (-7.7%) and August (-3.7%). This resulted in statistically significant reduction in mean tuber wet mass between control (153g) and RCP85 (113g) with a p-value of 0.01. This implies a reduction of 26% in future yields under RCP85 by 2050.

Results in this thesis further show how changes and variation in precipitation are intertwined with changes in soil moisture and yield of sugar beet plants. The findings will enable UK sugar beet farmers to identify potential areas of challenges in order to adapt their management practices to ensure maximum crop yield in future growing seasons. Moreover, from a global perspective, the findings here are also broadly applicable to a variety of agricultural crops in different parts of the world, where changes in yield may have important consequences to food security and food prices.

Declaration

I can confirm that this is my own work and the use of all materials from other sources has been properly and fully acknowledged.

Stanley Ob Joseph

Table of contents

Abstr	act		ii
Decla	ration	i	ii
Table	e of cor	itentsi	V
List o	of figur	'es	X
List o	of table	SX	vi
Ackn	owledg	gementsxi	X
Chap	ter 1 -	Introduction	1
1.1	Cha	nges in precipitation: Reasons for concern?	1
1.2	Var	iations in Eastern England and UK precipitation	3
1.3	Clir	nate models simulation of precipitation	6
1.4	Imp	acts of precipitation changes on agriculture	7
1.5	Sug	ar beet production in the UK1	0
1.6	Ain	ns and scope 1	2
1.7	Cor	ceptual model of the research study 1	3
1.8	The	sis outline	4
Chap	ter 2 –	Literature review1	6
2.1	Intr	oduction1	6
2.2	Clir	nate change impacts and precipitation variability1	8
2	.2.1	Influence of global warming on precipitation variability1	.8
2.3	Pas	t and present climatic changes across the UK2	1
2	.3.1	Variations and trends in UK precipitation	1
2	.3.2	Variations and trends in UK temperature	4
2.4	Mo	dels simulations of past and projected precipitations changes2	5
2	.4.1	Introduction	5
2	.4.2	Model simulation of observed changes in the climate system	5
2	.4.3	Climate model simulation of climate and precipitation variability2	:9

2.4	4.4	Climate model evaluation of projected precipitation	31
2.4	4.5	Climate model simulation of past and future UK precipitation	34
2.5	Cl	imate change and agriculture	
2.5	5.1	Introduction	
2.5	5.2	Climate change impacts on agriculture	
2.5	5.3	Climate change: Implications for food security	41
2.5	5.4	Future impacts of precipitation changes on UK agriculture	44
2.5	5.5	Impacts of climate and precipitation on UK sugar beet production	46
2.6	Re	search gaps and thesis objectives	
Chapte	er 3:	Methods, Data and Materials	52
3.1	In	troduction	
3.2	Pro	ecipitation observations	53
3.2	2.1	Weather stations observations dataset	53
3.3	Hi	storical CMIP5 model precipitation data	55
3.4	0	verview of climate models	57
3.4	4.1	Model criteria	58
3.5	CN	AIP5 precipitation projections	59
3.5	5.1	Introduction	59
3.5	5.2	Climate change and greenhouse gas emissions scenarios	60
3.5	5.3	Projections of changes in future precipitation	61
3.6	Ca	lculation of watering regimes	
3.6	5.1	Introduction	62
3.6	5.2	Calculation of the climatological watering regimes	62
3.6	5.3	Calculation of realistic precipitation distribution of events and sizes	63
3.7	Me	ethods of statistical analysis	65
3.7	7.1	Independent student's t-test	65
3.7	7.2	Pearson's correlation coefficient (CC)	66
3.7	7.3	Analysis of variance (ANOVA) and the Tukey post hoc test	66

	3 The	greenhouse experiments	7
	3.8.1	Description of the greenhouse6	7
	3.8.2	Selection of seed	8
	3.8.3	Compost	0
	3.8.4	Plant pots	0
	3.8.5	Sowing dates7	1
	3.8.6	The plant experiment7	2
	3.8.7	Method of plants watering (Irrigation)7	3
	3.8.8	Plant parameter measurements	6
3.9	We	eding and pest control7	9
	3.9.1	Weeds	9
	3.9.2	Powdery Mildew Disease (PMD)8	0
	3.9.3	Armyworms	1
Cha	pter 4 –	Results: precipitation analysis8	2
4.1	l Intr	oduction	2
4.2	2 Ana	lysis of weather station data	2
			_
	4.2.1	Weather stations May-October annual mean precipitation analysis	2
	4.2.1 4.2.2	Weather stations May-October annual mean precipitation analysis	2 2 6
	4.2.1 4.2.2 4.2.3	Weather stations May-October annual mean precipitation analysis	2 6 9
	4.2.1 4.2.2 4.2.3 4.2.4	Weather stations May-October annual mean precipitation analysis	2 6 9 3
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9	2 6 9 3
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 	Weather stations May-October annual mean precipitation analysis	2 6 9 3 8 8
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Proj 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9 Comparison of weather station data and the "historical" phase of the CMIP5 9 jections of future precipitations 10	2 6 9 3 8 8 3
4.2	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Pro 4.4.1 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9 Comparison of weather station data and the "historical" phase of the CMIP5 9 jections of future precipitations 10 Analysis of the May-October daily mean precipitations under the "historical" and	2 6 9 3 8 8 3
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Proj 4.4.1 "future" 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9 Comparison of weather station data and the "historical" phase of the CMIP5 9 jections of future precipitations 10 Analysis of the May-October daily mean precipitations under the "historical" and phases of the CMIP5 models 10	2 6 9 3 8 8 3 3
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Pro 4.4.1 "future" 4.4.2 "future" 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9 Comparison of weather station data and the "historical" phase of the CMIP5 9 jections of future precipitations 10 Analysis of the May-October daily mean precipitations under the "historical" and phases of the CMIP5 models 10 Changes in the May-October daily mean precipitations under the "historical" and phases of the CMIP5 models 10	2 2 6 9 9 3 8 8 3 3 3 5 F
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Pro 4.4.1 "future" 4.4.2 "future" 4.4.2 	Weather stations May-October annual mean precipitation analysis 8 Weather stations total daily May-October wet days precipitation analysis 8 Weather stations monthly May-October daily precipitation 8 Summer (June, July, August; JJA precipitation) 9 sent climate 9 Comparison of weather station data and the "historical" phase of the CMIP5 9 jections of future precipitations 10 Analysis of the May-October daily mean precipitations under the "historical" and phases of the CMIP5 models 10 Changes in the May-October daily mean precipitations under the "historical" and phases (RCP45 and RCP85) of the CMIP5 models 10	2 2 6 9 3 8 8 3 3 3 5 5
4.3	 4.2.1 4.2.2 4.2.3 4.2.4 3 Pres 4.3.1 4 Pro 4.4.1 "future" 4.4.2 "future" 4.4.3 RCP85 so 	Weather stations May-October annual mean precipitation analysis	2 2 6 9 3 8 8 3 3 3 5 8 8

4.4.4		Changes in total May-October wet day precipitation under the "historical" phase,	
"R	CP45"	and RCP85 scenarios	111
4.4.5		CMIP5 models individual month's May-October daily precipitation	. 113
4.4.6		Changes in individual month's May-October daily precipitation	116
4.4	4.7	Summer (June, July and August; JJA) precipitation from the CMIP5 models	118
4.5	Eve	nt size distribution of monthly May-October precipitation	120
4.	5.1	Analysis of monthly May-October wet day distribution	120
4.	5.2	Changes in monthly May-October distribution of event sizes	122
4.6	Wat	tering regimes (Irrigation)	124
4.	6.1	Introduction	124
4.	6.2	Results of the climatological watering regimes calculations	124
4.	6.3	Results of the realistic distribution of monthly watering regimes calculations	127
Chapt	er 5 –	The Greenhouse plant experiments	.129
5.1	Intro	oduction	129
5.2	Clir	natological watering regime measurements	130
5.	2.1	Above-ground measurements	130
5.3	Belo	ow-ground measurements for the climatological experiment	133
5.	3.1	Wet weight of yield under the climatological watering regimes	. 133
5.	3.2	Dry weight of yield under the climatological watering regimes	. 135
5.	3.3	Soil moisture	137
5.	3.4	Correlation between soil moisture and wet weight of yield	139
5.	3.5	Correlation between soil moisture and dry weight of yield	141
5.4	Rea	listic distribution of precipitation watering regimes parameters measurement	142
5.4	4.1	Above-ground measurements	142
5.5	Belo	ow-ground measurements	145
5.	5.1	Wet weight of yield under the distribution of watering events and sizes	. 145
5.	5.2	Dry weight of yield under the distribution of watering events and sizes	149
5.	5.3	Soil moisture	152

5.5.4		Correlation between soil moisture and wet weight of yield	155
5.5.5		Correlation between soil moisture and dry weight of yield	
5.6 Hea		athrow weather station	157
5.6.1		Temperature	157
5.7	7 Su	gar Content Analysis	161
	5.7.1	Total sugar content	161
Cha	pter 6	- Discussion	
6.1	l Int	roduction	164
6.2	2 Ea	stern England climate	
	6.2.1	Weather stations observations	165
	6.2.2	CMIP5 climate model simulation of precipitation under the historical phase	
6.3	3 Pr	ojected changes in future precipitation	167
	6.3.1	Changes in past and future Eastern England precipitation	167
	6.3.2	Analysis of future total May-October wet day precipitation	
	6.3.3	Monthly distribution of precipitation events and sizes.	
	6.3.4	Model simulation of precipitation changes	
6.4	4 Gr	eenhouse Experiment	175
	6.4.1	Sowing and emergence	
	6.4.2	Event based impacts	177
	6.4.4	Above-ground parameter measurements	181
	6.4.5	Below-ground parameter measurements: Yield	181
	6.4.6	Below-ground parameter measurements: Soil moisture	
6.5	5 Ef	fects of climate change	
	6.5.1	Possible implications of future reduction in precipitation	
	6.5.2	Possible implications of reduced precipitation on sugar beet plants	
	6.5.3	Implications for food security	
Cha	pter 7 -	– Conclusions	192
7.1	l Su	mmary	

Append	lix			
References				
Glossar	ry			
7.5	Suggestions and future studies			
7.4	Impacts on soil moisture			
7.3	Dry weight yield			
7.2	Wet weight yield			

List of figures

Figure 1. 1: Average summer UK rainfall for the period 1981-2010. Source: Met	Office 2016
	4
Figure 1. 2: Annual UK sugar beet yield and annual rainfall. Data: (DEFRA 2015	; Met Office
(n/d))	11
Figure 1. 3: A conceptual model of the study illustrating the pathways that was	of particular
interest in this thesis (i.e. factors that will ultimately show the impact of changes in	precipitation
on sugar beet yield.	14

2. 1: Observed, individual models and CMIP5 model simulated time series of the anomalies and global mean surface temperature from 1961-1990 reference period. The reference period 1961-1990 is indicated by the yellow shading, vertical dash grey lines represent major volcanic eruptions, the single CMIP5 models are represented by thin lines and the multi model mean are represented by the thick red line. The different observations are in thick black lines. Source: (IPCC 2014a; IPCC 2013b)......20 2. 2: Mean seasonal cycle of (a) temperature (°C) and (b) precipitation (mm day-1). "The average is taken over land areas within the indicated regions, and over the period 1980–1999. The red line is the average over 45 CMIP5 models; the blue line is the average over 22 CMIP3 models. The standard deviation of the respective data set is indicated with shading. The different line styles in black refer to observational and reanalysis data: Climatic Research Unit (CRU) TS3.10, ECMWF 40-year reanalysis (ERA40) and ERA-Interim for temperature; CRU TS3.10.1, Global Precipitation Climatology Project (GPCP), and CPC Merged Analysis of Precipitation (CMAP) for precipitation. Note the different axis-ranges for some of the subplots. The 15 regions shown are: Western North America (WNA), Eastern North America (ENA), Central America (CAM), Tropical South America (TSA), Southern South America (SSA), Europe and Mediterranean (EUM), North Africa (NAF), Central Africa (CAF), South Africa (SAF), North Asia (NAS), Central Asia (CAS), East Asia (EAS), South Asia (SAS), Southeast Asia (SEA) and Australia (AUS)". Source: (IPCC, 2013)......28

Figure 3. 1: Map of the study area showing the locations of the weather stations examined in

the analysis. The dashed line indicates the area covered by MOHC HadGEM2-ES model grid
cell used here. (See 3.2.1 and Table 3.1)
Figure 3. 2: Panoramic view of plants inside the greenhouse67
Figure 3. 3: Picture of sugar beet seeds used in this study69
Figure 3. 4: Picture of the plant pots used for the experiments with drainage holes at the bottom
to drain off excess water from the pots71
Figure 3. 5: Illustration of how plants in the control (C) and future (F) categories were arranged
inside the greenhouse in the first season of the study73
Figure 3. 6: Illustration of how plants in the control (C), RCP45 (M) and RCP85 (H) categories
were arranged on the tables inside the greenhouse in the second season of the study
Figure 3. 7: The method used in measuring plants height and growth ratio index. A tape rule is
placed at the crown of the plant to the highest tip of the tallest stem. The highest stem
measurement is divided by the number of leaves on the plant to get an indication of the plants
growth ratio77
Figure 3. 8: Method of manual soil moisture reading

Figure 4. 1: Monthly mean rainfall from six stations in three different regions of the UK. Two stations from the area of interest in Eastern England (a, b), two stations from the South West of England (c, f) and two stations from northern Scotland (e, f). Source: Met Office (2016).83 Figure 4. 2: Observation plot of the May-October annual mean precipitation data from the six Figure 4. 3: Box-and-whisker plot displaying the characteristics of the May-October precipitation data for 1971-2000 from the weather stations observations. The solid thick black line within the box represents the median (2nd quartile) of the distribution. The extremes of the box represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and Figure 4. 4: Distribution of the total May-October precipitation from each of the stations for Figure 4. 5: Distribution of the monthly May-October daily precipitation of the growing season Figure 4. 6: Observation plots of the monthly May-October mean precipitation of the growing Figure 4. 7: Box and whisker plot displaying the characteristics of the monthly May-October precipitation data from the weather station observations for the period 1971-2000. The boxplot details are the same as in Figure 4.3......92 Figure 4. 8: Summer JJA precipitation of each of the stations for the period 1971-2000......95 Figure 4. 9: Summer JJA mean precipitation for each month for the stations for the stations Figure 4. 10: JJA monthly May-October precipitation data from the weather station observations for the period 1971-2000. The boxplot details are the same as in Figure 4.3. ...98 Figure 4. 11: Box-and-whisker plot displaying the characteristics of the station observations and the historical phase of the CMIP5 models from May-October for 1971-2000. The solid thick blue line indicates the median of the distribution and the solid thick black line within the box represents the median (2nd quartile) of the distribution. The extremes of the box represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and highest values of the distribution. The blue line running across the plot shows the combined mean precipitation Figure 4. 12: Histogram distribution plots of the May-October daily mean precipitation for 1971-2000 from the historical phase of the daily weather stations observations and the CMIP5 Figure 4. 13: Boxplot of the May to October daily mean precipitation data from the a) historical phase (1971-2000), b) RCP45 (2021-2050) and c) RCP85 (2021-2050) output from the CCCma, MOHC and EC-Earth climate models. The boxplot details are the same as for Figure Figure 4. 14: Time series plot from CCCma, EC-Earth and the MOHC HadGEM2-ES climate models illustrating the trend of the May to October precipitation for the historical (1971-2000), RCP45 and RCP85 (2021-2050) scenarios. The blue horizontal line running across the plot represents the stations mean line as per Figure 4.9 plotted against the CMIP5 models......107 Figure 4. 15: May-October mean daily precipitation for the historical phase, RCP45 and RCP85 from the CCCma, EC-Earth and MOHC climate models for the periods 1971-2000 and 2021-Figure 4. 16: Distribution of the total May-October precipitation for the historical period (1971-2000) and RCP45 and RCP85 scenarios (2021-2050) from the CCCma, EC-Earth and MOHC Figure 4. 17: Boxplots showing the distribution of monthly precipitation from the models under the historical phase (1971-2000) and future scenarios (2021-2050). The boxplot details are the Figure 4. 18: Distribution of monthly JJA daily mean precipitation for the historical phase (1971-2000), RCP45 and RCP85 (2021-2050). The boxplot shows the 95% confidence interval for the JJA precipitation from the CCCma, EC-Earth and MOHC models......119 Figure 4. 19: Variations in the monthly distribution of precipitation events and sizes under the historical phase, RCP45 and RCP85 from the MOHC (HadGEM2-ES) model......123 Figure 4. 20: Result of the daily May-October data from the historical (1971-2000), RCP45 (2021-2050) and RCP85 (2021-2050) from the MOHC model. The boxplot details are the same Figure 4. 21: Tukey multiple comparison of means test for the daily May-October precipitation (mm per day) for the historical, RCP45 and RCP85.127

Figure 5. 1: Boxplot of the climatological watering regime data from the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extreme of the plots represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate

the lowest and highest values in the data
Figure 5. 2: Results of the tuber wet mass data analysis. Boxplot showing the tuber wet mass
data from the control and future categories. The boxplot details are the same as for Figure 5.1.
Figure 5. 3: Histogram showing the distribution of the tuber wet mass data for the control and
future categories
Figure 5. 4: Results of the tuber dry mass data analysis. Boxplot showing the tuber dry mass
data from the control and future categories. The boxplot details are the same as for Figure 5.1.
Figure 5. 5: Histogram showing the distribution of the tuber dry mass data for the control and
future categories
Figure 5. 6: Results of the soil moisture data analysis. The boxplot shows soil moisture data
from the control and future categories
Figure 5. 7: Line graph showing the mean monthly soil moisture measurement for the control
(blue line) and future (maroon line) categories
Figure 5. 8: Scatter plot showing the wet mass for individual tubers from the control (red
circles; solid line) and future (blue triangle; solid line) categories plotted against the mean soil
moisture data for each replicate
Figure 5. 9: Scatter plot showing the dry mass for individual tubers from the control (red circles;
solid line) and future (light green triangle; solid line) categories plotted against the mean soil
moisture data for each replicate141
Figure 5. 10: Boxplot of the Realistic precipitation distribution watering regime results from
the control and future experiments. The thick black line represents the median (2nd quartile)
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3^{rd} quartiles (top).
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data
the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3 rd quartiles (top). The whiskers indicate the lowest and highest values in the data

as for Figure 5.1
Figure 5. 15: Histogram showing the distribution of the tuber dry masscomplete data for the
control, RCP45 and RCP85 categories151
Figure 5. 16: Results of the Tukey test showing that the mean dry weight between RCP85 and
the Control experiment is significantly different from the others151
Figure 5. 17: Results of the soil moisture data analysis. Boxplot showing soil moisture
distribution from the control, RCP45 and RCP85 categories154
Figure 5. 18: Line graph showing the mean monthly soil moisture measurement for the control
(blue line), RCP45 (maroon line) and RCP85 (green line) categories
Figure 5. 19: Results of the Tukey test showing the differences in mean soil moisture between
the three experiments155
Figure 5. 20: Scatter plot showing the wet mass for individual tubers from the control (red
circles; solid line), RCP45 (green triangle; solid line) and RCP85 (blue squares; solid line)
categories plotted against the mean soil moisture data for each replicate156
Figure 5. 21: Scatter plot showing the dry mass for individual tubers from the control (red
circles; solid line), RCP45 (green triangle; solid line) and RCP85 (blue squares; solid line)
categories plotted against the mean soil moisture data for each replicate
Figure 5. 22: Daily mean temperature distribution for 2014 (orange colour) and 2015 (light
blue colour) from Heathrow weather station
Figure 5. 23: Monthly mean temperature distribution for 2014 (orange colour) and 2015 (light
blue colour) from Heathrow weather station

Figure 7	7. 1: Station	ns May-Octo	ber daily	/ mean	precipitation	(mm	per da	ay) from	1971-2000
							•••••		

List of tables

Table 3. 1: Details of the weather stations examined showing the length of their records	55
Table 3. 2: Details of the CMIP5 models examined.	56
Table 3. 3: An overview of the two Representative Concentration Pathways (RCPs) used	in
this thesis (Moss et al. 2010; Meinshausen et al. 2011)	61
Table 3. 4: Shows a comparison of the SY Muse with other sugar beet varieties. Source: BBH	RO
(2014)	69

Table 4. 1: Weather stations analysis of the May-October daily mean precipitation from 1971-
2000
Table 4. 2: Analysis of the stations monthly May-October daily precipitation from 1971-2000
from the five stations with similar characteristics
Table 4. 3: Analysis of the total June, July and August (JJA) mean precipitation from each of
the weather stations for the period (1971-2000)
Table 4. 4: Analysis of monthly June, July, August (JJA) means precipitation for the period
1971-2000
Table 4. 5: May-October annual means precipitation analysis from the weather stations and
CMIP5 data for the period (1971-2000)
Table 4. 6: Ensemble means of the May-October mean precipitation from the best three
performing models (CCCma, MOHC and EC-Earth)104
Table 4. 7: Result of the CMIP5 climate models changes in May-October daily mean
precipitation for the sugar beet growing season
Table 4. 8: Mean analysis of total May-October distribution of daily precipitation between the
two different time frames (historical phase 1971-2000 and the RCP45 and RCP85 scenarios
2021-2050)
Table 4. 9: Result of the CMIP5 climate models mean wet days for the growing season total
May-October daily precipitation. Changes between the historical phase (1971-2000) and future
scenarios under RCP45 and RCP85 (2021-2050) are presented
Table 4. 10: Projected changes in the number of wet day precipitation from the models for the
total May-October daily mean precipitation between the historical phase and future scenarios
(RCP45 and RCP85)112
Table 4. 11: Models analysis of monthly May-October precipitation under the historical phase

(1971-2000) and future scenarios, RCP45 and RCP85 (2021-2050)115
Table 4. 12: Analysis of changes in Monthly May-October daily mean precipitation between
the historical phase (1971-2000), RCP45 and RCP85 (2021-2050) from the CMIP5 models.
Table 4. 13: JJA monthly mean precipitation analysis for the historical phase and the RCP45
and RCP85 scenarios
Table 4. 14: Comparison of monthly May-October event size distribution and non-zero
precipitation days (wet days) between the historical phase (1971-2000RCP45 and RCP85 from
the MOHC model122

Table 5. 1: Mean and Std. of the final measurements of non-destructive measurements	nents from
the control and future seasonal watering regimes	132
Table 5. 2: June-October soil moisture data of the growing season for the control	and future
watering categories calculated from the measurements taken every two weeks	138
Table 5. 3: Mean monthly soil moisture data for the control and future experiments	during the
growing season	
Table 5. 4: Mean and Std of the final non-destructive measurements from the cont	rol, RCP45
and RCP85 Realistic precipitation distribution watering regimes	144
Table 5. 5: Results of the Tukey analysis of contrasts means test from the parameter	s measured
for the control, RCP45 and RCP85 experiments also illustrated in Figure 5.9	144
Table 5. 6: Results of mean tuber wet weight for the growing season from the Cont	rol, RCP45
and RCP85 experiments	146
Table 5. 7: Tuber wet weight data analyses showing results of the Tukey multiple of	comparison
of means test for the control, RCP45, and RCP85	147
Table 5. 8: Results of mean tuber dry weight for the growing season from the Cont	rol, RCP45
and RCP85 experiments	150
Table 5. 9: Tuber dry weight data analyses showing results of the Tukey multiple of	comparison
of means test for the control, RCP45, and RCP85	150
Table 5. 10: Result of the mean monthly (July-October) soil moisture data collected	during the
growing season for the control, RCP45 and RCP85 experiments	153
Table 5. 11: Soil moisture data analyses showing results of the Tukey multiple con	nparison of
means test for the control, RCP45, and RCP85 watering regimes	153
Table 5. 12: Result of the mean monthly soil moisture data collected during the grow	ving season
for the control, RCP45 and RCP85 experiments.	153

Acknowledgements

Firstly, I am most grateful to the Most High God for giving me the grace and strength to complete this work in spite of numerous challenges. Indeed, without the grace of God in my life, I doubt if this work would have come to light. For this I am thankful.

There are so many people to thank for their support into my research work in the pursuit of my doctorate. I owe so much thanks to Dr Abdul Chaudhary (AJ) for his constant support from day one and agreeing to supervise me when I was at a cross road. I thank you so much for helping me cross the finish line. A big thank you to Dr Ed Routledge for his role and support in seeing that what would have been a problem was resolved and made sure I crossed the finish line. I am equally thankful to Dr Andrew Russell, my former supervisor for taking me on and assisting me with advice and suggestions on how to improve my work. I owe so much of my development as a researcher to you, Andy. Thank you also for taking the time out of your busy work schedule to provide me with valuable comments on this work. Thanks are also extended to members of my annual review panel committee comprising of Dr Mark Scrimshaw, Dr Ed Routledge and Dr Elisabete Silva. Their official recommendations and personal advice helped redirect my steps in the right direction when I needed help most.

I gratefully acknowledge the support of Mr Paul Varney (Head of Field Sales) Sygenta UK for providing me with free sugar beet seeds for the two seasons of the experiment. Special thanks to Mr Simon Carroll (Head of Grounds and Biodiversity), Brunel University London, and his staff for making it possible for me to use the greenhouse for my plant experiment. I acknowledge the contribution of my uncle, Barrister Chidi Egwuenu whose financial support helped keep me in school to accomplish this goal. I also like to thank Professor Susan Joblin for providing us with the funds to carry out the experiments and Dr Rob Francis for his input in the greenhouse experiment.

Finally, I want to thank my children and wife for their sacrifices, encouragement and support. Emmanuel, Brooklyn, Chelsea, Charles and Stacy Joseph, this is for you and thanks for believing in me. God win!

Chapter 1 - Introduction

1.1 Changes in precipitation: Reasons for concern?

Globally, climate change and variability are having serious impacts on hydrological processes via changes in precipitation with sometimes severe consequences for society. Changes in precipitation characteristics and delivery are expected to occur under a warmer climate and the contribution from the concentrations of greenhouse gases in the atmosphere is attributed to be the major cause of the change: the Intergovernmental Panel on Climate Change (IPCC) Fifth Assessment Report (AR5) provides evidence that "warming of the climate system is unequivocal and that human influence have been detected in the warming of the atmosphere" (IPCC 2013a). The warming causes temperature to increase, changes in precipitation patterns, increase in sea level rise and a change in the frequency of extreme weather events. These impacts are evident from increasing surface temperature, which causes an increase in evapotranspiration and, in turn, causes changes to precipitation patterns thereby altering hydrological processes (Cramer et al. 2014; IPCC 2013a; Meehl et al. 2007; Huntington 2006; Meehl et al. 2005) and further impacting on water resources in terms of quantity and quality (Cramer et al. 2014; IPCC 2013a).

Furthermore, the risks associated with climate change are becoming more visible through the movement of climatic variables such as precipitation and temperature with varying degree of impacts in different regions. Observations show that certain regions and areas have been identified as more vulnerable to the adverse effects of climatic changes, such as changes in precipitation patterns. Most Earth System Models (ESM's) predict an increase in precipitation in the mid to high latitudes but a decrease in the sub-tropics (IPCC 2013b). In continental Europe, Christensen et al. (2007) reported a decrease in annual mean precipitation for southern Europe and an increase in northern Europe. Their analysis also showed increases in winter precipitation in central and northern Europe, and decreases in the summer over central and southern Europe. Several other studies on the different aspects of precipitation have consistently agreed on an increase in global mean precipitation with an expected increase in variation of precipitation events resulting from warming of the atmosphere (IPCC AR5, Stocker et al. 2013). Studies on

the theoretical aspects of precipitation conducted by Trenberth (2011) and Trenberth & Josey (2007), observational studies by Halmstad et al. (2013) and Min et al. (2011), and simulation studies using climate models by Pincus et al. (2008) and Phillips & Gleckler (2006) all agreed on an increasing global mean precipitation.

In addition to increase in precipitation, changes in the character of local and regional precipitations are emerging and depends largely on the variation of atmospheric circulation because a shift in the direction of the wind causes changes in the delivery of precipitation which makes some regions to be wet and others dry (Trenberth 2011). These changes vary from season to season, and from one year to another which suggests that some years will have more or less precipitation than others. As a result, years with reduced mean precipitation can increase the risk of drought whereas; years with increased mean precipitation can equally increase the risk of floods. Irregularity in precipitation appears to be the major source of variations in the distribution of precipitation events in most regions because precipitation characteristics are highly variable and unpredictable. This causes intensification of hydrological cycle with expected increases in the frequency of dry and wet spells, as well as increases in the frequency and magnitude of precipitation events with significant consequences for society (Cramer et al. 2014; IPCC 2013; Huntington 2006).

Emori & Brown (2005) reported in an earlier study that the effects of global warming on precipitation frequency and intensity on a global scale does not reflect the impact on regional and/or local scales because of thermo-dynamic influences affecting precipitation unevenly across the world. Emory and Brown indicated that secondary influences such as atmospheric moisture content are the more likely to dominate and influence precipitation amount and intensity on local scales as a result of warmer climate. Furthermore, the warming of the atmosphere, climate and weather variability are widely projected to continue for the rest of the century (IPCC 2012) and will significantly impact on hydrological processes (Meehl et al. 2007; Meehl et al. 2005) which typically leads to flood or droughts. Floods are generally associated with extreme precipitation while droughts are associated with deficient precipitation and high temperatures which often leads to drying of the soil. Flood incidents occur locally while droughts develop over a period of months or years. Both type of events can be mitigated – floods with good drainage system and drought with the use of irrigation (Trenberth 2011). This thesis is not about extreme events, but,

it is important to point out that both floods and droughts are a manifestation of changes in precipitation and variability; hence their occurrence is also of great societal concern.

In response to the problem of variability in precipitation patterns, Brunet & Jones (2011) called for digitalisation of long-term climate data all over the world. They identified the lack of longterm high-quality data in some parts of the world as a condition hampering a more robust assessment of the climate. The position of Brunet and Jones supports earlier issues identified by Trenberth et al. (2007). Since then, several studies and reports have recommended several approaches to climate change mitigation such as reducing greenhouse gas emissions in order to keep global mean temperature increase less than 2°C above pre-industrial levels (IPCC 2014b), which will be helpful in limiting the impacts of global warming (Allison et al. 2009) and Successive IPCC reports from AR4 to AR5 (IPCC 2014b; IPCC 2012; IPCC 2007). The most recent global agreement in this direction is the United Nations Climate Change Agreement in Paris in 2015 designed to increase efforts to further limit temperature increase to 1.5°C above pre-industrial levels (UNFCCC 2015).

Considering the potential impacts of this problem, it is important to examine and understand how climatic changes will affect precipitation delivery in Eastern England to enable reliable projections of future changes in the region. Assessing precipitation characteristics on daily, monthly or seasonal time scales, as well as the capacity to predict precipitation changes and evaluate its impacts on relevant sectors is very important in developing adaptation strategies (Allan & Soden 2008) and one of the principal aims of this study.

1.2 Variations in Eastern England and UK precipitation

In Eastern England, the climate is mild and dry according to the Met Office (2016) and is influenced by the high grounds in the west of the UK including the mountains in Wales, North-West England and Scotland, where the mountains are more than 1300m high. These high grounds leave the east of the UK in a "rain shadow" from the prevailing westerly winds resulting in a distinct pattern of west-east average precipitation amounts (Met Office 2016). The orographic influences in the west means that the highest amount of precipitation (which by implication means the wettest places) are found in the north-west England, Scotland and Wales with an

annual mean precipitation of over 3000 mm in some places (Met Office 2016). Typical annual mean precipitation of different locations from north to south of the UK ranges from 1170 mm in Stornoway, 1050 mm in Glasgow, 810 mm in Manchester to 760 mm in Exeter. In the east, Edinburgh and London have 670 mm and 610 mm annual mean precipitation respectively. East Anglia in contrast has the smallest amount of precipitation in the UK with a total annual precipitation of about 500 mm in some places but has more precipitation in the autumn and winter periods than the summer (Met Office 2016).



Figure 1. 1: Average summer UK rainfall for the period 1981-2010. Source: Met Office 2016

Precipitation in Eastern England and the UK in general is highly variable from year to year over decades, and changes in the amount, intensity and frequency often affecting the environment and society. This variability further exacerbates the multiple stresses already present in the region, making the region more sensitive to possible future changes in precipitation. The impacts from yearly variations can for example have an impact on the magnitude, frequency and distribution of precipitation events. In future, global warming resulting from greenhouse gas emissions is expected to lead to variability and changes in precipitation patterns (Fowler & Ekström 2009; Frei et al. 2006). This result have been consistently produced from climate models in current and past studies (IPCC 2013b; Trenberth 2011; Tebaldi et al. 2006; Houghton et al. 2001). Other studies such as Cramer et al. (2014); Williams et al. (2007); Huntington (2006) suggests that global warming may result in intensification of the hydrological cycle with changes in the frequency and magnitude of precipitation events as well as changes in the frequency of dry events (Westra et al. 2013; Seneviratne et al. 2012).

These events may result in an increase or decrease in mean precipitation and are expected to affect agriculture in future (e.g. sugar beet production in Eastern England) and therefore, a source of concern to famers and society at large. However, the assessment of precipitation is somewhat made difficult by the discontinuous and irregular nature of precipitation which makes it difficult to detect trends creating additional disadvantage if shortages were to persist. Detection of trends in precipitation analysis is also hampered by the need for long time homogenous observed datasets necessary to identify shifts in precipitation (Klein Tank & Können 2003; Klein Tank et al. 2009). This sort of data is often limited to developed countries with advanced technologies and unhindered long-term data as opposed to developing countries with sparse or interrupted data thereby creating a bias and providing uneven global representation.

A number of approaches have been employed to assess trend in precipitation data. Daily mean rainfall amount of wet days have been used to identify changes in global mean precipitation in future climate (Meehl et al. 2005). Klein Tank & Können (2003) used fixed value threshold to calculate changes in mean precipitation and identify associated changes when threshold values are exceeded. Changes and trend in precipitation are also assessed by comparing trend over a long period of time (e.g. 30 years) or assessed between two different time frames (Zhang et al. 2004) as employed in this study.

1.3 Climate models simulation of precipitation

Climate models are the main tools used for investigating projected long-term precipitation changes. The models are used in two ways: firstly, to simulate past precipitation and secondly, they are used to simulate possible future precipitation under different greenhouse gas emission scenarios. Climate models have been used to replicate long-term climatological changes in observed precipitation with considerable success (IPCC 2013b). Equally, climate models have also been used to make projections of future changes in precipitation. The use of models to simulate historical precipitation represents an alternative method for understanding changes in past precipitation and also used to validate model performance in reproducing past precipitation of the study area. This method enables the selection of desirable models based on the notion that if they can reproduce past precipitation of a region very well, they should also be able to reasonably reproduce future precipitation of the same region.

However, in spite of the skill exhibited by models, their performance is rife with uncertainties largely because of the coarse resolution of the models making them unable to adequately handle some of the mechanism associated with precipitation particularly on local scales. Such "sub grid scale" processes are usually treated statistically in models via calculations known as "parameterisation". The reliability of model output is, therefore, not only dependent on the quality of the input data, but also on the ability of the adopted parameterisations to represent all the required processes. This means that models are likely to produce valuable results if accurate assumptions are made. This uncertainty is evident in models' disagreement on the direction of potential changes in precipitation (IPCC 2013b, Bates et al. 2008). Clouds and radiation specifically are not properly simulated by models as they vary temporally thereby making individual measurements unrepresentative of the areal mean (Pincus et al. 2008). However, the greatest source of uncertainty remains the pathway of future greenhouse gas emissions, which will ultimately determine how our climate will change.

In spite of these challenges, climate models have nonetheless been used in various studies to simulate observed changes in precipitation with great success. Climate models projection of changes in precipitation over the UK has consistently shown increasing winter precipitation and decreasing summer precipitation (Jenkins et al. 2008). For this study, climate model ensemble simulations of daily precipitation data was extracted from the most recent state-of-the-art Climate Model Intercomparison Project Phase 5 (CMIP5) (Taylor et al. 2012). The CMIP5 climate models provide valuable climate information used to understand and enhance our knowledge of climate processes and their effect on society. The current study evaluated Eastern England precipitation using three Earth System Model (ESM's) ensembles for the historical phase (1971-2000) and the future period (2021-2050).

It is important to stress that reliance on one climate model run can increase uncertainties, therefore, ensembles average were used to reduce model uncertainties and error specific to individual models. Ensembles have been widely used in precipitation, meteorological and hydrological studies with a great degree of success (McSweeney et al. 2014; Duan et al. 2007) by obtaining the averages of individual model performance and spreading the results of possible outcomes.

1.4 Impacts of precipitation changes on agriculture

Climate change impacts on agriculture is arguably the most affected of all sectors and it is widely envisaged that future agricultural productions will face multiple changes as a result of continued emissions of carbon dioxide and other greenhouse gases in the atmosphere (Ainsworth & Ort 2010; Meehl et al. 2005). This situation is set to increase annual global temperature which will in turn lead to changes in precipitation with adverse consequences for agriculture. Global temperatures may rise by ~4°C or more and the combined effect of this with a growing population will increase food demands and further add to the challenge of food security (Porter et al. 2014). Furthermore, rising temperatures and changing precipitation patterns will not only affect overall food production, it will also increase the inter-annual variability of crop yields. These sort of impacts are expected to be widespread, complex and vary from one region to another, and heavily influenced by socio-economic factors (Vermeulen et al. 2012) and are already being felt in important crop growing regions of the world (Porter et al. 2014; Lobell et al. 2009) experiencing more negative impacts than positive (Porter et al. 2014).

A review of climatic impacts on the agricultural sector has shown more negative effects than positive (IPCC 2014a) and the reason for this is because agriculture is inherently sensitive to climate: any change in climate will almost certainly affect plant growth positively or negatively. These effects are already detectable where, for instance, temperature changes have been shown to have an impact on the growing season (Menzel et al. 2006). This type of sensitivity is reflected where and when food prices increase following extreme weather events in food producing areas (Porter et al. 2014). Some regions in the world are already experiencing reduction in yields and this trend is predicted to increase price volatility for agricultural commodities and reduce quality. Current research suggests that decreases in yields will grow larger and will affect both temperate and tropical regions, and will become more erratic as the weather becomes more unpredictable (Porter et al. 2014).

In Europe, variations in precipitation events, some of which resulted in floods and inundation of agricultural farmlands in June 2013 caused significant negative impacts to crop yields. Losses of crop yield in financial terms were estimated to be around €5.8 billion (Schröter et al. 2013) in Germany, Austria, Czech Republic, Poland and Switzerland put together. More than 400 farms in Germany reported crop losses while 20% of vegetable crop were lost in Austria (Euractiv 2013). These historical events demonstrate the profound damage that can be caused by extreme events resulting from variations in precipitation.

The United Kingdom (UK) like other countries has her own fair share of climatic challenges with Central England Temperature (CET) rising by one degree Celsius since 1980 (Jenkins et al. 2008), while precipitation in England has shown high seasonal and annual variations making trends difficult to detect (Met Office 2014). According to the Met Office (2014), "the character of UK rainfall has changed with more frequent heavy rainfall events very likely. What might have been a 1 in 125 day precipitation event in the 60s and 70s are now likely to become 1 in 85 day event" (Met Office 2014). This sort of change had been identified in an earlier study by Fowler & Kilsby (2003). In their study, Fowler and Kilsby reported that precipitation intensities that were previously experienced "once in 25 years, now occurs at 6 year intervals"; a consequence of increased variation of precipitation events and seasonal changes. Climate change

projections for the UK indicate that there will likely be wetter winters and drier summers (Jenkins et al. 2009; Met Office 2014). Consequently, dryer summers will impact on the growing season water supply for crop use and will potentially reduce soil moisture (Richter et al. 2006) and potentially affect yield.

In recent years, the highest profile agricultural losses occurred at the hands of extreme events resulting from variations in precipitation. For example, the UK has experienced a number of extreme precipitation events that have impacted the agricultural community. January and February 2014 in England saw precipitation totals of approximately 150 mm and 109 mm, respectively, which are well above the average precipitation values for these months (Met Office 2014). This resulted in around 49,000 ha of farmlands being flooded in a single event during February 2014 in Somerset and the Thames and Severn catchments (EFRA Committee 2014). The extent and duration of this flood resulted in more than 44,000 ha of farmland being underwater for more than one day and 40% of that area (17,800 ha) being flooded for 15 days causing significant damages to the farmlands and harvest ready crops, and loss of income to farmers (DEFRA 2014). Similarly, the May-July 2007 floods in the UK caused heavy damages to farmlands with 78 farms flooded causing a significant £50 million in agricultural losses (Chatterton et al. 2007).

According to Dorling (2014), changes and variations in weather conditions in recent years and expected future changes in precipitation coupled with the associated impacts on many agricultural systems when compared with long-term averages have presented UK farmers with some tough conditions revealing a pressing need to consider a range of adaptation strategies that will be beneficial to farmers now and in future growing seasons. In this study, how variations in precipitation impacts sugar beet yield was investigated. This is useful in order to understand the link between changes in precipitation and sugar beet yields in the near-to-medium term future (2021-2050). Furthermore, the method of assessment employed in this study is equally applicable to a wide range of agricultural crops grown over the summer season in a way that enables farmers to identify risks and opportunities associated with a changing climate, and also improve decision making with regards to future growing seasons.

In the light of the above, it is important to understand potential impacts of climate change for the different regions to enable the agricultural industries to prepare and adapt to the changes that are likely to occur.

1.5 Sugar beet production in the UK

Sugar Beet (*Beta vulgaris* L.) is a large pale brown root crop that is usually sown between March and April and harvested between September and February (British Sugar 2011). The tuber contains between 14-20% sugar (Rinaldi 2012) and accounts for about 60% of the total sugar consumed in the UK (British Sugar 2016). The crop is mainly grown under rain-fed conditions and it is an important agricultural crop in the UK: the sugar industry contributes significantly to the UK rural economy and supports up to 9,500 jobs in the wider economy and directly involves 7,000 businesses in the supply chain (British Sugar 2016). Approximately 3,500 farmers grow the crop, predominantly in Eastern England, on over 100,000 hectares of farmland (British Sugar 2016).

Sugar beet productivity in the UK increased between 1976 and 2004 (Jaggard et al. 2007). An examination of farmer delivered sugar yield was conducted by Jaggard et al. (2007) and they reported an annual increasing trend of 111 kg/ha for the period. Further, British Sugar (2011) reported an average increase of 11 tonnes of sugar beet per hectare (an approximate increase of 60% between 1981 and 2011). These increases are generally assumed to result from improved agronomy, better seed varieties and favourable weather, but these assumptions cannot be justified without taking the climate related changes and local weather patterns into consideration. According to the IPCC (2013a,b), "warming of the climate system is unequivocal" and has resulted in a lot of positive and negative impacts on agriculture. Therefore, it is important to factor climate and weather related variables into yield assessment analysis.

The most important economic aspects of sugar beet for farmers are the size of the root yield and its sugar content, which are influenced by a number of environmental factors, including weather patterns and soil conditions. Sugar beet farming in England is over 95% rain-fed with the use of irrigation being minimal (British Sugar 2011). Watering volume and timing is critically

important to the successful growth of sugar beet plants, as indicated by Richter et al. (2006) who modelled the variability of UK sugar beet under climate change using a regional climate model. They found that water will be a major stress factor in the future and that relative soil moisture will be reduced under high greenhouse gas emissions scenarios.

A review of historical annual sugar beet yields with rainfall between 2000 and 2015 (See Figure 1.2) revealed stability in inter-annual yield between 2002 and 2006, however, 2007 to 2015 showed high variations in annual yield. The yields in the later years reflect how changes in precipitation impacted on annual yield of sugar beet in the UK. For example, years of drought (2010) and flood (2012) showed low sugar beet yield. 2011 had the second highest yield of 75.4 t/ha as a result of favourable weather conditions and yield declined again in 2012 (60.7 t/ha) as a result of the summer floods. Similarly, 2014 had the highest yield of 80.1 t/ha, a result of similar good conditions as in 2011 but declined in 2015 with 69 t/ha. It is important to note that low-yield variability give rise to stable income for farmers and stable food supply, which in turn help to prevent increase in food prices. Naturally, other factors (e.g. farming practices, temperature, sunlight, diseases etc) affect yield and the overall relationship with precipitation would not necessarily be linear as the influence, particularly losses as a result of flood or drought, will be localised but, nonetheless, the data raise some interesting questions.



Figure 1. 2: Annual UK sugar beet yield and annual rainfall. Data: (DEFRA 2015; Met Office (n/d)).

In Europe, sugar beet yield is generally seen to decrease when stressed via low water conditions: Pidgeon et al. (2001) estimated potential sugar beet losses, calculated from climate and crop model projections due to water stress, vary between 15% and 30% for England. Given the nature of UK sugar beet production, past and present water limitations have most likely been driven by changes in precipitation patterns. Furthermore, many past studies have indicated that sugar beet is, more specifically, sensitive to water supply in terms of: leaf growth (Rytter 2005); storage root formation (Rytter 2005; Brown et al. 1987) and yield (Choluj et al. 2014; Richter et al. 2006; Kenter et al. 2006; Jones et al. 2003).

As sugar beet is economically significant in the UK and is sensitive to water supply, it was considered an ideal crop to investigate in the context of future changes in precipitation. The time frame for the study considered that most climate change impact assessment studies on agriculture and indeed, other sectors have focused on greenhouse gas emissions that could be either avoided or reduced by the end of the 21st century. However, this study considered the need to assess climatic impacts in the near-to-medium term future as a result of the unavoidable levels of greenhouse gas emissions already in the climate system. Furthermore, there are currently no sugar beet growing experiments in the literature that are informed by ensemble model projections – one of the aims of this study is to address this.

1.6 Aims and scope

The main aim of this study is to understand the impact of climatological precipitation changes in Eastern England on sugar beet yield. State-of-the-art CMIP5 climate model ensemble projections were used to inform greenhouse experiment in a novel way. In this thesis, the following results are presented and interpreted:

- An examination of precipitation data from weather station observations and climate model projections for Eastern England;
- A series of watering regimes, calculated from the precipitation observations and projections, which represent the climatological precipitation levels delivered to Eastern England over two seasons for the present day and future climate scenarios; and

• Measurements of sugar beet yields from a greenhouse experiment, where 150 sugar beet plants were grown in the first season and, 201 in the second season with the application of the calculated watering regimes.

Hypothesis:

- Changes in future Eastern England precipitation patterns will impact on sugar beet productivity
- Reduction in future Eastern England precipitation will reduce sugar beet yield by 2050.

The same hypothesis may be approached differently using the following research questions:

- What are the potential impacts that changes in precipitation patterns will have on sugar beet yield in Eastern England by 2050?
- How will the sugar beet plant respond to the changes in precipitation?
- What types of mitigation and/ or adaptation measures can be implemented?

Answering the research questions will contribute to understanding the potential risks and opportunities associated with changing precipitation patterns in the study area and how precipitation variations may impact on the crop investigated. This is useful in order to target research into the causal relations between yield and precipitation variability.

1.7 Conceptual model of the research study

The study employed a multi-level system approach to examine the issue of climate change and its impacts on agriculture. This is important because understanding the potential impacts of global environmental change on the sequence of interrelated and interlocking elements will help illustrate the important links and processes involved in climate change and changes in precipitation patterns which ultimately affect the farmlands that provide society with food. These links are depicted in the conceptual framework below (Figure 1.3). The shaded areas starting

from the concentration of emissions in the atmosphere through to the implications for sugar beet yield shows the pathway that this current research investigated.



Figure 1. 3: A conceptual model of the study illustrating the pathways that was of particular interest in this thesis (i.e. factors that will ultimately show the impact of changes in precipitation on sugar beet yield.

1.8 Thesis outline

This thesis is organised into seven chapters and the introduction provides the basis on which the rest of the chapters were developed. **Chapter 2** reviews current literature to-date on a range of topics covered in this thesis, including precipitation changes using climate models for present and future climates, climate observations and the impact of precipitation changes on sugar beet yield. This study expatiated on the concepts established in the introduction and identified gaps and current challenges in this field which informed on the research objectives for this work. **Chapter 3** outlines and appraises the methods employed in this thesis and is divided into two Sections: the precipitation data and analysis, and the greenhouse experiment materials and

methods used in this thesis. **Chapter 4** presents findings of changes in precipitation over Eastern England for two different time frames: 1971-2000 and 2021-2050, using daily precipitation data from eight CMIP5 climate models. **Chapter 5** extends the findings from the precipitation analysis and applied the changes to the watering regimes in the greenhouse experiment. This enabled the evaluation of how changes in climatological and more realistic watering patterns affected sugar beet yield. **Chapter 6** provides a synthesis of the findings in the study and discusses the implications for future yields. **Chapter 7** gives some conclusions and pointers in the direction of possible future work.

Chapter 2 – Literature review

2.1 Introduction

Climatic changes associated with the increasing concentration of greenhouse gas emissions in the atmosphere are expected to cause changes in mean climate conditions and more importantly, changes in the variability of climate. One of the consequences of this is the changes in precipitation characteristics and delivery which will have significant impacts on agriculture and agricultural systems. The resultant impact of these changes varies from region to region sometimes creating very challenging conditions for farmers worldwide. This highlights the need for research into this important area of food production and equally calls for timely and quality seasonal precipitation forecast that can enable farmers to plan ahead by implementing adaptation measures and reduce risks to crop production.

In the UK, reduction in future summer precipitation has been widely reported (Met Office 2014; UKCP9-Jenkins et al. 2008) but how variations in seasonal precipitation will affect crop yield and how yield will vary over time (See Section 1.5, Figure 1.2) has received far less attention (Osborne & Wheeler 2013). In the same vein, a large body of work on precipitation assessments in the UK have been heavily focused on extreme or intense precipitation. It is reasonable to assume that the focus on extreme events by most researchers has been as a result of recent extreme events mentioned in Section 1.4. This might have caused research to be less focused on longer-term, mean changes in UK precipitation with the exception of a few studies conducted over ten years ago, such as (Watterson & Dix 2003; Osborn & Hulme 2002; Osborn et al. 2000; Gregory et al. 1991; Wigley & Jones 1987).

The more recent works that focused on variations in UK precipitation are few and far between and included such studies as Met Office (2014); Maraun et al. (2008); UKCP09-Jenkins et al. (2008). Maraun et al. (2008) particularly identified changes in UK summer precipitation that are consistent with inter-decadal variations in precipitation. The work in this thesis reviews and addresses how seasonal changes in precipitation will affect sugar beet productivity in Eastern England by 2050 using the latest suite of CMIP5 climate model ensembles. The use of the CMIP5
models to inform a greenhouse crop experiment in this way has not been previously addressed in any of the literature.

In light of these studies, it is important to understand the likely impacts of future precipitation changes on sugar beet productivity in the UK. In order to do this, climate model results have been used to provide insight and make projections of likely precipitation changes now and in the future. This is important because the prudent use of quality precipitation forecasts has the potential to increase preparedness and resilience, reduce risks, optimize management practices and implement adaptation measures that can lead to desired economic and environmental outcomes for farmers and society at large. In this thesis, the following review of literature confirms the likelihood of a reduction in future UK summer precipitation and assesses how this reduction will impact on sugar beet productivity in Eastern England. The review present problems associated with precipitation variability and discuss specific solutions, and conclude by proposing that reduction in future precipitation will reduce sugar beet yield without adaptation strategies to safeguard tomorrow's agricultural productions.

Some of the issues introduced in Chapter 1 are further explored here (Section 2.2) with reference to existing literature so that gaps in the literature might be addressed. The Section delivers an overview of past and present precipitation across the UK and explores the physical theory, observational and model evidence of precipitation variability under a warmer climate from a global perspective. Furthermore, the influence of precipitation changes on hydrological cycles are discussed along with observed changes in precipitation variability. Section 2.3 addresses the issue of past and present precipitation variations and trends in UK while Section 2.4 reviews issues surrounding model simulation of precipitation events and methods of verifying model reliability of historical data from past and present precipitation data and comparing it against observed data. Previous studies on model evaluation are reviewed and impact of changes in precipitation on hydrological processes is also considered. Section 2.5 addresses the impacts of precipitation change on agriculture, food security and sugar beet yield in Eastern England. Finally, Section 2.6 summarises the review, identifies research gaps and present the research objectives in furtherance of the research aims in Chapter 1.6.

2.2 Climate change impacts and precipitation variability

2.2.1 Influence of global warming on precipitation variability

Climate change and variability have long been recognised as one of the biggest threats facing societies today and the risks are becoming more visible through the changes in climatic variables such as precipitation and temperature (IPCC 2014a). Precipitation is an important climate variable whose changes in nature, frequency, intensity and coverage has had a significant impact on agricultural production and society at large. The causes of this variation include orographic influences such as the high mountains in western parts of England and Scotland (Met Office 2016) (discussed in detail in Section 1.2) and large water bodies. The impacts of these variations vary from one region to another, and the possible consequences of these variations on crop yields have generated global concern (Gornall et al. 2010; Fischer et al. 2005). As described in Section one, variations in precipitation such as an increase in mean precipitation could lead to the risk of floods whilst a mean decrease in precipitation could lead to the risk of drought has been reported to intensify hydrological cycles (Fischer & Knutti 2016).

This problem has been at the fore-front of public debate as far back as Pre-1990, when the First Assessment Report (FAR) of the Inter-Governmental Panel on Climate Change (IPCC) was released stating that anthropogenic activities are substantially increasing greenhouse gases in the atmosphere resulting in greenhouse effect and warming of the Earth's surface (Houghton et al. 1990). In the 1990's, with reference to the FAR, the processes were partially understood and required more research to reduce the uncertainties:

- By better understanding of climate related processes particularly clouds and the carbon cycle
- In order to develop improved models of the Earth's climate system and
- Facilitate international exchange of climate data

The call to action was heeded by researchers worldwide who began to investigate the impacts of climate change on our world. Successive IPCC reports have come to the conclusion that global temperatures are rising and are likely due to the increasing concentration of greenhouse gases in the atmosphere with the Fifth Assessment Report (AR5) concluding that "there is now a

widespread scientific consensus that the global climate is changing and that human activities are responsible" and, in fact, IPCC (2013a) and IPCC (2007) concludes that this change is unequivocal.

The changes associated with increasing greenhouse gas emissions in the atmosphere continue to increase and chief among them is carbon dioxide with 76% of the total emissions in 2010 (IPCC 2014a). These emissions have increased from 27 (\pm 3.2) to 49 (\pm 4.5) GtCO₂ eq/yr between 1970 and 2010 (IPCC 2014a). This according to the IPCC (2014a) has increased in every part of the world with varying degree of impacts. In Asia, greenhouse gas emissions rose by 33% in 2010, Middle East and Africa by 70%, Latin America by 57% and the Organisation for Economic Cooperation and Development (OECD) countries by 22%. These gases are long-lived in the atmosphere as a result of their chemical stability which enables them to persist in the atmosphere for a long time. In the last three decades, the Earth's surface has been successively warmer than any other decade since 1850 (IPCC 2013b). Globally, land and ocean surface temperatures have risen by 0.85 [0.65 to 1.06] °C over the past 132 years (1880 to 2012) (IPCC 2013b).

Figure 2.1 shows increasing trend in annual mean temperature since 1980 based on simulation of observed and CMIP5 multi model mean surface temperature. The CMIP5 multi model mean showed good agreement with observations better than the individual CMIP5 models. There are however, much larger variations in the observations than the simulated multi model mean which could be explained by models tendency to simulate stronger warming in response to increasing greenhouse gas emissions in the atmosphere (IPCC 2014c; IPCC 2013b). These increases, using Carbon dioxide (CO₂) for example, contains 2 C-O bonds which are capable of absorbing infrared radiation and by so doing causes surface temperature to increase. This in turn causes changes in evaporation leading to changes in precipitation patterns and it is the reason why there are droughts in some areas and flooding or other types of extreme event such as landslide, heatwave and wildfires in others as reported in the IPCC (2013b) and successive IPCC reports.



2. 1: Observed, individual models and CMIP5 model simulated time series of the anomalies and global mean surface temperature from 1961-1990 reference period. The reference period 1961-1990 is indicated by the yellow shading, vertical dash grey lines represent major volcanic eruptions, the single CMIP5 models are represented by thin lines and the multi model mean are represented by the thick red line. The different observations are in thick black lines. Source: (IPCC 2014a; IPCC 2013b).

The impacts associated with climate change and variability have been observed to affect hydrological systems in all continents (Cramer et al. 2014). Fischer & Knutti (2016) further this position in their review of recently observed precipitation. Their report shows that hydrological intensification is beginning to emerge across many regions in the world confirming theory and early model predictions. However, this trend is not just emerging because studies such as Wang (2005) had already reported impacts on water processes. Wang, as far back as 2005 used 15 Global Climate Models (GCM's) to examine the impact of warming on soil moisture. The models were consistent with warming impact on hydrological processes and predicted that summers are getting drier and winters are getting wetter in the mid and high latitudes, and in Central Europe and Asia, about half of the models predicted wetness, and all-year-round dryness in Siberia. Findings from the study indicated high variations in precipitation in different regions with a suggestion of worldwide drought as a result of greenhouse gas warming. In another study, Bates et al. (2008) reported that changes in hydrological cycle observed over several decades are linked to global warming and are the result of changes in precipitation patterns. This, according

to the authors is because climate change is inherently associated with all aspects of the hydrological cycle in a way that changes in the cycle affects available water resources through changing precipitation patterns, changes in evapotranspiration, changes in soil moisture and runoff.

A number of other studies have reported that global warming may cause intensification of hydrological processes via increases in magnitude and frequency of precipitation events (Trenberth 2011; Williams et al. 2007). These sort of changes may impact on water resources either through floods or droughts, changes in the frequency of dry and wet events. Trenberth et al. (2007) contend that an increase in cloud cover and water vapour over the last four decades has increased the intensity of precipitation events. Furthermore, in the most current IPCC report to date (IPCC 2014a), it is projected that variations in precipitation may lead to increased episodes of extreme floods and droughts with significant impacts not only on agriculture but also on water resources, ecosystems and infrastructure (Cramer et al. 2014). Gu et al. (2007) examined changes in global and tropical rainfall using Global Precipitation Climatology Project (GPCP) data. Their report showed large variations in rainfall patterns over land which was compensated for by changes over the ocean.

These international findings have been replicated in the UK showing the global nature of this problem. For the UK, the signals tend to be drier summers and wetter winters (Jenkins et al. 2009) and it is reviewed in details in Section 2.3.2.

2.3 Past and present climatic changes across the UK

2.3.1 Variations and trends in UK precipitation

Changes in precipitation characteristics across the UK suggest that precipitation variability will continue to occur under a warmer climate with consequent changes for hydrological processes resulting in floods, droughts, and landslides, amongst others. Assessing changes in precipitation is made more complex because precipitation is such a highly discontinuous variable occurring only during wet events which are non-uniform in intensity, duration and frequency but at the same time, a very important factor in regional climate analysis.

The Met Office (2014) reported that there is no discernible trend in annual UK precipitation but reported large annual variability and provided evidence suggesting that the character of UK precipitation has changed. The Met Office claimed that "five out of the last seven summers have been wetter than average, but prior to that, there was a run of drier than average summers". The same result was reported by UKCP09 which indicated that annual mean precipitation over England and Wales has showed no significant changes since 1766 but the seasonal precipitation has shown large year-to-year variations (Jenkins et al. 2008). The records also showed an increase in winter precipitation and a decrease in the summer. Imagine these impacts being felt after only 0.8°C warming (UKCP09-Jenkins et al. 2008) and with projections of future warming and changes to the climate system (IPCC 2013b), it highlights the importance of investigating the potential consequences of further warming.

A historical look at past studies on UK precipitation indicates a reduction in summer precipitation. For example, Wigley & Jones (1987) used a simple correlation analyses to examine time series data from five regions in England and Wales and reported an increase in the frequency of extreme wet springs and dry summers. Gregory et al. (1991) expanded on the work of Wigley & Jones (1987) by increasing the number of examined regions to include Scotland and Northern Ireland. Each of the regions were divided into seven sites and daily and monthly time series data were investigated for correlation among the sites in the different regions for the period 1931-1989. Their results indicated no long-term trends in annual precipitation but showed a reduction in summer precipitation. Findings from both studies for the annual and seasonal mean precipitation showed that south-west England and south-west Wales had the highest precipitation in the UK, with an annual precipitation of 2.83 mm/day from 1931-1989. The north-west of England and north-west Wales was reported to have the second largest precipitation of 2.76 mm/day while the central and eastern parts of the UK was reported to have the driest conditions with precipitation amounts of 1.77 mm/day. Results presented here provide further synopsis of the large variations between the precipitation amounts in the different regions of the UK.

Other studies such as Osborn et al. (2000) analysed precipitation data from 110 UK stations during the period 1961-1995 and found a decline in light and medium precipitation events and an increase in heavy events during the winter. Conversely, their results also revealed a decline in summer precipitation. Similarly, Osborn & Hulme (2002) reported that UK daily precipitation

has changed over the period 1961-2000 with average rainfall becoming more intense in the winter than the summer period. In other studies conducted using gamma distribution, (e.g. Watterson & Dix 2003), simulated precipitation distribution showed consistency with events of heavy rainfall during the winter. Beyond that, non-parametric analysis was further conducted and it confirmed the contribution of heavy rainfall events to total winter rainfall amount. Summer rainfall on the other hand, indicated a decrease.

Results from more recent studies on precipitation variability in the UK (e.g. Kendon et al. 2015; Met Office 2014; Kosanic et al. 2014; Prudhomme et al. 2012; Maraun et al. 2008; Jenkins et al. 2008), have shown consistency with past studies described above. Maraun et al. (2008) examined the trends in the contribution of heavy precipitation events to UK precipitation. The authors employed the use of 689 rain gauges covering almost all of the UK. They found out that there was a widespread shifts towards heavy precipitation events in the winter and also detected the existence of a long-term increase in precipitation intensity in the winter and to a lesser extent in the spring and autumn. In contrast, the study reported much lighter and moderate precipitation events in the summer. The summer precipitation on the other hand, showed more consistency with inter-decadal variability. Prudhomme et al. (2012) used the Met Office HadRM3 climate model to assess changes in UK River flows in the 2050s. Their results showed that River flows may either increase or decrease in precipitation tends to be largely driven by variation in precipitation and reduced wet-days amount, and fewer wet days or rainfall events.

UKCP09-Jenkins et al. (2008) also reported high seasonal variations in UK precipitation for the winter (December, January and February) and summer (June, July and August). For example, the winter precipitation totals varied between 88.9 mm in 1964 and 423 mm in 1915. Likewise, summer precipitation totals varied between 66.9 mm in 1964 and 49.7 mm in 1912. In spite of these variabilities and the lack of discernible trend in annual mean precipitation, a general trend towards reduced summer precipitation was observed. It's worthy of note that some of the studies cited here are more than 10 years old but their results are still relevant with regards to UK precipitation trend and it clearly shows a trail of reduced summer precipitation from past decades compared to future scenarios. The reduction in UK summer precipitation is acknowledged and supported by recent model simulations (Met Office 2014; IPCC 2013b; UKCP09-Jenkins et al. 2008) and is further discussed in Section 2.4.

2.3.2 Variations and trends in UK temperature

Given the theoretical basis described in Section 2.2.1, it will be logical to expect that observed precipitation records will show evidence of changes related to warming that has already occurred in the UK. Like other countries, the UK is not immune to the impacts of climate change and has experienced rising temperatures, sea level rise and increasing episodes of extreme events such as floods, droughts and heat waves to mention but a few. These types of impacts have already been explored in past and recent studies (Cramer et al. 2014; Huntington 2006) which shows that higher temperatures affect hydrological cycles through changes in precipitation patterns.

Climate records in the UK show that Central England Temperature (CET) has risen by 1°C since 1970 with 2006 being the warmest year on record (CCRA 2016; UKCP09-Jenkins et al. 2008). The UK Climate Change Risk Assessment predicted an average temperature increase of 0.9°C for 2005-2014 compared to 1961-1990 with 2014 being the warmest year (CCRA 2016). In an earlier but similar report, Karoly & Stott (2006) using coupled-ocean atmosphere model simulation also reported an increase of 1°C in annual mean CET since 1950. They contend that the increase could not be explained by natural climate factors alone and concluded that greenhouse gas emissions in the atmosphere played a role in it. Similarly, Sexton et al. (2004) indicated that comparisons between observed changes and atmospheric climate models show that long-term warming of the CET cannot be explained by natural variations alone; they contend that anthropogenic greenhouse gases might have a significant influence on the CET.

In another study using individual stations across the UK, Croxton et al. (2006) reported highly correlated temperature values from the stations with the CET making it applicable beyond central England. According to the United Kingdom Climate Projections (UKCP09), UK temperature trend has risen faster in comparison to the global average of land surface temperature. Kendon et al. (2016) reported an average warming of 0.3°C in 2006-2015 compared to 1980-2010 and 0.9°C warmer than 1961-1990. Furthermore, Kendon et al. (2016) in their study explained that the top 10 warmest years in the UK temperature series have occurred since 1990 with 8 of those occurring since 2002. The highest changes occurred in the spring and autumn with 2006-2015 being 1.0°C above 1961-1990, and the smallest change occurring in the winter with 2006-2015

being 0.6°C above 1961-1990.

2.4 Models simulations of past and projected precipitations changes

2.4.1 Introduction

Climate models are mathematical simulations of the climate system and provide the main basis for forecasting changes in future precipitation under different greenhouse gas emission scenarios. These simulations produce powerful representations of the Earth's physical processes but the full complexities of the real world cannot be fully represented in any single model. This is due to internal and external variabilities in the climate system, some of which are processes acting on a much smaller scale than the models (Pincus et al. 2008). Much research has been conducted into forecasting the potential impacts of future climatic changes on the environment with considerable success however, much uncertainty exist in the process given that future emission scenarios and societal development are unknown. According to the IPCC (2013b), weather and seasonal climate predictions can be easily verified whereas climate projections spanning a century or more cannot be verified and will always be a limitation. In spite of this limitation, climate models have proved very useful in terms of skills and performance in projecting future climates. One major approach is to evaluate model performance relative to past observations and taking into account natural variability.

Therefore, to have confidence in models ability in the projections of future climates, such models must have simulated the observed climate, its variability and changes reasonably well (IPCC 2013b). The truth is, models perform differently for different regions and some models perform better than others for some variables (McSweeney et al. 2014). This implies that no individual model is better than the others. In this Section, the study reviews earliest climate model predictions, and then, considers the contributions of past and recent model projections with respect to UK summer precipitation in particular.

2.4.2 Model simulation of observed changes in the climate system

Atmospheric models have provided significant successes in predicting daily-to-weekly weather

because such forecasts are easily verifiable against local events for accuracy and reliability, whereas this approach is not applicable to climate projections because the course of future emissions is unknown. One of the earliest use of climate models in the 70s was by Broecker (1975) to simulate increasing levels of future Carbon Dioxide (CO₂) emissions in the atmosphere. Broecker used simple climate models to effectively model expected increases in future CO₂ resulting from the burning of fossil fuels. Broecker predicted that CO₂ would rise to 373 ppm (Parts Per Million) in 2000 and 403 ppm in 2010. This prediction was closely matched with actual values in later years with CO₂ recorded at 369 ppm in 2000 and 390 ppm in 2010. Hansen et al. (1981) in a subsequent experiment simulated global temperatures in the 80s and reported a rise in temperature of 0.2°C between 1960 and 1980 consequently causing a warming of 0.4°C in the past century. These increases according to Siegenthaler & Oeschger (1978) were consistent with increases of atmospheric CO₂ associated with burning of fossil fuels.

Since that time, climate model forecast have improved significantly and consistently from the time of the First Assessment Report (FAR) to the Fifth Assessment Report (AR5). After the Fourth Assessment Report (AR4), there has been further improvement in the new climate models, incorporating the Earth System model and additional physics. So, past and present climate models have been used to reproduce observed features of past and recent climates (IPCC 2013b). They have also been used to produce credible quantitative estimates of future climate change on global and regional scales. However, the predictive power of climate models is much higher for temperature than precipitation (IPCC 2013b).

Co-ordinated international collaboration has been used to facilitate, collect and disseminate outputs from model experiments under the same conditions thereby encouraging a comprehensive and open evaluation of models. This diverse effort has helped to reduce the potential of errors being overlooked and led to massive improvement in models with the advent of the Earth System Models (ESM) which include different aspects of biogeochemical elements of the climate system. ESMs have helped reduced uncertainty associated with models and the use of multi-model ensemble mean provides weighted averages arising from individual model runs by predicting the distribution of possible outcomes. This helps to reduce errors arising from the use of a single model (See Section 3.4). Figure 2.2 from IPCC (2013b) provides evidence of agreement from different models in simulating temperature and precipitation. Although, model

representation of the observed climate variables.

Moreover, the performance of climate models under similar anthropogenic scenarios produces agreement on some aspects of the climate and disagrees on others. This topic has been the focal point of some studies in the past (e.g. Harvey 2004; Covey et al. 2003; Räisänen 2001; Kittel et al. 1997). In a recent study, McSweeney et al. (2014) used downscaling technique for selecting CMIP5 GCMs over multiple regions and found that different models perform differently in different regions. This enabled selection of the best performing models to be used for the analysis and a similar approach has been adopted in this study. This Section of the thesis reviews climate model simulations of precipitation (i.e. model skill in simulating past and present day precipitation), models simulations of projected changes in precipitation, reliability of climate models, uncertainties in model simulation and CMIP5 climate model ensemble means.



2. 2: Mean seasonal cycle of (a) temperature (°C) and (b) precipitation (mm day–1). "The average is taken over land areas within the indicated regions, and over the period 1980–1999. The red line is the average over 45 CMIP5 models; the blue line is the average over 22 CMIP3 models. The standard deviation of the respective data set is indicated with shading. The different line styles in black refer to observational and reanalysis data: Climatic Research Unit (CRU) TS3.10, ECMWF 40-year reanalysis (ERA40) and ERA-Interim for temperature; CRU TS3.10.1, Global Precipitation Climatology Project (GPCP), and CPC Merged Analysis of Precipitation (CMAP) for precipitation. Note the different axis-ranges for some of the sub-plots. The 15 regions shown are: Western North America (WNA), Eastern North America (ENA), Central America (CAM), Tropical South America (TSA), Southern South America (SAF), North Asia (NAS), Central Asia (CAS), East Asia (EAS), South Asia (SAS), Southeast Asia (SEA) and Australia (AUS)". Source: (IPCC, 2013).

2.4.3 Climate model simulation of climate and precipitation variability

The issue of variations in the climate system have already been presented in Section 1.2 and is further expanded in this Section in the form of fluctuations which are directly influenced by two factors: natural climate variability and anthropogenic climate change (Deser et al. 2014). Studies such as Meehl et al. (2013); Santer et al. (2011) contend that climatic changes resulting from anthropogenic activities are now likely to dominate over natural variability for time periods as long as a decade. IPCC (2013b) addressed the issue of models simulating climate variability, both external and unforced internal variability. The IPCC report predicts that climate change will be felt more profoundly in terms frequency, intensity or duration of precipitation and extreme events (e.g., floods, droughts, heatwaves, etc).

Changes in precipitation and extreme events in all their different forms are a consequence of climatic and weather variability (IPCC 2013b) and therefore, the ability of models to simulate variability in climate is of great importance for simulating future impacts (Turner & Annamalai 2012; Colman et al. 2011). IPCC (2013b) provided evidence of simulating interannual-to-interdecadal variability including natural and anthropogenic forcing. The analysis revealed important differences between forced and unforced simulations of observed data in different time scales (Fernández-Donado et al. 2013). Their results suggest that models can reproduce variability on a wide range of time scales, although better for temperature than precipitation IPCC (2013b).

Review of literature so far point to precipitation as one variable expected to be affected by changes in climate. Model simulation of precipitation is quite challenging because precipitation is such a highly discontinuous variable occurring only during wet events which are non-uniform in intensity, duration and frequency. However, the use of climate models form the basis of our understanding for evaluating observed precipitation and projecting changes in future precipitation under different greenhouse gas emissions scenarios. Presently, key processes behind precipitation generation are fraught with uncertainties emanating from their vast complexities, including topographic influences, clouds and radiation (Pincus et al. 2008). Studies conducted to examine uncertainty in model simulation of the climate system generally agree that the magnitude and direction of potential changes in future climate system is unclear (Fowler &

Ekström 2009; Bates et al. 2008).

Furthermore, climate model simulation of precipitation is hindered by poor representation of atmospheric processes including convection, cloud microphysics, boundary layer processes and mesoscale atmospheric circulation (van Weverberg et al. 2013; Dai 2006). Models tend to underestimate heavy precipitation events and overestimate small or moderate events. According to Emori & Brown (2005), models also overestimate the contribution arising from synoptic precipitation compared to convective precipitation. Prudhomme et al. (2002) in their study showed that models have demonstrated good skills in reproducing large-scale patterns but the temporal and spatial characteristics of precipitation are still fraught with uncertainties as a result of different synoptic regimes.

Tebaldi et al. (2006) used an ensemble of 9 GCMs to simulate historical and future climates under different emissions scenarios. Their results showed that the historical simulations generally showed trends that agreed with previous observational studies thereby giving credence to the reliability of GCMs in reproducing observed climates. Model projections for the 21st century under the different scenarios showed greater temperature extremes, and increases in the frequency and intensity of precipitation events consistent with a warming climate. Meehl et al. (2007) and Trenberth et al. (2007) in their different studies came to the same conclusion that under future global warming caused by anthropogenic greenhouse gas emissions, there will be an increase in precipitation variability and the occurrence of extreme precipitation events, even in regions where mean precipitation is expected to decline (Trenberth 2011; Fowler & Ekström 2009; Frei et al. 2006).

Multiple lines of evidence from model simulations has emerged to support the physical theory that variations in precipitation will increase under a warmer climate in spite of uncertainties associated with model projections. Climate models have consistently produced accurately verifiable results (Meehl et al. 2007). For example, the high latitudes in the northern hemisphere are currently experiencing a trend towards increased precipitation and variability projected from models simulations (Trenberth et al. 2007; Alexander et al. 2006). The ability of models to reproduce observed climates and make valuable and quality projections of future climates lends support to the use of climate models in this study.

2.4.4 Climate model evaluation of projected precipitation

Climate model simulations of future precipitation under enhanced greenhouse gas emissions are broadly consistent with global increases in precipitation due to increases in precipitation event intensity (Barnett et al. 2006; Tebaldi et al. 2006). This is supported by evidence in observational data for many regions of the world which have shown greater precipitation intensity and variability. Past and current studies on climate change showed that increasing concentration of greenhouse gas emissions into the atmosphere have the potential to cause substantial changes to the climate system (IPCC 2013a; Trenberth 2011; Tebaldi et al. 2006; Houghton et al. 2001) and hydrological cycles (Garner et al. 2017). Changes in future precipitation simulated from climate model have consistently indicated an increase in global mean precipitation under increased concentrations of atmospheric greenhouse gas emissions over the 21st century (Yonetani & Gordon 2001).

According to the IPCC (2013b), model projections showed a general increase in global mean precipitation in the tropical regions and high latitudes, particularly in areas that are already showing an increase in precipitation, but show a general decrease in the arid and sub-tropical regions. The report further indicated that precipitation increase in high latitudes over the winter and summer seasons are consistent across models with widespread decreases in mid-latitude summer precipitation with the exception of Eastern Asia which indicated an increase. Several other studies such as Emori & Brown (2005) noted that although, global mean precipitation will increase, the increase will not be uniformly distributed across the globe and as a result, some regions will experience substantial increase in precipitation while others will experience drying. Pall et al. (2007) reported near-uniform increases in precipitation events at high latitudes but a strong bias of increases at low latitudes.

A number of other studies used a multi-model approach to assess the likely future changes in precipitation. Emori & Brown (2005) reported that models indicated heavy convective precipitation in low latitudes and that the intensity of heavy precipitation events will become greater in magnitude and spatial coherence. Kharin & Zwiers (2000) used the Canadian Coupled Global Climate Model (CCGCM) and reported global increases in heavy precipitation of about 8% by 2040-2060 and 14% by 2080-2100 in comparison to increases of 1% and 4% in annual mean precipitation. Emori & Brown (2005) reported findings of 6% in global mean precipitation

and 13% changes in extreme precipitation between 1981-2000 and 2081-2100 simulated from 6 climate model ensemble. Irrespective of the timeframes under consideration, these studies showed a general increase in total and mean global precipitation under future scenarios.

Different approaches have been used to gain better understanding of the processes involved in climate modelling of projected precipitation. Gleckler et al. (2008) initiated several metrics into performance analysis of climate models and emphasized the need to look further than the mean statistics for a comprehensive analysis of climate model performance. Dai (2006) evaluated mean spatial precipitation, frequency and intensity of precipitation and found that model simulations of precipitation were unrealistic although, most models captured the general precipitation pattern. The result of Dai's study is not unusual as models are known to simulate precipitation poorly. Trenberth (2011) reports that models projection have provided more robust findings in the simulation of temperature than precipitation. Similarly, Brown et al. (2010) reported findings that synoptic regimes are well described in models but the uncertainties in precipitation simulations arise due to problems in the nature of precipitation in the different synoptic regimes (e.g., frequency, intensity, duration). Catto et al. (2013) evaluated frontal and non-frontal precipitation. They reported that the frequency of frontal precipitation was well captured but the intensity of frontal precipitation was underestimated by the model.

More recent studies such as Lebsock et al. (2013) have reported biases in precipitation simulations arising from warm clouds that are not associated with parameterisation. Wehner et al. (2010) investigated the ability of GCMs to reproduce observed daily precipitation totals in the United States for a 20-year period. They reported that many of the models underestimated the return values and suggested that increasing the horizontal resolution could improve the estimates. Different studies have offered different explanations for the causes and/or sources of uncertainty in model simulation of precipitation. Tebaldi et al. (2011) explained that the criteria used to determine the agreement level between different models have often ignored internal variability in the process. They used a multi-model approach to separate the lack of climate signal from model agreement by evaluating the level of agreement on the significance of change. Their result showed that disagreement among models in precipitation projections is due to noise from climate variability masking the signal.

In continental Europe, models suggest that there may be more intense precipitation in many parts of Europe (Giorgi et al. 2004). A multi-model analysis used to construct precipitation trends over the 21^{st} century reported a decrease in precipitation to the south over Spain. The signal over Europe and indeed, other parts of the world showed some individual and seasonal weather patterns but these sorts of isolated or even infrequent events cannot be used as a sign of climate trend. Studies by Pal et al. (2004) reported that projections of increased frequency of floods and droughts over Europe are remarkably consistent with projections of climate change scenarios. Timbal et al. (1995) examined the effect of double CO₂ on precipitation using the Meteo-France atmospheric GCM and their results indicate a 6% increase in winter precipitation over Northern Europe. In past and present studies, globally and in continental Europe, models have been consistent in their projection of changes in future precipitation.

Overall, models have exhibited considerable skill in reproducing characteristics of observed precipitation and providing reliable estimates of predicted future precipitation that have come to pass. However, there remain areas where uncertainties exist and efforts to overcome this problem have led to improvement in models starting with the advent of the Coupled Model Intercomparison Project Phase 3 (CMIP3). The CMIP3 made evaluation of the atmospheric and oceans simulations better to understand. Prior to CMIP3, studies conducted before the introduction of various Model Intercomparison Projects (MIPs) had simulated climate using individual models. However, with advancement and improvement in climate models, the use of such MIPs as the Atmospheric Model Intercomparison Project (AMIP) (Gates et al. 1999) and Coupled Model Intercomparison Projects (CMIP) (Covey et al. 2003; Meehl et al. 2000a) have made it easier to compare models for differences and similarities. According to (Raisanen 2007), MIPs have enabled modellers to easily identify errors specific to individual models and make adjustments to areas that need attention. Several other studies have examined different aspects of precipitation using CMIP3 models. For example, Phillips & Gleckler (2006) in their study of seasonal mean continental precipitation used CMIP3 models to evaluate precipitation amounts and reported that many of the models differed from several observed estimates. The study also found that the ensemble mean precipitation was closer to that of the observed data than any individual model.

Further improvements to climate models continued with the introduction of the Couple Model Intercomparison Project Phase 5 (CMIP5) climate model. The current CMIP5 climate models were incorporated with an unprecedented suite of biogeochemical factors designed to simulate historical and future climate scenarios (Taylor et al. 2012). The ensemble mean precipitation of the CMIP5 models according to Kumar et al. (2013) matched that of ground based observation data but substantial biases exists in regional trend. A good number of other studies such as Chadwick et al. (2013); Hao et al. (2013); Kelley et al. (2012) have evaluated past and projected changes in precipitation using CMIP5 simulations at both regional and global levels with considerable success. A comparison study between CMIP5 and CMIP3 ensemble mean for reproducibility of observed precipitation distribution showed that the CMIP5 ensemble revealed a bit more skill than the CMIP3 ensemble (Hirota & Takayabu 2013). Using CMIP5 simulations Gaetani & Mohino (2013) studied decadal precipitation over the Sahel and reported that the predictive skills of the CMIP5 simulations vary significantly from model to model consistent with the findings of McSweeney et al. (2014).

The various studies relating to the CMIP5 in this Section have demonstrated that the CMIP5 showed remarkable skills in the projections of future precipitation. The CMIP5 climate model ensemble is used to explore uncertainties arising from internal variability, boundary conditions and parameter values in individual model structure (McSweeney et al. 2014; Gaetani & Mohino 2013; Kumar et al. 2013; Chadwick et al. 2013; Hao et al. 2013; Kelley et al. 2012). IPCC AR5 also agree that the CMIP5 climate model ensembles help to better characterise uncertainties associated with individual models, and thus, provide support for its use in this thesis (Flato et al. 2013).

2.4.5 Climate model simulation of past and future UK precipitation

In the United Kingdom (UK), predicted future warming are also expected to cause changes in precipitation patterns. In this Section, past and recent assessments of UK precipitation are reviewed under different climate change scenarios. Hennessy et al. (1997) examined zonal precipitation changes and extremes using the UK Met Office high resolution GCM and the Australian CSIRO9 GCM. They reported a 10% increase in annual mean precipitation under a doubling of CO_2 emission scenario. Jones & Reid (2001) conducted a similar study using the Met Office HadCM2 and reported an increase in winter and convective precipitation across the UK. Their results also indicated heavy precipitation events across Scotland for all seasons.

Similarly, Semenov & Bengtsson (2002) assessed the effect of increased CO₂ on precipitation from 1900-2099 using the Max-Planck Institute coupled atmosphere ocean GCM. Their study revealed an increase of about 20% in UK winter precipitation per century. Watterson & Dix (2003) used a five member climate model ensemble to examine changes in precipitation under rapid increases in CO₂ using the CSIRO Mark2 coupled atmosphere ocean GCM. The results were comparatively assessed and they indicated between 10-30% increase in winter precipitation over the UK between 1961-1990 and 2071-2100. Other precipitation assessments over the UK suggest a decrease in mean summer precipitation but with an increase in intensity of heavy precipitation events (Christensen & Christensen 2004). The work of Fowler & Kilsby (2003) also came to the same conclusion. Fowler and Kilsby used regional frequency analysis to estimate precipitation observations over the UK and their results showed that precipitation, particularly extreme precipitation intensities previously occurring once in 25 years, is now occurring at a 6 year interval. This, the authors claim was a consequence of increased precipitation frequency and seasonal changes.

Results of several studies conducted on UK precipitation showed an increase in annual total precipitation. This increase does not reflect the character of UK precipitation for all seasons because most of the increases occurred in the winter as a result of increase in precipitation frequency and intensity. Autumn and spring to a lesser degree also recorded increases in precipitation but, this is not the case for summer precipitation which has shown reduction in mean and total summer precipitation resulting from reduced number of precipitation events. The reduced precipitation observed in UK summers and projections of a reduced UK summer precipitation is one the issues addressed in this thesis.

Contrary to other studies which showed a reduction in summer precipitation, Fowler & Ekström (2009) used a combination of GCMs and RCMs to investigate and simulate precipitation in nine UK regions for the period 2070-2100 under the SRES A2 emission scenarios. Their results showed increasing precipitation trends in the winter, spring and autumn whereas projections for the summer returned no change in precipitation. However, Fowler and Ekstrom advised caution interpreting this result because of poor model resolution. This result deviates a little from other studies on UK precipitation which clearly shows a reduction in summer precipitation (e.g. Met

Office 2014; UKCP09-Jenkins et al. 2008). The study of Fowler and Ekstrom reflects the tendency of bias that can occur in any experiment resulting from the use of models that are less desirable for a region in addition to the problem of poor model resolution. Such models produce more intense and heavy precipitation in the summer which the other models may not produce because simulations are strongly dependent on physical local processes upon which parameterisation of individual models are based. Improper representation of the physical processes makes it difficult for models to estimate future precipitation on local scales (Haugen & Iversen 2008).

Two techniques have been used to overcome these challenges. Firstly, statistical downscaling is used to convert relatively coarse data from the models into local scale estimates based on statistical relationship. The second technique is dynamical downscaling technique which uses higher resolution Regional Climate Model (RCM) that are nested within GCM's in a way that the GCMs are used to provide initial boundary conditions. This technique was used by Jones et al. (1997) to examine the effect of doubling CO_2 in a comparative two ten-year experiment. Their results showed between 0.5 mm/day and 1 mm/day mean precipitation over the UK against a mean of 2-3 mm/day. In another study, Raisanen & Joelsson (2001) used two different climate models to examine annual mean precipitation over the UK. Their experiment which was classified into control run and increased greenhouse gas emissions was run over two ten-year integrations and their results showed a 10% increase in annual mean precipitation over the UK.

Recent advancements in model properties have also led to the use of multiple models as a way to address the limitations associated with individual models. For example, the United Kingdom Climate Projections (UKCP09) used probabilistic projections based on climate model ensemble to try and reduce inherent uncertainty in the models while also overcoming poor model resolution. Similarly, the recent use of the latest CMIP5 climate model ensembles in the IPCC Fifth Assessment Report (AR5) showed that the CMIP5 models perform very well in reducing uncertainties on local scales through improved model resolution (Taylor et al, 2012). These qualitative arguments have been supported by early and recent model simulations showing trends of reduced summer precipitation in the UK and the review of literature on this topic shows that the most consistently picked up signal from climate models for the UK is the increase in winter precipitation and decrease in summer precipitation a conclusion also reached by (Christensen et al. 2007).

2.5 Climate change and agriculture

2.5.1 Introduction

Climate change and variability is one of the biggest challenges facing societies today and reviews of its impacts on agriculture and agricultural systems have shown considerably more negative effects than positive (IPCC 2014a). Agriculture to a large extent depends on climatic variables such as precipitation, temperature, radiation, etc., as well as local environmental conditions in order to flourish (Crane et al. 2009). This makes agriculture susceptible to changes in local climatic conditions during the growing season (Crane et al. 2009; Sivakumar 2006). Both seasonal and/or annual variability in precipitation have the potential to lead to extreme events such as floods or droughts with significant negative impacts on crop productions and yield. This inherent sensitivity of agriculture to climate means that any change in climate will affect plant growth positively or negatively.

This sort of situation also has significant impacts on farmers income and food security (Cantelaube & Terres 2005). Therefore, to reduce the impact of changes in climate on agricultural productions, the use of climate model and climate information data are necessary to assess, anticipate and plan ahead for possible changes in climate. This has the potential to increase preparedness and lead to positive economic and environmental outcomes. In this Section, the following review of literature on climate change and agriculture will be used to address how changes in climate particularly, precipitation have impacted on agriculture in the past and present, and how it will affect agriculture in future. Furthermore, the knock-on effect it has on food security is examined and a detailed examination of the impacts of changes in precipitation on sugar beet production in Eastern England is carried out. Finally, the research gaps and objectives are identified in order to develop targeted and effective adaptation responses.

2.5.2 Climate change impacts on agriculture

Changes in future climate have been widely predicted in various studies to have significant impacts on agriculture (IPCC 2014a; Met Office 2014; Crane et al. 2009; Jenkins et al. 2009; Sivakumar 2006). Climate change and variability can have significant negative effect on agricultural productions and yield which can in turn lead to food shortages in some parts of the

world (Porter et al. 2014). Additionally, natural hazards associated with precipitation variability such as floods and/or droughts may have further impacts on food security which includes access to food, availability of food, food stabilization and utilization (Bates et al. 2008).

Historically, the use of land for agricultural purposes has always been a major occupation globally and an important aspect of human endeavour accounting for an estimated 1.2-1.5 billion hectares of cropped land (Howden et al. 2007). However, changes in climate and its impacts on agriculture and food production have not kept pace with population and socio-economic challenges. Current and future impacts suggest that agriculture faces the unenviable prospect of continued climatic changes with varying degrees of impacts in different regions of the world as explained in Section 2.2.1.

A region's vulnerability will depend on the nature and level of climate change experienced, and also the capacity of local systems and populations to adapt to change. Importantly, annual variations in climate have been the most dominant source of interannual crop/yield variations in many regions and continue to be a current and future challenge to agricultural systems worldwide. Majority of studies conducted to examine the impacts of climate change on agriculture have focused more on mean historical and projected climatic changes on agriculture (e.g. Wheeler & von Braun 2013; Urban et al. 2012; Lobell et al. 2011) but how climatic variations impact on interannual crop yield and how it may likely vary over time have received less attention (Osborne & Wheeler 2013; Chen et al. 2013).

Past studies such as Howden et al. (2007) examined the impacts of climate change on agriculture and reported that between 15%-35% variations in global yield of wheat, oil seed and coarse grains are as a result of climate change. Similarly, Ray et al. (2015) evaluated crop yields against climate change using detailed crop statistics time series and reported that about 32%-39% of the variations in observed yield were accountable to variations in climate. Simply put, crop yield variability will respond to changes in weather conditions of which crops are susceptible. For example, Agriculture was severely affected by the Russian heat wave of 2010 which contributed to a high rise in global food prices thereby affecting food affordability. The 2000 UK Autumn Floods also had negative impacts on agriculture reducing yield of sugar beet that year (see Section 1.5, Figure 1.2). These type of events created increased global interest on how climate change will affect future volatility in food prices and food security (Urban et al. 2012).

Likewise, high global temperatures have been found to be the cause of several reductions in crop yields in different regions. Some past studies have used historical crop data to examine the impacts of temperature on global agriculture and some of these studies indicated negative impacts on crop yields. Liu et al. (2016) showed that an increase of 1°C in global temperature could result in decline of 4.1%-6.4% in projected global wheat yield. Zhang & Huang (2012) found negative correlation between observed yield and temperature for maize but reported a positive correlation for rice. Similarly, Peng et al. (2004) reported a reduction in the yield of rice under a minimum increase in temperature. Future projections from climate models suggest an increase in heavy precipitation events on the one hand, while increasing temperatures on the other hand, will cause an increase in evapotranspiration thereby contributing to dry conditions (Karl et al. 2009). This sort of changes is characterised by high level of uncertainty in wet and dry events because the direction of future changes are unknown (Bates et al. 2008; Giannini et al. 2008). These changes manifest as seasonal and annual variations with some seasons characterised by high temperatures and inadequate precipitation and other seasons by low temperatures and heavy precipitation. Both of these hazards have been widely reported to have negative impacts on agriculture and agricultural productions (IPCC 2014a; Tao et al. 2003).

In future, the world faces a range of climate scenarios depending on the level of greenhouse gas emissions in the atmosphere. Some of the changes will be inevitable but the direction, extent and severity of the changes are uncertain and will vary across different regions. For example, increasing average temperatures across the globe will have positive impacts on crop production in the mid to high latitudes but will have a negative effect in the arid and tropical regions (Porter et al. 2014). Average global precipitation is expected to increase but regional patterns will vary: some areas will have more precipitation while others will have less. This uncertainty makes regions that are highly dependent on seasonal and rain-fed agriculture particularly vulnerable to the impacts of climate change. This is the case with sugar beet production in Eastern England because it is heavily dependent on seasonal rainfall (i.e. 95% dependent on seasonal rainfall) (British Sugar 2016).

Additionally, availability of water could become a challenge under a changing climate with serious implications for global agriculture. It has been reported that agriculture is greatly hampered by insufficient water which in effect reduces crop yields (Choluj et al. 2014).

Moreover, projected high temperatures with reduced precipitation will have a direct impact on underground water thereby causing a reduction in the water table which affects plants Available Water Capacity (AWC) and soil moisture necessary for crop growth, development and yield. Conversely, heavy precipitation events have been shown to impact negatively on crop yields reported in different parts of the world. These sorts of events are particularly challenging for agriculture because agriculture is inherently sensitive to climate: any change in climate will almost certainly affect plant growth positively or negatively. These effects are already evident in regions where, for example, temperature changes have adversely affected agricultural crops during the growing season (Menzel et al. 2006).

Furthermore, the anticipated future changes in climatic conditions and its associated impacts on many agricultural systems suggest the need for a broad and pressing adaptation strategy (Dorling 2014). Stocker et al. (2013) reported that climate change and variability resulting in changing precipitation patterns are likely to become more frequent and more intense but the impact on crop yield is difficult to calculate and adapt to: in the isolated regions where floods occur, yields is reduced to zero but other areas may not be affected (Okom et al. 2017). More so, analyses of extreme events over future timescales are likely to be dominated by uncertainty due to the nature of modelling studies (Maraun et al. 2010). Conversely, a more climatological analysis has the potential to produce results that can more confidently be used to plan and adapt operating practices in order to maximise yields with particular reference to food security.

2.5.3 Climate change: Implications for food security

Food security is defined as a "situation that exists when all people, at all times, have physical, social and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life" (FAO 1996). Food systems and productions depend on agriculture which is intrinsically tied to climate. Climate variability and change have been shown to affect agriculture more negatively than positive (IPCC 2014a) but the impacts vary from region to region particularly on local scales threatening food security. The threats on food security have revealed a robust global pattern of impacts on crop yields in many regions. This is very challenging for agriculture and threatens the overall ability of farms to supply sufficient food for an ever increasing global population predicted to reach 9.1 billion in 2050 (FAO 2009), while sustaining an already stressed environment. Notable impacts such as floods and droughts have direct effect on farmland conditions and indirect impacts on yields and therefore income and demand for agricultural produce.

According to Jones & Thornton (2003), "climate change impacts has the potential to increase the development challenge of safeguarding food security and reducing global poverty". It is important to state here that challenges to food security may lead to food insecurity with associated increase in food prices and limited access to food. This have been observed in places where and when food prices increase following cases of extreme weather events in food producing areas (IPCC 2014a) thereby creating food affordability issues. The World Bank (2013) estimated that 767 million people in the world still live below the poverty line of less than \$1.25 per day. Continuing population growth and consumption on the one hand will increase the global demand for food while on the other hand, competition for land, water, energy and other resources will impact on the farms ability to produce sufficient food for future population and that is in addition to the challenges of climate change. In a nut shell, climatic changes resulting in annual variations in weather events increases the risk of hunger in affected areas.

Variations in climate have been reported in many studies to affect food security. von Braun (2016) reported that warmer temperatures and variation in precipitation is likely to reduce global food production by 20% in 2050 while, Liu et al. (2016) reported 4.1% to 6.4% yield loss in global wheat production as a result of 1°C rise in temperature using crop models. Several other studies such as Lobell et al. (2008) used 20 crop and climate models to analyse climate risks

projections on crops in 12 food-insecure regions by 2030. Their results showed that without adaptation, several crops will suffer reduction in yield especially in food-insecure regions thereby increasing the gap between average and potential yields. Ray et al. (2015) reported that more than 60% of global crop yield variability could be explained by climate variability and that different climate variables such as temperature and precipitation, affect crops disproportionately resulting in year-to-year variations. Their study revealed that majority of crops harvested in different regions experienced the influence of climate variability on yields: ~52% of rice yield experienced yield variability, ~78% of maize, 75% of wheat and 67% of soya beans. These variabilities in crop yields have obvious consequences for food security.

Review on the assessments of climatic variabilities on food security under different assumptions point to future challenges ahead for food security in spite of the progress made in recent decades. In a recent report by the World Food Programme, 200 million people have been said to be lifted out of poverty in the last twenty years with a significant reduction in the level of persistent malnutrition from 40% to 26% (WFP 2017). However, the FAO (2016), believes that the challenge is still monumental with a global demand for food projected to rise by 60% in 2050. The reason why this challenge is so big is because the impacts from climate variability affect all four dimension of food security, namely: access, availability, stability and utilisation.

2.5.3.1 Challenges in addressing food security

In order for communities and societies to be food secure, they will have reliable source of affordable food at all times for all. This explanation is hinged on the four key dimensions of food security:

- a) Availability: availability of sufficient food from agricultural systems and land use to meet food demand
- b) Access: physical access to food produce and the ability of individuals to have financial resources to acquire appropriate food for a nutritious diet.
- c) Stability: stability of seasonal and inter-annual food supplies. One important aspect of food instability results from climatic variations e.g. extreme events
- d) Utilisation: deals with food safety and quality aspects of nutrition

Climate change affects food security directly via weather because of changes in the length of the growing season. The changes in weather are reflected in changing market conditions, food prices and the supply infrastructure (Gregory et al. 2005). For example, a flooded farmland with extensive destruction to crops and harvest ready crops will create a yield gap which will inevitably drive food prices up with bigger impacts on the poor that may not have access to the food or may not be able to afford it based on high prices. Achieving food security in the coming decades according to Poppy et al. (2014) will require 50% increase in food production by 2030. A number of past studies have suggested different ways to address the challenges of climate change on food security. Watson (2001) suggested the expansion of arable lands to increase food productions in order to meet future needs. This suggestion will historically impact on the environment, reduce biodiversity and limit ecosystem services. Poppy et al. (2014); Godfray & Garnett (2014) proposed sustainable intensification of agriculture as a way of reducing hunger without further environmental degradation

The Institute of Mechanical Engineers (IMechE 2013) attributes part of the problem to food waste with current practices wasting up to 50% of food before reaching the dining table. Gregory et al. (2005) argued that computer models and scenarios used in the assessment of food security did not put political response, behavioural patterns and technological change into consideration and as a result, predictions have been for a worst case scenario because only climate factors were considered while all other factors are held constant. Addressing the issue of food security and agricultural productions becomes more challenging with climate change impacts on agriculture. In the light of these, it is important to understand potential impacts of climate change for the different regions to enable the agricultural industries to prepare and adapt to the changes that are likely to occur.

In the UK, there are robust signals in climatic variables such as precipitation and temperature which can be assessed on annual and seasonal timescales in order to understand potential future impacts of climate on agriculture and food security. For the UK, this signal tends to be wetter winters and drier summers (Met Office 2014; Jenkins et al. 2008) described in Section 1.2.

2.5.4 Future impacts of precipitation changes on UK agriculture

Changes and variability in future UK precipitation have been widely reported in various studies including Met Office (2014); Jenkins et al. (2009) as being expected to have significant impacts on agriculture. In the United Kingdom, Agriculture is a major land use covering an estimated 70% of UK's total land area and employing around half a million people and contributing about £7.1 billion to the UK economy (CCRA 2016) and therefore considered to be a very important contributor in the UK. However, the impacts from climate change threaten the sustainability and future of this sector. For example, climatic variables associated with climate change and warming have given rise to changing precipitation patterns resulting in different impacts on agriculture such as flooding of agricultural lands and drier soils leading to droughts. Most notable events and impacts on UK agriculture have been more negative than positive as described in Section 1.1, but there are also potential opportunities that can benefit existing agricultural systems and encourage diversity (CCRA 2016). To extend the argument further, impacts on agriculture are manifested in a number of negative (threats) and positive (opportunities) ways on crop production including:

Threats

- Increase in flood risks to productive agricultural lands
- Drier soils resulting from warmer and drier conditions
- Increased risks of damage to crops and harvest ready crops
- Depletion of soil nutrients through soil erosion
- Increase in water demand for crop irrigation
- Increased risks of pests and diseases

Opportunities

- Opportunities to grow new crops
- Warmer temperature changes favouring some C₃ crops such as sugar beet
- Changes in the length of the growing season due to warmer conditions
- Earlier sowing date

Future projections of climate change in the UK follow current trajectory of increasing

temperatures, rising sea levels, changing precipitation patterns, increased frequency and intensity of extreme weather events, risk of flooding from rivers and seas, wetter winters and drier summer (CCRA 2016; UKCP09-Jenkins et al. 2008). Impacts on agricultural crop productions will vary across the UK with projections showing that warmer, drier summer conditions may have more adverse effect in the south and east than in the wetter areas of the north and west of the UK. This sort of variations makes climate change a current issue and future concern for agricultural productions. Agricultural productions may also favour the cool, wet upland areas due to warmer, drier conditions than the lowland areas.

Future climate trends for the UK indicate wetter winter, spring and autumn while summer is projected to be drier compared to the past. This will impact negatively on water availability for summer-sown crops which the winter-sown crops are not likely to experience. DEFRA (2004) reassessed drought related yield loss of sugar beet using LARS-WG stochastic weather generator and the HadRM3 climate model for low and high emission scenarios. Their study showed negative consequences for soil moisture with a deficit of 8-12% for low-high emissions scenarios by 2050's. Their results also showed greater warming in the future with wetter winters, spring and autumn but a much drier summer than in the past. The study concludes that drought related yield loss for sugar beet yield will increase between 30-50% by the 2050's (low-high) depending on the region.

Furthermore, warmer temperatures will increase the length of the growing season thereby encouraging earlier sowing date for crops and enabling the crops to attain full canopy cover to aid the capture of maximum sunlight in the spring. Additionally, increased CO_2 in the atmosphere is expected to increase the production of some C_3 crops such as sugar beet. However, continued high temperature if accompanied by shortages in summer precipitation will increase the risk of heat stress significantly reducing soil moisture and also impacting on crop yield. Conversely, increased precipitation intensity and frequent storm events could also lead to the risk of flood causing substantial losses in crop production in the low-lying agricultural areas. This could contribute to the risk of soil erosion, compaction and water logging thereby making the agricultural farmlands unsuitable for productions (Morison and Matthews 2016). A number of studies such as Schmidhuber & Tubiello (2007) have reported that climate change could significantly alter agriculture and agricultural productions in a number of ways. Their study revealed that increasing greenhouse gas emissions in the atmosphere will cause warming and

bring about changes in the suitability of farmlands and crop yields.

Higher temperatures could mean increased risk of drought and increased evaporation rate with potential impacts on soil moisture and crop yield. In addition to that, there is a potential threat to water resources resulting from declining summer flows, reduced ground water replenishment and increased soil and atmospheric evaporation. These factors all contribute to considerable negative impacts on crop yield and agriculture in general. Although, the UK is considered to be food secure, continued impact of weather variations could have severe implication for food security with extended socio-economic impacts in the UK.

2.5.5 Impacts of climate and precipitation on UK sugar beet production

Sugar beet production in the UK has shown considerable annual variations in recorded sugar beet yield in the last decade (See Section 1.5 and Figure 1.2). Some past studies have reported increases in annual UK sugar beet yield over the years including Jaggard et al. (2007) who reported 111kg increase in annual farmer delivered yield between 1976-2004. Likewise, British Sugar (2011) reported a 60% increase in sugar beet tonnage from 1981-2011. These studies have considered cumulative annual mean yield over a long period of time (e.g., 30 years) but have not considered climate related changes such as variations in annual precipitation in relation to variations in annual yield. These studies provide an indication of mean changes in yield but do not reveal the complete impact of the changes in annual related weather events on yields (e.g., years of low and high yields).

The review and assessment of changes in weather events such as precipitation is very important in the growth and yield of crops because crops are susceptible to weather events and respond to shifts in weather conditions causing variations in yields. Considering the variable nature of UK annual and seasonal precipitation, sugar beet productivity in Eastern England for example has been reported to be seriously affected by water stress as reported in the works of (Chołuj et al. 2014; Shrestha et al. 2010; Richter et al. 2006; Choluj et al. 2004; Pidgeon et al. 2001). Therefore, in order to ensure a crop secure future for sugar beet, sugar beet farming and production must be more resilient and adaptive to changes in climate particularly precipitation patterns. Several past studies on sugar beet have addressed a range of issues focusing on climate change impacts (Jaggard et al. 2007) and the effects of weather variables on sugar beet yield (Kenter et al. 2006). Kenter et al. (2006) and Scott & Jaggard (2000) have reported that sugar beet growth is characterised by slow leaf formation particularly in the spring. Sugar beet leaf has been identified as having a poor performance in capturing solar radiation making them unable to achieve full canopy closure necessary for light interception by the middle of June at which time they must have missed out on the much needed seasonal radiation (Kenter et al. 2006; Scott & Jaggard 2000; CGIAR 2000). This is important because sugar beet plants require canopy development and radiation to coincide in order to capture maximum radiation and therefore maximum yield. Scott & Jaggard (1978) suggested that sugar beet yield could be improved by enhancing leaf formation through earlier sowing dates to improve radiation capture early in the growing season.

Studies have also been carried out by Lauer (1997; Smit (1993) as far back as the 90s to investigate various sowing dates in order to primarily identify optimum sowing dates for sugar beet plants. Results from these studies show that early sowing in March could improve yield of sugar beet as leaf development is enhanced and plants are able to reach optimum canopy cover enabling maximum radiation capture. The results of these studies are still relevant to date. In recent years, however, there has been improvement in radiation capture and yield increases in the UK (Jaggard et al. 2009). Increase in mean yields from official trials rose to 110 kg/ha per year and some of these increases were attributable to improved seed quality and earlier sowing dates to synchronize with availability of solar radiation (Jaggard et al. 2009). Making a case for sugar beet production in England, the current study aligns with Jaggard et al. (2009) on the earlier sowing dates of sugar beet in the UK (e.g. between March and early April).

A considerable body of work examined the impact of weather variables (e.g. precipitation, temperature and extreme weather events) on rain-fed sugar beet yield and yield variability in the UK. Higher temperatures during the growing season allow spring-sown crops to be sown earlier due to an increase in the length of the growing season in addition to rising levels of CO_2 in the atmosphere which increases the rate of photosynthesis (Lawlor & Mitchell 1991). Climate model projections for the UK predict an increase in winter precipitation and a decrease in the summer which simply means wetter winters and, warmer and drier summers (UKCP09-Jenkins et al. 2008; Hulme & Jenkins 1998). Consequently, drier summer for spring-sown crops will likely

impact on water availability for crops use and potentially reduce soil moisture particularly in the months of July and August (Richter et al. 2006). In this type of circumstance, yield could therefore decline especially in areas that are already under water stress conditions as pointed out by (DEFRA 2004; MAFF 1999). Eastern England, with a high annual and season precipitation variations and the smallest amount of precipitation in the UK is a good example of an area that could be susceptible to water stress which could potentially impact on sugar beet yield in the region.

Sugar beet yield is generally seen to decrease when stressed via low water conditions: Pidgeon et al. (2001) estimated potential sugar beet yield losses, calculated from climate and crop model projections due to water stress, vary between 15% and 30% for England. Given the nature of UK sugar beet production which is 95% rain-fed, past and present water limitations have most likely been driven by changes in precipitation patterns. Furthermore, many past studies have indicated that sugar beet is, more specifically, sensitive to water supply in terms of: leaf growth (Rytter 2005); storage root formation (e.g. Rytter 2005; Brown et al. 1987) and yield (e.g. Chołuj et al. 2014; Richter et al. 2006; Kenter et al. 2006; Jones et al. 2003).; and Drawing from this and other sources reveal that water availability is crucial for sugar beet yield and indeed agricultural productions and it is already a major stress factor globally and in the UK (Chołuj et al. 2014).

In Europe, Pidgeon et al. (2001) used the UK Hadley Centre Global Climate Model (HadRM3) to simulate potential and rain-fed yields in Europe from 1961-1995 and reported drought losses of about 40% in eastern Ukraine and southern Russia. Conversely, Jones et al. (2003) reported an increase in yields in northern Europe due to warmer springs but western and central Europe are predicted to show increased losses due to drought stress and year-to-year variability in precipitation. Jones et al. (2003) contend that this could be more challenging for crop growth and yields when increased industrial and urban water use are put into consideration. Other studies such as Richter et al. (2006) reported that yield gap is predicted to increase under low and high emission scenarios and may reach 22 and 26% by 2050. This variation in annual yield gap is a reflection of the variations in future climate.

Furthermore, Richter et al. (2006) also identified crops growing in areas of low water availability to be more challenged by prolonged drought and suggested two management strategies to address this issue: earlier sowing date and harvesting at a later date. Their model indicated that earlier

sowing date in March and mid-April strongly affected yields resulting in gains of about 0.2-0.3 t/ha^{-1} . Similar result was obtained in an earlier study by Durrant et al. (1993) in a field experiment showing that earlier sowing dates were beneficial for sugar beet growth and yield but a delayed sowing date (e.g. end of April) particularly during wet spring can be less beneficial and cost growers about 1.4 t/ha^{-1} of sugar. Qi & Jaggard (2008) assessed climate change impacts on sugar beet yield in the UK between 1976 and 2006, and future climate scenarios under high CO₂ in 2020, 2050 and 2080 using Broom's Barn crop model and GCM outputs. Their results showed that sugar beet yield were likely to become more variable in future, a result supported by the findings of Richter et al. (2006). Jaggard et al. (2007) also examined the impact of climate change on sugar beet in the UK using crop simulation model and daily weather data and arrived at the same conclusion.

To date, one of the strongest studies with regards to sugar beet yield and climate change in England include that of Richter et al. (2006). Their study modelled differential water distribution and its impact on future drier climate under a weather generated historical period for 1961-1990 and projections based on low and high emissions scenarios. Findings from this study revealed a deficit in relative soil moisture and yield gap resulting from drought related yield loss. The current study builds on the work of Richter et al. (2006) by investigating the potential impacts of likely future precipitation changes on sugar beet yield in Eastern England using CMIP5 climate model ensembles to inform a greenhouse crop experiment. The ensemble is a combination of multiple individual models. The individual models in an ensemble are initiated using slightly different but equally realistic initial conditions in order to capture some internal elements of climate variability. The ensemble means provide better averages over individual model mean.

In conclusion, sugar beet has been shown to be economically significant in the UK (see Section 1.5) and is sensitive to water supply, and was considered an ideal crop to investigate in the context of future changes in precipitation. Furthermore, there are currently no sugar beet growing experiments in the literature that are informed by ensemble model projections – one of the aims of this research is to address this. The review of literature shows that the nature of climate change and indeed, precipitation variability is very dynamic and its future changes are uncertain. Research into this problem have been diverse and varied, but most studies had focused on assessment of long-term changes often neglecting the short-term or medium-term changes that

is also addressed in this thesis. One limitation in this study is the fact that agricultural production is not affected by precipitation alone, but also by other factors such as temperature and radiation amongst others. Therefore, further research will be required in future to expand on the work in this thesis. This will provide the basis for and enable extensive adaptation strategies aimed at reducing errors in precipitation and climate predictions over future growing seasons. This will enable farmers to make informed decisions that will help reduce potential risks, optimise farm management practices and implement adaptation measures that will be beneficial in future growing seasons.

2.6 Research gaps and thesis objectives

A couple of research gaps were identified in the literature and presented in this Section with its associated research objectives.

• Much of the literature reviewed assessed precipitation impacts arising from greenhouse gas emissions on annual or seasonal basis in the long-term until 2100. However, because the climate system is already committed to a certain degree of greenhouse gas emissions in the atmosphere, this will always have an impact on precipitation delivery and therefore, quite important to also consider the impacts of these emissions in the near-to-medium term future (2050).

Objective 1: Understand the precipitation regime in Eastern England for the period 1971-2000 and assess how precipitation patterns might change between 1971-2000 and 2021-2050) using a range of state-of-the-art CMIP5 climate model simulations under the medium range (RCP45) and high end (RCP85) greenhouse gas emission scenarios.

• Past and present studies of precipitation impacts on crops in Eastern England have applied different climate and crop models for assessments but have not used the new CMIP5 climate model ensembles to inform a greenhouse crop experiment.

Objective 2: Examine the precipitation characteristics of the study area in Eastern England and using the CMIP5 climate models to evaluate changes in precipitation between the two time

frames (stated above) and assessing how the changes will impact on future sugar beet yield.

Chapter 3: Methods, Data and Materials

3.1 Introduction

This chapter outlines and appraises the methods employed in this study and it is divided into two distinct Sections: the precipitation analysis methods and data (Sections 3.2 - 3.7) and the greenhouse experiment materials and method (Sections 3.8 - 3.10). The research aim was to firstly determine the present day precipitation regime in Eastern England using weather station observations for the period 1971-2000 and then, to determine the likely changes in precipitation over the same area using a range of climate models. Two time slices were selected for the climate model analysis: one for a historical period (1971-2000); and the other for the projected future period (2021-2050). The focus was to define the rainfall attributes of the present environment using observational data and to investigate the near-to-medium term future using a range of climate models. Secondly, the study investigated the impact of the identified precipitation changes on sugar beet plants in a greenhouse experiment. 150 sugar beet plants were studied in the first season (2014) and 201 the second season (2015) of the experiment. All the experiments were conducted in the greenhouse of Brunel University London located on the global grid as the area between 51° 31'58" N and 0° 28'22" W.

In the greenhouse experiment, past precipitation data from the climate model historical phase was compared with climate model projections in the same area and the precipitation changes between the two was used in calculating the watering regimes for the plant experiments. Much of the previous work on past precipitation as discussed in the literature review (Chapter 2) focused on changes in annual precipitation and trends in seasonal mean for detecting long-term changes in precipitation. This is a useful approach because such changes are usually evident in the seasonal mean due to the sub-seasonal variations being averaged out, thereby leaving out the yearly variations. However, none of the reviewed studies examined changes in precipitation using the new suite of models from the 5th Phase of the Coupled Model Intercomparison Project (CMIP5, Taylor et al. 2012) to inform a greenhouse experiment. This thesis explores the viability of two different methods of investigating precipitation changes based on CMIP5 data. The central thinking here was to use the climate projections to detect long-term changes in precipitation in a
way that mimics a realistic future precipitation scenario for the sugar beet growing season in Eastern England.

The two approaches used for analysing precipitation changes were: long-term growing season (May-October) daily mean precipitation which was used to investigate the impact of climatological changes in precipitation; and the "realistic" distribution of precipitation event sizes and frequency which was used to investigate the potential impact of precipitation characteristic changes in the area. Section 3.2 discusses the weather station observations used in the work while Section 3.3 focuses on the CMIP5 models used in the study. Section 3.4 examines the models and then, precipitation analyses were conducted to ascertain which models best replicated observed precipitation characteristics in the area of study, which then provided the basis for the suite of models to be used for analyses of future precipitation projections. It is important to note that due to the discontinuous nature of precipitation, it is useful to have a continuous homogenous precipitation dataset. This is dealt with in Section 3.5 with the study heavily reliant on the Met Office Integrated Data Archive System (MIDAS) dataset that collates UK daily climate and weather observations data (Met Office 2012). The models used for this study were also used in the most recent IPCC AR5 report which were from the CMIP5 project that included simulations using different Representative Concentration Pathways (RCP) (van Vuuren et al. 2011). Section 3.6 describes the watering regimes used in the plant experiment and how it was derived and used for the plant experiment. Section 3.7 describes the statistical analysis used in this study and Sections 3.8 - 3.10 describes the greenhouse experiment carried out in this study. These Sections further describe the crop criteria for this research explaining why sugar beet was chosen for the study and assesses the impacts of precipitation changes on the plants by measuring certain plants parameters over the growing season and yield of the plants upon harvest.

3.2 Precipitation observations

3.2.1 Weather stations observations dataset

This Section describes the precipitation data used in this study. To understand the present day regime, daily precipitation data that covered the area of Eastern England between 51° 35'N - 52°

75'N and 0° 29'E – 1° 35'E (see Figure 3.1) were extracted from the MIDAS dataset Met Office (2012) for this purpose. The observational dataset examined the period from 1971-2000 herein referred to as the 'historical' period – this is the period over which the CMIP5 models were tested. There are at least 6 weather stations that have operated in or near the sugar beet producing areas of East Anglia in Eastern England for over 30 years with continuous precipitation data and have little or no missing data. The selected weather stations are: Terrington St Clement (2 m asl, 0.29 ° E, 52.745 ° N); Santon Downham (6 m asl, 0.675 ° E, 52.458 ° N); Coltishall (17 m asl, 1.356 ° E, 52.756 ° N); Writtle (32 m asl, 0.432°E, 51.733°N); Manston (44 m asl, 1.35°E, 51.35 N); and Stansted Mountfichet (70 m asl, 0.184°E, 51.897°N). (See Figure 3.1 for the locations of these weather stations and corresponding Table 3.1 for the length of their records).



Figure 3. 1: Map of the study area showing the locations of the weather stations examined in the analysis. The dashed line indicates the area covered by MOHC HadGEM2-ES model grid cell used here. (See 3.2.1 and Table 3.1).

Region	Station Name	Elevation	Latitude	Longitude	Start Date	End Date
CEE	Coltishall	17	52.756	1.356	1963	2009
SEE	Manston	44	51.35	1.35	1962	2009
CEE	Santon Downham	6	52.458	0.675	1960	2008
SEE	Stansted Mountfichet	70	51.897	0.184	1961	2009
CEE	Terrington St Clement	2	52.745	0.29	1961	2009
SEE	Writtle	32	51.733	0.432	1971	2000

Table 3. 1: Details of the weather stations examined showing the length of their records.

These weather station datasets were extracted from the MIDAS dataset (Met Office 2012) via the British Atmospheric Data Centre (BADC). The MIDAS dataset contains surface observations over land areas of the UK and other world-wide stations, and have been quality controlled by the BADC to integrate data from advanced observing systems to support improved forecasting of model performance. The dataset compiles daily and hourly weather measurements including wind parameters, maximum and minimum temperatures, soil temperature, radiation and sunshine duration measurements as well as daily and hourly precipitation measurements. The scope of the current study is limited to precipitation and only considered the key sugar beet growing period in England (i.e. May – October).

3.3 Historical CMIP5 model precipitation data

Climate models constitute the main basis for evaluating and projecting climate change under different anthropogenic forcing scenarios. General Circulation Models (GCMs) and subsequent generation of models have provided science with reasonable level of confidence in simulating future precipitation and climatic changes based on the model's abilities to reproduce observed changes in past and present climates (Wang et al. 2016; Giorgi et al. 2004). In this Section, "historical" CMIP5 model precipitation data were evaluated using Earth System Models (ESMs). These models are mathematical simulations of the climate system, which includes the Earth's major biogeochemical processes. They have provided a relatively high level of confidence in simulating climatic changes and can be easily verified and validated by comparison with simulations of past observations.

The gridded data included longitude, latitude; and the date. The data provided a benchmark

period that allows the model data to be compared with observations. Furthermore, the model evaluation was also conducted to establish the relationship between model data, which represents an area and the precipitation data from individual stations. Averages from the GCM data were compared to the stations averages and statistically analysed at 95% confidence interval to provide an insight into the differences and /or similarities in the two dataset.

The individual members of the ensemble were initiated using slightly different, though equally realistic, initial conditions in order to capture some element of internal climate variability. The use of ensemble model was aimed at understanding past and future climates in vital areas of uncertainty (Taylor et al. 2012). The simulation of the "historical" experiment helped ascertain model fidelity and reliability in reproducing current climate as a guide to the model's reliability of future climate. The historical run were driven by realistic anthropogenic and natural forcing which includes greenhouse gases, aerosols, changes in solar output and ozone among others.

The main variable analysed in this study is precipitation. The points of interest from the utilised models that met these criteria are detailed in Table 3.2. Data from these models will be retained for further analysis depending on how closely they replicate precipitation observations for the region (see Chapter 4). The dataset for the two scenarios was used to perform two different CMIP5 analyses with the aim of assessing different levels of greenhouse gas emissions (GHG) over the 21st century using the Representative Concentration Pathways (RCPs; van Vuuren et al. 2011). This study examined future GHG emissions represented by mid-range (RCP45) and highend (RCP85) impacts on radiative forcing for the period 2021-2050. This temporal window was chosen as it is not too far into the future and it is, therefore, of interest to the sugar industry for future planning (British Sugar 2011).

				Ensemble Size		
Climate Model Names	Resolution (Latitude & Longitude)	Institution	Grid Cells	Historical RCP45	RCP85	
CanESM2	64 X 128 (2.8 X 2.8)	Canadian Centre for Climate Modelling and Analysis (CCCma), Canada	1,50	5	5	
CSIRO-Mk3.6.0	96 X192 (1.875 X 1.875)	Commonwealth Scientific and Industrial Research Organisation (CSIRO), Australia	1,75	10 1	.0 1	
HadGEM2-ES	145 X 192 (1.25 X 1.875)	Met Office Hadley Centre (MOHC), UK	1,114	4	4	
EC-Earth ESM	160 X 320 (1.125 X 1.125)	EC-Earth consortium; managed by Irish Centre for High-End Computing (ICHEC)	1,125	5	5	
IPSL-CM5A-LR	96 X 96 (1.875 X 3.75)	Istitut Pierre-Simon Laplace (IPSL), France	1,75	6	4	
MIROC5	128 X 256 (1.41 X 1.41)	Atmospheric and Ocean Research Institute, Japan	1,100	3	3	
MPI-ESM-LR	96 X 192 (1.875 X 1.875)	Max Planck Institute for Meteorology (MPI-M), Germany	1,75	3	3	
CCSM4	192 X 288 (0.94 X 1.25)	National Centre for Atmospheric Research (NCAR), USA	1,150	3	6	

 Table 3. 2: Details of the CMIP5 models examined.

Model inclusion in this study avoided the use of less realistic models for the study area. According to McSweeney et al. (2014), not all models are suitable for all regions. They assessed some CMIP5 models to identify their suitability for use across multiple regions through dynamical downscaling. Their study showed that certain models are more suitable than others depending on the region of use and specifically identified the MOHC (HadGEM2-ES) as very reliable in replicating past observations for Eastern England and the UK in general. Thus, the HadGEM2-ES was prominently used in this thesis for the analysis of changes in precipitation.

3.4 Overview of climate models

In order to obtain the optimum analysis of future climate projections, it is important to use data from the weather stations observations in the area of study for past and present climate to validate the use climate models. The observed data will also provide a benchmark period that will allow model data to be compared with observations over a 30 year period for the presence or lack of trend in the temporal window from 1971-2000. Observed data from the weather stations were the basis of the information and was cross-validated with the CMIP5 historic run used in this study to ascertain model reliability and performance (Mitchell & Jones 2005). The ability of climate models to reproduce observed changes provides the best examination of models performance rather than simulation of present day climate or intercomparison between models which tend to underestimate the uncertainty if the models were similar compared to the real world (Raisanen & Joelsson 2001). Although, much work have been carried out in the past on the reliability and limitations of climate models, it was still necessary to discuss their suitability for this research.

The climate models used in this study were all sourced from the CMIP5 project and were not only used to assess past changes in precipitation, but also used to examine projected changes in future precipitation under two different greenhouse gas emission scenarios, namely: the Representative Concentration Pathways (RCPs) (RCP45 and RCP85). Two of the RCPs were chosen for analysis here: RCP45 represents the mid-range greenhouse gas emission scenarios and the RCP85 represents the high-end greenhouse gas emission scenarios which according to Friedlingstein et al. (2014) is consistent with a 'worst case scenario'. Confidence in the use of climate models comes from their ability to simulate key aspects of the climate system (IPCC 2013b). For quality assurance, models are continually collated with atmosphere, ocean, cryosphere and land surface observations and one key aspect is their ability to imitate real climate system which can be verified from past observations and present climate.

Past studies on various Model Intercomparison Projects (MIPs) had simulated climate using individual models which can lead to unreliable and uncertain forecast as a result of statistical bias and structural errors. However, with advancement and improvements in climate models, the use of such MIPs and the Atmospheric Model Intercomparison Project, (AMIPs) (Gates et al. 1999) and the Coupled Model Intercomparison Project, (CMIP) have made it easier to compare models for differences and similarities (Covey et al. 2003; Meehl et al. 2000b). According to (Raisanen & Joelsson 2001), MIPs specific to individual models have enabled modellers to easily identify errors in models and areas that need attention.

In contrast to individual models, the use of Multi-Model Ensembles (MMEs) to examine structural uncertainty and internal variability have improved forecasting by predicting the distribution of possible outcomes (Flowerdew 2012). Ensembles can provide extra samples and average over errors specific to individual models. The multi-model ensemble forecasts are weighted averages arising from individual models, and the sum of all weights is equal to one based on their predictions. This study used a multi-model ensemble mean to provide a realistic projection from the models. In addition, models used in this study were selected based on meeting specific criteria described in Section 3.4.1.

3.4.1 Model criteria

In spite of the positives brought about by the use of climate models in climate forecasts and impact assessment of crop yields, there are still challenges in their uses as a result of, for example, different parameterization of chemical, physical and biological components of the climate system which can result in different simulations with varying degrees of values. CMIP5 comprises of models with different levels of complexity and with different spatial resolutions which, in some cases, alters how cloud processes are represented on a smaller scale resolution than the models. Therefore, to improve forecast, this study employed the use of MMEs which gives credence to

the study of Tubiello & Ewert (2002) not to depend on a single model for climate change assessment studies, but to use several model predictions in developing assessment for impact studies on crop yields.

The selection of climate models in this study was based on certain criteria (e.g. the dataset must cover the two temporal time frames adopted for the study – the "historical" period 1971-2000 and the "projected" period 2021-2050). The criteria used will assess the suitability and applicability of models to reproduce observed precipitation over Eastern England for a 30-year period (1971-2000). This will provide validity and justification for the use of selected models for assessment of future precipitation projections. It is assumed that the models that simulate observed precipitation well in the study area will better represent projections of future precipitation over the region.

In this thesis, data was sourced from the latest suite of the Coupled Model Intercomparison Project (CMIP5, Taylor et al. 2012) with models fulfilling the following criteria for selected assessment:

- The models must contain daily precipitation data for the study periods
- That the models are run as ensembles
- They are Earth System Model (ESM) or Coupled General Circulation Model (CGCM) and
- That the models contain historical and Representative Concentration Pathways (RCPs) records for the medium (RCP45) and high greenhouse gas emissions (RCP85)

3.5 CMIP5 precipitation projections

3.5.1 Introduction

In order to determine the likely climatic future changes over Eastern England, projected precipitation data from a range of CMIP5 climate models were examined. Projection data were extracted from the fifth phase of the Coupled Model Intercomparison Project (CMIP5, Taylor et al. 2012). Each of the experiments utilised a multi-member ensemble for the model run whereby each individual member of the ensemble were initiated using slightly different but equally

realistic initial conditions. The purpose of this is to enable the model capture some elements of internal climate variability as reported by Taylor et al. (2012). The models used in "historical" and "future" parts of the study were the same. All the models were selected base on the criteria contained in Section 3.4.1 and described in Table 3.2.

3.5.2 Climate change and greenhouse gas emissions scenarios

In order to properly evaluate the influence of future greenhouse gas emissions on climate and the corresponding socio-economic development, a range of future greenhouse gas emission scenarios were developed called the Representative Concentration Pathways (RCPs) (Meinshausen et al. 2011). The RCPs are categorised and named according to their radiative forcing target by 2100 with estimates based on forcing of greenhouse gas emissions scenarios. These RCPs were used in the last IPCC report (AR5, Fifth Assessment Report) and the associated cycle of the fifth phase of CMIP5 for projected climate change run.

There are four RCPs developed and they include: One mitigation scenario leading to a very low forcing level (RCP26), two medium stabilisation scenarios (RCP45 and RCP 6) and one very high end emissions scenario (RCP85). In the light of these four scenarios, future climate projections can be assessed from different models accessible under the CMIP5 project. It is acknowledged that the CMIP5 data has higher spatial resolution for regional analysis (Sperber et al. 2013).

In this thesis, RCP45 and RCP85 were employed and the choice of this two was based on the premise that RCP45 is the one that society is currently closest to and the RCP85 was based on current and future projections of increasing greenhouse gas emissions in the atmosphere. Moreover, because of the high end emissions of RCP85, it provides a much stronger climate signal and easier identification compared to the lower emissions and it gives an indication of the "worst case scenario". These scenarios may give us an indication of what possible climates to expect in future and how it may affect sugar beet productivity and agriculture in general in the UK. Table 3.3 shows details of the RCPs analysed.

Scenarios	Description
RCP45	Stabilized radiative forcing without overshoot pathway
	to 4.5 W/m ² (~850ppm CO ₂ eq at stabilization after 2100
RCP85	Rising levels of radiative forcing pathway leading to8.5
	W/m ² (~1370ppm CO ₂ eq by 2100

 Table 3. 3: An overview of the two Representative Concentration Pathways (RCPs) used in this thesis (Moss et al. 2010; Meinshausen et al. 2011).

3.5.3 Projections of changes in future precipitation

Global climate models (GCMs) constitute the main basis for evaluation and projections of climate change under different anthropogenic forcing. The current study examined eight CMIP5 climate models which all implemented the "historical" run experiment (see Section 3.3) with each of the experiments utilising a multi-model ensemble for the run. The models used in this section for projections of changes in future precipitation were based on how closely they replicated precipitation observations for the region (see Chapter 4).

In order to gain an understanding of possible future changes in precipitation, it was useful to first gain an indication of the overall changes in mean daily precipitation over the area of interest. Therefore, the mean and total daily precipitation in the model runs were analysed for both the "historical" (1971-2000) and "future" (2021-2050) time windows. Secondly, the study examined the differences between the changes in the characteristics of the daily precipitation between the "historical" and "future" precipitation.

The mean daily values of each ensemble of model runs from the daily precipitation analysis of the "future" experimental period 2021-2050 were quantitatively compared with the mean daily values of precipitation from the "historical" experimental period 1971-2000. The aim of using the mean values from both datasets is to give as close as possible estimate to the "true" areal values in the study region. This method was done by averaging all the values in each model for both the "historical" and "future" and then using statistical analysis to detect for differences between the two time frames. For all statistical analysis of relationships between the datasets, days with precipitation less than 0.05mm were removed from the calculations (see Section 3.6.3.1 for details) due to the potential of those days causing bias in the results.

3.6 Calculation of watering regimes

3.6.1 Introduction

This Section describes the calculation of the watering regimes (irrigation) used in the greenhouse crop experiments over two growing seasons and it is divided into two Sections: the "seasonal" mean watering regimes used in the first year of experiment (2014) and the "realistic" distribution of watering regimes used in the second year of the experiment (2015). The irrigation water need represents the difference between crop water requirements and the effective precipitation (Allen et al. 1998). The watering regimes employed in this study as well as the changes in event size and frequency allows for the effect of climatological or mean changes in precipitation to be investigated.

3.6.2 Calculation of the climatological watering regimes

A key aim of this thesis was to investigate changes in precipitation from climate model projections. In this Section, the study investigated the impact of seasonal precipitation changes on the greenhouse crop experiment using climate model projections. Considering this, the watering regimes in the first season were not designed to replicate realistic precipitation events but to deliver the total growing season (i.e. May-October) precipitation in a series of regular and equal watering events. This method entailed removing the random distribution periods of low and high precipitation.

Within the climatological experiment, all the plants in the experiment were watered every other day (i.e. watering day – dry day – watering day – dry day and so on) with the same amount of water per watering day for each regime. Two watering regimes were implemented; the "control", or "present day", experiment where precipitation observations for the period 1971-2000 and the recommended level of water for a successful sugar beet crop from the Food and Agricultural Organisation of the United Nations were used to calculate the watering event size (FAO, Brouwer & Heibloem 1986). Secondly, there was a "future" watering regime, which was based on a modification of the "control" experiment watering event size determined by the growing season (i.e. May-October) changes in precipitation from the climate projections for the period

2021-2050.

Studies by Carlson & Bauder (n.d.) in the US also reported that sugar beet requires approximately 22-28 inches of rainfall during the growing season which is equivalent to 558-711 mm. The FAO minimum and maximum recommended seasonal crop water need per sugar beet plant reveal that sugar beet requires 550-750 mm of rainfall per season (FAO n.d.); and also requires a minimum and maximum growing period of 160-230 days. Therefore, to calculate the average water per day for the plants, the lowest watering range was divided by the lowest growing range; the highest watering range divided by the highest growing range, and the highest watering range divided by the lowest growing range. The output from the highest watering range and the highest growing range (3.26 mm) was multiplied by the surface area of the pot used to grow the beets to obtain the water per day for the plants which was then converted to litres per day to conform to the measuring can in use for the watering.

The values obtained in this study are comparable to the mean daily May-October precipitation for the region from the observations. The highest daily mean May-October precipitation from the stations ranged from a low of 432 mm to a high of 522 mm. Likewise, the total May-October precipitation from the stations ranged from a low of 664 mm to a high of 725 mm compared with 550 mm-750 mm by the FAO.

3.6.3 Calculation of realistic precipitation distribution of events and sizes

Calculation of the watering regimes in the second season (2015) was based on the changes in the monthly distribution of precipitation events and sizes. Similar to the first season experiment, it was designed to examine the impacts of changes in precipitation. In the second year, the experimental focus was designed to replicate as close as possible future realistic precipitation changes based on monthly events and sizes during the growing season (i.e. May-October). This method entailed working out the total number of wet days and dry days to determine the number of watering events in a given month. This was done by calculating the mean number of precipitation observations in each month and then distributing the watering events in a given month bearing in mind that the different months in the growing season have different number of days

(see Appendix 1).

Then, the size of the watering events was worked out by splitting the observed precipitation distribution into percentiles and then determining watering events that replicate that distribution (see Appendix 2-4). The future wet days were distributed over the month taking into consideration the probability between wet and dry days (see Appendix 5). Studies Geng et al. (1986) have used similar approach to assess daily rainfall and its impacts on agriculture while Waha et al. (2012) used past rainfall deterministically to determine sowing dates.

3.6.3.1 Selection of wet days

The occurrence, intensity and amount of precipitation on any given day varies in time and space and is expected to change over time due to climate change (Trenberth 1999). Trenberth (2011) also reported that climatic changes will alter several direct processes involved in the amount of daily precipitation distribution. This is an important factor as the rate of precipitation accumulation over a day varies from day to day and from one location to another. Studies such as Sun et al. (2006); Dai (2006); Osborn & Hulme (1998) have examined daily precipitation characteristics in GCM simulations and all arrived at similar conclusions that models tend to overestimate light or small precipitation events while underestimating heavy precipitation events in comparison to observations. Alexander & Jones (2001) showed that there is an equal chance of accurate precipitation events in England and Wales, however, in some regions such as South-Eastern England, models can be less reliable in determining dry events (e.g. events < 1 mm day-¹).

This bias towards light precipitation events has been attributed to problems with parametrization Kharin & Zwiers (2005), while Kopparla et al. (2013); Chen & Knutson (2008) reportedly identified spatial resolution as a contributing factor to the disparity between models and observational datasets. With these in mind, it was therefore necessary to remove small precipitation events from the precipitation analysis in this study in order to have better consistency and higher correlations.

Furthermore, to improve monitoring and prediction, precipitation analysis requires accurate

estimates and careful consideration of the input data. Daily gridded observations and model data corresponds to a spatial average in space rather than a single point thereby obscuring days of no rain and creating a bias in the analysis (Pendergrass & Hartmann 2014). It is therefore important to select a wet-day threshold in order to by-pass this issue. Some studies have used guidelines for wet day precipitation as having accumulation of 1 mm (Klein Tank et al. 2009; Sun et al. 2006). Chou et al. (2012) in their study used a dry day threshold of 0.1 mm day-¹. Different studies have used different guidelines and in this thesis, 0.05 mm day-¹ was adopted as it was considered low enough to reflect only minimal amount of precipitation lost but also high enough to allow the level of precipitation accumulation in this analysis such that any value >0.05 mm day-¹ was considered a wet day and values <0.0 mm day-¹ was considered a non-precipitation (dry) day.

3.7 Methods of statistical analysis

Statistical analysis in this study was subjected to a check for normality of distribution. The normality test informed on what type of test followed. Where data was normal, a parametric test was conducted and where data was skewed, a non-parametric test was carried out. Compatibility in the experiment was assessed using the null hypothesis. The null hypothesis assumes that there is no difference between the samples under test. Therefore, by applying a 95% (0.05%) confidence level, the p-value gives us an indication if there is a significant difference in the samples. A lower p-value from 0.05% in the observed sample indicates a significant difference from the hypothesised sample.

3.7.1 Independent student's t-test

The Student's t-test for equal and unequal variances was used to find the significant difference between two samples, for example, the historical CMIP5 precipitation data was compared to the future (projected) CMIP5 precipitation data. The null hypothesis was set at no effect between the historical and future samples and an alternative hypothesis of 0.05%.

 $H_0: \mu 1 \le \mu 2$ $H_1: \mu 1 > \mu 2$

Where $\mu 1$ and $\mu 2$ are the means of the historical and the future samples; the p-value in the analysis then gives an indication of the existence of a significant difference between the two samples (Spintali, 2011; Kim, 2015). If the p-value is less than 0.05, then it can be inferred that that there is a significant difference between the samples making the null hypothesis untenable, therefore reject H₀ and accept the alternative hypothesis (H₁).

3.7.2 Pearson's correlation coefficient (CC)

This test was used to evaluate similarities or differences between the observed CMIP5 precipitation and model CMIP5 precipitation data (Sedgwick, 2012). It is described as the covariance of the observed and the model means divided by their standard deviation. The CC is denoted with the letter "r" where r= 1, it suggests a positive linear relationship exists between the two datasets; where r = -1, it implies a negative correlation between the datasets; and where r = 0, it implies no relationship exists between the observed and model datasets.

3.7.3 Analysis of variance (ANOVA) and the Tukey post hoc test

ANOVA single factor comparison test was carried out to compare the means of the three different CMIP5 historical, RCP45 and RCP85 precipitation data and crop yields. This test helps to determine whether there are any significant differences between the means of the datasets (Kim, 2014). The test hypothesis assumes that there is no significant difference in the means of the three datasets and an alternative hypothesis that there should be at least one difference between the means from the three datasets. After ANOVA had been conducted, the Tukey multiple comparisons test was used to identify which of the sampled means that is significantly different from each other (Kim, 2014). These calculations were based on a mean statistics and normality of data at 95% confidence interval.

3.8 The greenhouse experiments

3.8.1 Description of the greenhouse

The greenhouse plant experiment was carried out using sugar beet plants that were grown in individual pots in the greenhouse facility located on the grounds of Brunel University London campus. The greenhouse measures 19m x 8m on the inside and contained plant tables of different lengths and breadths. The basic structural design of the greenhouse includes end walls, side post, 2 feet side walls to the side of the building onto which a semi-circular hoop shape plastic PVC sheets are built on. The sheets are covered all the way to the roof top with a single layer of polycarbonate plastic covering. The polycarbonate plastic sheets are transparent allowing sunlight and radiation spread more evenly throughout the facility. There is a front door to the facility and an emergency exit door at the back of the building. The greenhouse also has a water hose connected to the main water supply for watering and irrigation purposes. Overall, the greenhouse was considered an ideal environment for the experiments as it allowed water to be controlled. Temperature and humidity were not controllable in the greenhouse and was therefore not assessed in this study. It is worth noting that the different variables were consistent for the different watering regimes and plant experimental groups.



Figure 3. 2: Panoramic view of plants inside the greenhouse.

3.8.2 Selection of seed

The ultimate agricultural goal for farmers and crop growers in agricultural productions is the yield of the crop. Crop yields, apart from weather and growing conditions are influenced by the chosen variety of seed to sow. Sugar beet have different seed varieties and some of them perform better than others in terms of fresh weight yield and sugar percentage (BBRO, 2013). Sugar beet farmers in selecting sugar beet seeds for sowing considers a combination of factors including yield, sugar content, resistance to diseases and good commercial track record. This makes the process of seed selection a very important aspect of sugar beet production (BBRO, 2013). So, for this experiment, uniform sugar beet seeds of the same variety (SY Muse) was supplied free of charge by Sygenta UK. The seeds were pelleted (coated seeds) and were used for all replicates in the experiment.

The pelleted seeds were coated with a round protective and nutritive casing which ensures uniformity of the seeds makes it easier to handle during sowing and increases the growing conditions of the seeds. Another important factor in using pelleted seeds is because seeds are affected by diseases which reduces yield. One way of controlling seed diseases is by coating the seeds before planting. This technique involves adding several materials such as fertilizers, nutritional materials, moisture attractive agents, pesticides and plant growth regulators to the seeds. This technique has proved beneficial to growers for improving yield of crops (Ehsanfar & Modarres-Sanavy 2005).



Figure 3. 3: Picture of sugar beet seeds used in this study

According to the British Beet Research Organisation, BBRO (2013), the fresh weight of SY Muse is a high yielding variety that performs consistently well with excellent establishment and resistant to drought and rhizomania, and it's widely used by UK farmers. SY Muse compares favourably well with other varieties (see Table 3.4) and is third on the official yield variety list of the BBRO (2013) in terms of root yield and sugar content. It is rated "3" and "4" on a scale of 1-9, with 1 being "susceptible" and 9 being "tolerant" on the BBRO (2013) rust and powdery mildew disease scales respectively.

Table 3. 4: Shows a comparison	of the SY Muse with	other sugar beet	varieties. Source:	BBRO (2014).
--------------------------------	---------------------	------------------	--------------------	--------------

Mean of sugar beet variety ranked in sugar yield order											
Varieties	Sugar t/ha	Adjusted tonnes t/ha	Root yield t/ha	Sugar content (%)	Plant establishment (%)	Rust scale	Powdery M.D scale	Year listed			
Hornet	101.9	102	101.5	18.5	101	6	1	2014			
Haydn	101.6	101.7	100.8	18.5	101	5	5	2013			
SY Muse	100	99.9	100.6	18.3	101	3	4	2012			
Master	99.7	99.8	99.2	18.5	102	8	5	2013			
Aimanta	98.9	98.7	100.2	18.1	98	3	3	2009			

As SY Muse is not on the extreme end of these scales, it was considered further justification for its use in this experiment.

3.8.3 Compost

The planting medium used for the plant experiments was the John Innes No. 2 soil based compost. All the compost used for the experiment were of the same brand but different bag sizes of 20kg and 50kg, and was sourced and supplied by LBS Horticultural Ltd. Lancashire, BB8 7BW, UK.

The compost contained ingredients of loamy soil which is the dominant soil type in the sugar beet producing areas of Eastern England (Scott & Jaggard 2000). This soil type is compact and retains moisture after draining off excess water and by so doing encourages germination and good plant establishment under field or controlled conditions. The soil also provides natural reservoir of plant food, trace elements and contains some organic matters which releases nitrogen slowly to the plants. The loam in John Innes No. 2 is screened and sterilised to avoid any transfer of soil-borne diseases and insects. The spaghum moss peat in the compost improves the aeration and water retaining capacity which enables the plants to make maximum use of nutrients. The lime-free grit and sand included in the soil allows excess water to drain from the plant pots and prevent water logging whilst adding weight and stability to the pot. The horticultural grade lime in the soil provides the pH that encourages healthy growth in plants. These characteristics are important for healthy plant growth, especially under controlled conditions.

3.8.4 Plant pots

A total of 150 plant pots were used for the experiment in the first season while 201 plant pots were used in the second season. The plant pots were black in colour and were all of the same size: 33cm in depth, 35cm in diameter at the top and 27cm in diameter at the bottom and had a volume of 33L. The pots were designed with holes at the bottom to drain off excess water from the pots as shown in Figure 3.4. The holes at the bottom of the pots ensured that there was no water logging in the pots while the soil type helps to retain moisture. These combined characteristics encouraged healthy plant development. The depth of the pots was also deep enough for sugar beet roots to grow unhindered as observed over the growing season.



Figure 3. 4: Picture of the plant pots used for the experiments with drainage holes at the bottom to drain off excess water from the pots.

3.8.5 Sowing dates

Sowing dates effectively means the dates that seeds are planted depending on soil factors and environmental conditions of the area where the crop is grown. In agricultural calendar, planting dates are considered one of the most important factors influencing field crops as it plays a vital role in germination, growth and yield of sugar beet and crops in general. For sugar beet, planting dates are also important in the sugar factory calendar as it enables organisations in planning of their work schedule. Thus, sowing sugar beet on suitable planting dates under the environmental conditions of the region is the best method to ensure adequate comparison of yields. In this study, 300 sugar beet seeds were sown into 150 plastic 33L plant pots in the first season on the 15 April 2014 and 402 seeds were sown in the same manner in the second season of the study on the 25 May 2015.

3.8.6 The plant experiment

The plant experiment in the first season commenced on the 15th April 2014 and on the 25th May 2015 in the second season. Two sugar beet seeds were sown into each plant pot containing 30 kg of "John Innes No. 2" compost which was used as a planting medium. The soil in the pot was shaken to eliminate pockets of air in the soil and keep the soil level and compact. This enabled the soil to retain moisture after draining off excess water. The timing of watering is important to maximise yields and ensure a fair comparison between the watering regimes. Water was applied in the mornings when the plants can maximise the available water because of lower evapotranspiration. A watering procedure was used that ensured the water was added in a consistent way to all pots and was as uniform as possible around the surface area of the soil.

This method was successful in terms of germination: 298 seedlings out of the 300 seeds sown in the first season and 360 emerged out of 402 seeds sown in the second season. Plant seedlings were thinned at their 4-6 leaves growth stages from two to one seedling per pot to encourage uniform establishment. Thereafter, in their 10-12 leaf growth stages, the plants were classified into the watering regimes as described in Sections 3.6.2 and 3.6.3: the "control" and "future" watering regimes in the first season, and the "control", "RCP45" (denoted as M) and "RCP85" (denoted as H) watering regimes in the second season.

Moreover, plants in the different groups were systematically arranged (See Figures 3.5 and 3.6) around the greenhouse so that there would be minimal bias in temperature, humidity or sunlight for any group. In light of the foregoing, the greenhouse was considered an ideal environment to conduct the plant experiments. Furthermore, categorisation of plants at this time was done to coincide with the precipitation analysis from May-October for the study periods and because the biggest changes in precipitation was projected for the summer.



Figure 3. 5: Illustration of how plants in the control (C) and future (F) categories were arranged inside the greenhouse in the first season of the study.

3.8.7 Method of plants watering (Irrigation)

In both years of the experiment, plant pots in each watering regime were colour coded and the measuring cylinders used to add water to the pots were also colour coded to match each watering regime so that the potential for human error was reduced to a minimum. Each of the plants in the different categories was assigned a number so that growth and yield parameters could be recorded for specific plants.

3.8.7.1 Irrigation under "climatological" watering regimes

In the first year of the experiment (2014) under the climatological watering regime, plants were categorised into "control" and "future" groups and were watered (irrigated) every other day whereby irrigation was administered on a day on - day off basis. Plants were given the same amount of water from the sowing stage until they were categorised into different watering regimes. This gives all the plants an equal chance of developmental growth before allocating them into groups. Thereafter, plants in the "control" watering regime were watered according to the calculation described in Section 3.6.2 and the plants in the "future" watering regime were given 16% less water than the "control" based on the calculation of future precipitation projections. Assessment of the plants commenced after categorisation when they had started forming tubers to gauge the impact of different watering treatment on the plants in the two categories.

Plants were categorised into the control and future (See figure 3.6) watering regimes and the different watering regimes were implemented after the plants had reached their 10-12 leaves growth stages. In order to account for natural variability in plant sizes, the plants assigned to each watering regime were selected to result in an equal distribution of plant sizes in each watering experiment. Categorisation of the plants at this time was done to coincide with the precipitation analysis from May to October for the study periods and because the biggest changes in precipitation was projected for the summer. Output of changes calculated from the precipitation analysis was then imposed on plants in the future category.

Irrigation was applied by measuring the appropriate amount of water (i.e. 460ml for control and 390ml for future) into a 500ml cylinder; pouring it into a watering can and then applying the water on the plants through the surface of the soil. To do this effectively, a watering procedure was put in place to ensure that water was added in a consistent manner to all pots and was as uniform as possible around the surface area of the soil. This is important to ensure that water was accessed by plant roots regardless of their location inside the pots. Moreover, plant pots in each watering regime were colour coded along with the watering cylinders to ensure that the right plants get the appropriate amount of water and again reduce the potential of human error to a minimum. Each plant was assigned a number so that growth and yield parameters could be recorded for specific plants in the different experiments.

3.8.7.2 Irrigation under "realistic" distribution of precipitation events and sizes

In the second year of experiment (2015), irrigation was conducted in much the same way as the previous season except for the number of experimental categories and the calculation of the watering regimes which was based on a "realistic" distribution of precipitation events and sizes rather than a climatological representation of precipitation. All other materials used remain the same as in the first season experiment.

Plants were categorised into control, RCP45 and RCP85 watering regimes (See Figure 3.7) and were implemented after the plants had reached their 10-12 leaves growth stages. The "control" or "present day" experimental watering regime represents precipitation for the baseline period from 1971-2000; the "RCP45" in this experiment represents "medium greenhouse gas emissions

scenario" otherwise called "medium" for experimental purposes and the "RCP85" represents "high greenhouse gas emissions scenario" or "high" for this experiment and both RCPs represents the projected period from 2021-2050. The three experiments were designed to examine changes in absolute precipitation determined by event sizes and distribution of precipitation events over the growing season. As in the first year experiment, accounting for natural variability in plant sizes helped ensure that the plants assigned to each watering regime were selected to result in an equal distribution of plant sizes in each watering experiment. Categorisation of the plants at this time was also done to coincide with the rainfall analysis from May to October for the study.

The calculation of the watering regimes entailed working out the total number of wet days and dry days to find out how many days the plants will receive irrigation and the sizes and distribution of irrigation in each month. Therefore, under this regime, plants were only watered on wet days within the given month and left without water on dry days.

Irrigation was conducted by measuring the appropriate amount of water for each of the plants in the different categories into a 500ml cylinder; pouring it into a watering can and then applying the water on the plants through the surface of the soil. To do this effectively, a watering procedure was also put in place to ensure that water was added in a consistent manner to all pots and was as uniform as possible around the surface area of the soil. The timing of watering is vital to maximise yields and ensure a fair comparison between the watering regimes. Other watering applications such as watering in the mornings and colour coding of pots and cylinders followed pretty much the same pattern as already discussed in Section 3.8.6.1.

Н	С	Μ	С	Μ	Н	С	Μ
Μ	Н	С	Н	С	Μ	Н	С
С	Μ	Н	Μ	Н	С	Μ	Η

Figure 3. 6: Illustration of how plants in the control (C), RCP45 (M) and RCP85 (H) categories were arranged on the tables inside the greenhouse in the second season of the study.

3.8.8 Plant parameter measurements

3.8.8.1 Above-ground measurements

The measurement of plant parameters help to capture the overall health and the impacts of environmental conditions on the plants. The measurement of plant parameters in this study was an important aspect for comparing the impacts of differential watering regimes on the "control" and "future" categories of plant experiments. These measurements help to determine the extent of plant growth in each experiment based on the different watering regimes impacts on yield.

Plants in the different categories of the experiments have been exposed to the same conditions prior to categorisation. Upon categorisation, this enabled the measurement of important parameters and collection of data from the different stages of plants growth and development to be compared and statistically analysed in relation to yield. Essentially, a number of non-destructive parameters were used to assess the yield potential of the plants over the growing season including: the number of leaves on each plant, height of the plants (i.e. height of the tallest stem); the growth ratio of the stem (i.e. height divided by the number of stems) and leaf width (i.e. width of the widest leaf). The above-ground parameters were made with the use of a tape measure. These parameters were measured every two weeks to enable the assessment of water reduction on the plants development and productivity. This places the yield of the plants in the context of the growing season.



Figure 3. 7: The method used in measuring plants height and growth ratio index. A tape rule is placed at the crown of the plant to the highest tip of the tallest stem. The highest stem measurement is divided by the number of leaves on the plant to get an indication of the plants growth ratio.

3.8.8.2 Soil moisture

During the growing season, soil moisture measurements were undertaken every two weeks. The soil moisture measurement was conducted using a manual soil moisture meter (Brand: Lutron PMS-714) to measure the level of moisture in the soil. The soil moisture meter was manufactured by NHBS Ltd, Devon, TQ9 5LE, UK. The Lutron was used to measure the soil moisture content in individual plant pot and has a range of 0-50% with a resolution of 0.1. The microprocessor ensures high accuracy of data and the durable ABS plastic housing makes it ideal for use under field and controlled conditions. The LCD screen makes it easy to read data and it also has a data hold function which allows freezing of the current value on display thereby helping to minimise recording errors.

The probe was powered by 4 Duracell batteries and the prong is thrust vertically 10cm depth into the soil to obtain the soil moisture reading (see Figure 3.8). These readings were taken every two weeks from individual plants across all the experiments in similar manner.

The data was collected for each individual plant in the different categories to enable the assessment of the different watering regimes on soil moisture.



Figure 3. 8: Method of manual soil moisture reading.

3.8.8.3 Below-ground measurements

At the end of each experimental season, destructive measurements were taken to determine the mean mass of the tubers as harvested and when dried. When harvested, the tubers were uprooted from the soil, washed and dried. Thereafter, the leaves of the plants were cut off from the crown leaving the tubers, which are of most interest in this research. Each individual tuber was weighed without the leaves – these measurements are reported in this thesis as the "wet" weight. The

tubers were then labelled with their numbers and put in open transparent bags so that the yield data could be added to the database of growth parameters recorded over the growing season. Analysis of the dry weight of the tubers was conducted using a laboratory method to remove the moisture content. Obtaining the dry weight was done by cutting each tuber into smaller pieces to speed the drying rate. The size of the pieces was kept as equal as possible for all tubers so that drying rates were as equal as possible. A tuber of median size was cut into 8 pieces whereas larger tubers were cut into more pieces. The pieces were put into individual aluminium trays and numbered for identification purposes and then put inside an oven for drying at 80 °C until constancy, *as per* Mohammadzadeh & Hatamipour (2010). The cut tubers were weighed periodically, typically every 2 hours, until there was no more appreciable change in weight. At this point the value was recorded as the "dry" weight.

Measurements and data collected at different stages of the plants' growth and following harvest were statistically analysed to enable quantification of impacts. All measured parameters were tested for normality, which then determines the type of test to be carried out. Parametric tests were conducted where data was normal and non-parametric tests were conducted where data was skewed. Following this, a two tailed t-test was carried out for the two watering regimes. The outcome of the experiment was assessed using the null hypothesis: "there is no difference in the categories". Therefore, applying a confidence interval (CI) of 95% with alpha set at 0.05%; the p-value then gives an indication if significant differences exist in the parameters assessed.

3.9 Weeding and pest control

3.9.1 Weeds

The presence of weeds in sugar beet and plants generally can result in the crops competing directly for vital resources such as water. Weeds of any kind can result in yield losses and therefore adequate protection, attention and plans must be put in place to safeguard against potential losses.

Although, the incident of weeds in this experiment were minimal and not considered a threat to yield, measures were still put in place to safeguard against the threat of weeds by instituting a

twice-weekly programme of consistent weeding to ensure early detection of weed/s in any of the pots. This programme also helped to remove direct competition from the weeds and ensured that the plants received and utilised the amount of water allocated to each plant, thereby eliminating bias in the experiment as a result of competition from weeds.

3.9.2 Powdery Mildew Disease (PMD)

Powdery Mildew Disease is a fungal disease that affects the leaves of sugar beet and is prevalent in sugar beet producing areas such as the UK mainland, Europe and the United States, causing potential sugar beet yield losses of up to 30-35% (Neher & Gallian 1947). The fungus is mainly airborne and introduced annually around the end of June through July. Its lifecycle is linked to that of the host and the prevailing environmental conditions. Symptoms of PMD are usually first seen on older leaves on the underside. They appear as small, scattered, white powdery mycelium coating on the surface of the leaves preventing the leaves from performing the vital function of photosynthesis which can ultimately lead to yield reduction in cases of severe infection.

Control of PMD entails the use of fungicides to control attack. Early detection and intervention is important for effective control of the disease and the timing of control measures (i.e. treating the disease at the onset is more effective than when the disease has spread through the plant population). PMD attack on the plant experiments in this study was minimal, although, when the attack was observed in the first season of the experiment, it was treated with Systhane Fungus Fighter which was sprayed across the plant population to prevent infection of new leaves. The second season PMD attack was even more insignificant because the disease was detected early and control measure using Systhane Fungus Fighter was immediately applied to stop further infection.

3.9.3 Armyworms

Armyworms are pests that cause perforations on sugar beet leaves. These pests usually thrive in warm locations and their eggs are normally deposited on the surface of the leaves. In severe cases, armyworm attacks could damage leaves so badly that only the leaf veins and petioles will be left. If attack is severe, the larvae migrate from plant to plant via the leaves infesting new leaves and seriously damaging the plants ability to photosynthesise. The management and control of armyworms in this study was by early detection and manually removing the pests from the affected leaves and plants, and gently cleansing more affected plant leaves with mild soapy water. This management procedure removes the eggs and larvae from the affected plants/leaves and was very effective in treating the pest. This may not be the case in field operations or much larger indoor studies with very large population.

Chapter 4 – Results: precipitation analysis

4.1 Introduction

The results in this thesis cover two chapters: the precipitation analysis (Chapter 4) and the greenhouse experiments (Chapter 5). The analysis of precipitation is of fundamental importance to understanding the possible impacts of future changes in precipitation and how it will affect sugar beet farming in Eastern England. The current chapter presents findings of changes in precipitation over Eastern England for two different time frames: "historical", 1971-2000 and "future", 2021-2050 using daily precipitation data from the CMIP5 climate models. Model output was used as a basis for quantitative comparison of past and future precipitation analysis using the growing season mean and changes in day-to-day precipitation event size and frequency. The results relating to the precipitation analysis were used to calculate the watering regime (Section 4.5) that were applied to the greenhouse experiments presented in Chapter 5.

4.2 Analysis of weather station data

4.2.1 Weather stations May-October annual mean precipitation analysis

The climate of Eastern England is generally dry and mild and considered one of the driest regions in the UK with many areas receiving less than 700 mm of precipitation per year compared to other areas such as the Lake District with an annual average of 3000 mm and the western Scottish mountains with an annual average of about 4000 mm per year (Met Office 2016). Figure 4.1 provides a broad reflection of monthly mean rainfall variations in different parts of the UK and shows the rainfall output from two stations in Eastern England used in this study (a, b), two stations in southwest England (c, d) and two stations in northern Scotland (e, f). Comparison of the six stations shown in Figure 4.1 indicates that the two stations in Eastern England has the lowest amount of monthly mean rainfall of about 65 mm while the south-west of England has about 125 mm and northern Scotland with a high of around 145 mm (Met Office 2016).



Figure 4. 1: Monthly mean rainfall from six stations in three different regions of the UK. Two stations from the area of interest in Eastern England (a, b), two stations from the South West of England (c, f) and two stations from northern Scotland (e, f). Source: Met Office (2016).

The general nature of precipitation variability in time and space makes it important to rely on long-term climate analysis from weather stations with reliable homogenous data and in this thesis, the analyses presented cover a 30 year period (i.e. 1971-2000) using precipitation data from six weather stations with continuous and reliable precipitation data of Eastern England, in order to gain an understanding of the current precipitation regime in the region. Results of the weather stations May-October daily mean precipitation analysis over Eastern England showed

high annual variations depicted in Figure 4.2. The highest daily mean precipitation value from the six stations was recorded in Santon Downham for 2.9 mm in 1987 followed by the second highest of 2.82 mm recorded in 2000 at Stansted Mountfichet. Other high daily mean May-October precipitation were also recorded in the remaining four stations with some stations recording the same values, such as Writtle and Coltishall with 2.72 mm respectively followed by Manston with 2.63 mm and Coltishall with 2.44 mm. Conversely, low daily mean precipitation was recorded across all stations in 1976, 1978, 1995, 1996, 1997 and 1998. Figure 4.2 shows the pattern of annual variations displayed from the weather stations data.



Figure 4. 2: Observation plot of the May-October annual mean precipitation data from the six weather stations used.

The May-October daily mean precipitation was calculated using the May-October mean daily precipitation for each year in each station for the 30 year period. Over the 30 years, all the stations data exhibited similar year-to-year variations. The data also showed that in the years that precipitation was high, this applied to all the stations and conversely, in the years that precipitation was low, it also applied to all the stations. This suggests that similar precipitation mechanism prevailed over the region. Overall, statistical analysis of the data showed no significant trend in the observed precipitation. This was based on Pearson's product-moment correlation which showed a negative trend of -0.13, however, the trend was not significant with a p-value of 0.81.

Further mean statistical analysis of the May-October weather stations data using boxplots showed that the precipitation characteristics of the majority of the stations showed good agreement across the region. The May-October daily mean precipitation recorded for the six stations from 1971-2000 are shown in Figure 4.3 and Table 4.1 with the highest mean of 1.78 mm day-¹ recorded for Santon Downham, followed by Coltishall and Stansted Mountfichet with 1.76 mm day-¹ respectively, Terrington St Clement 1.66 mm day-¹, Writtle 1.61 mm day-¹, and Manston with 1.56 mm day-¹. Only the median and distribution of Manston, despite falling within the region of interest (see Figure 3.1) is not representative of the region where most of the sugar beet farming occurs. So, this is not considered important for the analysis and is therefore disregarded in the analysis presented here. However, the results of the mean daily precipitation in this study compares well with the findings of Gregory et al, (1991) who reported a daily mean precipitation of 1.77 mm per day for Eastern England (see Table 4.1).

Table 4.1 also showed small Standard Error of Mean (SE) values which indicated that the sampled mean is a good reflection of the actual mean population of the data. Similarly, the Correlation Variation (CV) also showed low values and in fact, smaller than the Standard Deviation (Std) which is to be expected, it revealed the amount of variability in the data relative to the mean. The CV showed the closeness of the distribution around the mean.

Stations	Mean	Standard Deviation	Median	Standard Error (mean)	Correlation of Variation (CV)	Observations
Coltishall	1.76	0.4	1.65	0.08	0.25	30
Manston	1.56	0.4	1.51	0.08	0.28	30
Santon Downham	1.78	0.5	1.81	0.08	0.25	30
Stansted Mounfichet	1.76	0.5	1.71	0.09	0.29	30
Terrington SC	1.65	0.4	1.67	0.07	0.26	30
Writtle	1.61	0.5	1.57	0.08	0.28	30

Table 4. 1: Weather stations analysis of the May-October daily mean precipitation from 1971-2000.



Figure 4. 3: Box-and-whisker plot displaying the characteristics of the May-October precipitation data for 1971-2000 from the weather stations observations. The solid thick black line within the box represents the median (2nd quartile) of the distribution. The extremes of the box represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and highest values of the distribution.

4.2.2 Weather stations total daily May-October wet days precipitation analysis

In this Section, the total May-October daily precipitation for the growing season was summed up for each year in each of the station for the period 1971-2000. This enabled the days that have precipitation events to be separated from days without precipitation. Results of the distribution of precipitation events showed good agreement among majority of the stations with similar patterns of distribution over the May-October growing season (Figure 4.4). The stations distribution of total May-October precipitation over the study area showed year-to-year variations similar to the May-October daily mean precipitation described in Section 4.2.1. The characteristics of the total May-October distribution of precipitation for all stations (Figures 4.4) revealed a maximum precipitation of 68 mm, a minimum of 0.1 mm and an overall mean of 3.8 mm.

Further, the total May-October mean precipitation analysis of individual stations for the period 1971-2000 in Table 4.2 shows Santon Downham and Stansted Mountfichet returning the highest mean precipitation values of 4.03 mm day-¹ (Std 5.6) and 4.02 mm day-¹ (Std 5.9) respectively. The next station with the highest mean precipitation was Writtle with 3.98 mm day-¹ (Std 6.0)

followed by Manston with 3.76 mm day-¹ (Std 5.4), Terrington St. Clement with 3.72 mm day-¹ (Std 5.5) and Coltishall with 3.70 mm day-¹ (Std 5.6). The mean precipitation among the stations ranged from the lowest 3.69 mm to 4.03 mm day-¹. Statistical analysis conducted using ANOVA multiple means comparison test revealed that there were no statistically significant differences in the total May-October precipitation from the different stations. The test returned a p-value of 0.08 at 95% confidence interval suggesting that there were no statistically significant long-term trends in the total May-October daily precipitation over the region from 1971-2000.

The May-October daily precipitation count (see Table 4.2) for all the stations indicated less number of wet day precipitation compared to dry days. Although, the number of wet days vary from one station to another; the station with the highest number of wet days in this case was Coltishall and had the lowest percentage decrease. Conversely, the stations with the lowest number of wet days (e.g. Manston and Writtle) had the highest number of dry days. The remaining three stations (Santon Downham, Stansted Mountfichet and Terrington St. Clement) were within the same range. The result of low precipitation for Manston is consistent with result in Section 4.2.1 and Table 4.2 shows Manston with the lowest number of wet day precipitation out of the range and, again implies that it is in a different precipitation regime to the other stations. However, all the stations showed a percentage decrease in the number of wet days than dry days for the period 1971-2000 (Table 4.2).



Figure 4. 4: Distribution of the total May-October precipitation from each of the stations for the period 1971-2000.
Stations	Mean	Standard Deviation	Wet Days	Dry Days	Difference between wet & dry days
Coltishall	3.7	5.6	2616	2904	288
Manston	3.76	5.4	2201	3319	1118
Santon Downham	4.03	5.6	2462	3058	596
Stansted Mounfichet	4.02	5.9	2413	3107	694
Terrington SC	3.72	5.5	2486	3034	548
Writtle	3.98	6	2230	3290	1060

Table 4.2: Analysis of the total May-October daily precipitation of each station from 1971-2000.

4.2.3 Weather stations monthly May-October daily precipitation

The monthly May-October daily precipitation provides a general view of the precipitation characteristics of the region and results indicated similar pattern of positively skewed precipitation distribution (Figure 4.5) in all the stations from May to October over the 30-year (1971-2000) period suggesting that precipitation over the region was governed by similar factors. The stations again showed similar patterns of distribution for the study area as the stations May-October mean precipitation described in Section 4.2.1. This type of consistency implies that the data could be relied upon for the precipitation analysis in this study.

Figure 4.6 showed similar precipitation trends and high monthly variations for each year of the study although, three of the stations (Santon Downham, Terrington St. Clement and Writtle) indicated episodes of heavy precipitation in June and August and a couple of events occurring in Writtle in the month of July. These notable exceptions (Figure 4.6) showed high precipitation events in June of 1975 in Santon Downham, Terrington St. Clement and Writtle. Coltishall and Writtle also recorded high precipitation events in July 1989 while Manston, Terrington St. Clement and Writtle recorded high precipitation events in August 1992. Results illustrated in Figure 4.7 suggest that the monthly May-October precipitation from the stations vary from month-to-month but without a long-term discernible trend.

Statistical analysis of the monthly May-October precipitation showed that the highest mean precipitation occurred in Writtle with a value of 4.53 mm (Std 7.5) in August (see Table 4.3). Conversely, the lowest monthly mean precipitation was recorded in Coltishall in May with a

value of 3.15 mm (Std 4.1). The mean monthly variations between the stations therefore vary between 3.15 mm and 4.53 mm with high standard deviations varying between 4.1 in Manston (lowest) in the month of May and 7.5 in Writtle (highest) in the month of August. The monthly variations in precipitation is important in this study in order to determine which months that may be adversely affected by the changes in precipitation particularly during the growing season of the summer months when temperatures are likely to be highest.



Figure 4. 5: Distribution of the monthly May-October daily precipitation of the growing season from the weather stations data for the period 1971-2000.



Figure 4. 6: Observation plots of the monthly May-October mean precipitation of the growing season from the weather stations data for the period 1971-2000.



Figure 4. 7: Box and whisker plot displaying the characteristics of the monthly May-October precipitation data from the weather station observations for the period 1971-2000. The boxplot details are the same as in Figure 4.3.

	Coltishall		Manston		Santon D		Stansted M		Terrington SC		Writtle	
Months	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std
May	3.15	4.1	3.18	4.3	3.37	4.3	3.36	4.5	3.41	4.7	3.55	4.9
June	3.89	5.4	3.59	4.7	4.26	6	4.21	5.2	3.61	5.5	4.2	5.9
July	3.82	5.7	3.42	4.9	4.29	5.2	3.88	6.1	3.84	5	3.53	6.1
August	4.09	6.9	3.85	5.8	4.29	6.3	4.22	6.1	4.17	6.6	4.53	7.5
September	3.91	5.8	4.22	6.6	4.19	5.8	4.52	6.7	3.85	5.8	4.11	5.8
October	3.41	5.3	4.17	5.6	3.83	5.7	4.01	6.4	3.52	5.1	3.98	6

 Table 4. 2: Analysis of the stations monthly May-October daily precipitation from 1971-2000 from the five stations with similar characteristics.

4.2.4 Summer (June, July, August; JJA precipitation)

In this Section, firstly, the summer June, July and August (JJA) mean precipitation was summed up for the three months in all the stations for the period 1971-2000 and secondly; an assessment of the individual JJA months was conducted. Result of the analysis also showed fairly good agreement between the stations which are consistent with results obtained from the total May-October mean precipitation analysis. The summer (JJA) mean precipitation for all the stations ranged from 3.61mm day-¹ in Manston (lowest) to the highest of 4.28mm day-¹ in Santon Downham (Table 4.4). The highest JJA number of precipitation days from all the stations was recorded in Coltishall with 1208 days followed by Terrington SC with 1176. Manston had the lowest number of precipitation days with 1005 (Table 4.4). Figure 4.8 shows the same pattern of precipitation distribution in all the stations and Table 4.4 shows a reduction in the number of wet precipitation events recorded for the stations.

Observed JJA precipitation distribution (Figure 4.8) from (1971-2000) showed the same pattern of distribution for all the stations further supporting the reliability of the data as representative of the study area. Manston again showed the highest percentage decrease and least number of wet day precipitations. Writtle had the second highest percentage decrease in the number of wet day precipitations but unlike Manston, Writtle is within the range of the study area.

Stations	Mean	Std	No. of precipitation days	No. of Non-precipitation days	% change in the No. of precipitation days
Coltishall	3.93	6	1208	1552	-28.5
Manston	3.61	5.1	1005	1755	-74.6
Santon D	4.28	5.8	1160	1600	-38
Stansted M	4.1	5.8	1123	1637	-46
Terringto SC	3.86	5.8	1176	1584	-34.6
Writtle	4.09	6.5	1038	1722	-65.8

 Table 4. 3: Analysis of the total June, July and August (JJA) mean precipitation from each of the weather stations for the period (1971-2000).



Figure 4. 8: Summer JJA precipitation of each of the stations for the period 1971-2000.

Similarly, Table 4.5 shows the individual summer JJA month's precipitation analysis and it agrees with the result from the Monthly May-October precipitation. Although, this result is obvious in Table 4.3, it was still necessary to present it separately to correspond with the precipitation distribution presented in Figure 4.9. Table 4.5 shows that the mean precipitation for the individual JJA months recorded higher mean values in August over the study period for all the stations. Mean precipitation in June and July varied among the stations with June recording more mean precipitations in Coltishall, Manston, Stansted Mounfichet and Writtle, while July recorded more in Santon Downham and Terrington St. Clement.

In spite of these differences, statistical analysis conducted using ANOVA multiple comparison of means test revealed that there were no significant differences in the means of the individual JJA precipitation from all the stations. The test returned a p-value of 0.09 at 95% confidence interval suggesting that there were no significant long-term trends in the JJA precipitation over the region from 1971-2000. Figure 4.9 shows the distribution of the summer JJA precipitation and also portrays similar precipitation characteristics for all the stations which is consistent with results from the seasonal and monthly precipitation analysis. Figure 4.10 suggest that the JJA May-October precipitation from the stations vary monthly with the mean distribution of precipitation events higher in June than July and August.

	Coltishall		Manston		Santon D		Stansted M		Terrington SC		Writtle	
Months	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std
June	3.89	5.4	3.59	4.7	4.26	6	4.21	5.2	3.61	5.5	4.2	5.9
July	3.82	5.7	3.42	4.9	4.29	5.2	3.88	6.1	3.84	5	3.53	6.1
August	4.09	6.9	3.85	5.8	4.29	6.3	4.22	6.1	4.17	6.6	4.53	7.5

Table 4. 4: Analysis of monthly June, July, August (JJA) means precipitation for the period 1971-2000.



Figure 4. 9: Summer JJA mean precipitation for each month for the stations observed data.



Figure 4. 10: JJA monthly May-October precipitation data from the weather station observations for the period 1971-2000. The boxplot details are the same as in Figure 4.3.

In summary, the weather stations in the study area showed good agreement of the characteristics and distribution of the annual, total, monthly and JJA May-October precipitation of the region from 1971-2000. The exception being Manston which have been described earlier as being outside of the sugar beet producing areas and has different characteristics from the rest of the stations in the region. Therefore, Manston will be excluded from subsequent calculations.

4.3 **Present climate**

4.3.1 Comparison of weather station data and the "historical" phase of the CMIP5

Annual and seasonal precipitation of Eastern England and England in general is highly variable but there are indications of a decrease in precipitation over the summer (Jenkins et al. 2008) which is directly related to one of the research hypotheses in this thesis (i.e. Reduction in future precipitation will reduce sugar beet yield by 2050). In this respect, it is useful to gain an insight into the overall growing season changes in precipitation over the area of interest using climate models. A study by Sillmann et al. (2013) examined a range of CMIP5 climate models in reproducing precipitation and found an improvement in the CMIP5 ensemble mean compared to the CMIP3. In this study, the ensemble mean is used to make comparisons of the weather station observations against the CMIP5 model outputs for the historical phase 19761-2000.

In doing this, a comprehensive assessment of the May-October mean precipitation of the area in Eastern England was investigated for the same 30 year period. The assessment covered the historical phase from 1971-2000 in which the stations data was compared to precipitation data from eight CMIP5 climate model ensembles under the "historical" forcing (Meinshausen et al. 2011; Eyring et al. 2016). The use of the eight ensemble models was aimed at improving mean estimates, providing multiple averages over single model runs and providing better prediction and distribution of possible outcomes (see Chapter 3.4). It also has the following advantages:

- It helps to obtain a better predictive performance as opposed to an individual model run
- Reduces errors specific to single models
- By combining multiple models, a more robust and accurate prediction is obtained over single model
- The ensemble produces multiple forecasts by making minor alterations to the initial conditions. By so doing, each member of the ensemble will be equally likely to produce forecast based on probabilities of different possible outcomes.

Past studies have shown that combining multiple models generally increases skill, reliability and consistency in model forecasts (McSweeney et al. 2014; Tebaldi & Knutti 2007; Tubiello & Ewert 2002). In this study, it enabled the verification of model performance in replicating past observations and thus, provides the basis for use in projections of future medium (RCP45) and high (RCP85) greenhouse gas emission scenarios. Simulation results of the CMIP5 "historical" phase vary from model to model as a result of different external forcing and internal variability leading to varying degree of model behaviour, skill and performance. According to McSweeney et al. (2014) models perform differently in different regions and therefore a model that performed very well in Europe might not necessarily perform well in Africa or Asia for example. The aim here is to identify the models that perform well in the region of interest.

The models precipitation under the "historical" forcing (1971-2000) were varied with three of the eight models output (see Figure 4.11) replicating observed data very well while the remaining five models differ significantly from those of the station observations. The ensemble mean of the growing season (May-October) precipitation recorded for the models and weather stations from 1971-2000 are shown in Table 4.6. The highest mean of 2.28 mm day-¹ recorded for CSIRO, followed by IPSL 2.09 mm day-¹, NCAR 2.05 mm day-¹, MPI-M 2.04 mm day-¹, MIROC 2.03 mm day-¹, CCCma 1.74 mm day-¹, EC-Earth 1.62 mm day-¹ and the MOHC (HadGEM2-ES) 1.61 mm day-¹.

Stations	Mean	Std	Median	Data: n
Coltishall	1.76	0.44	1.65	30
Manston	1.56	0.43	1.51	30
Santon D	1.78	0.45	1.81	30
Stansted M	1.76	0.5	1.69	30
Terrington SC	1.65	0.42	1.67	30
Writtle	1.61	0.45	1.57	30
CMIP5 Models				
CCCma	1.74	0.27	1.66	30
CSIRO	2.28	0.19	2.27	30
EC-Earth	1.62	0.14	1.63	30
IPSL	2.09	0.21	2.07	30
MIROC	2.03	0.22	2.05	30
MOHC	1.61	0.21	1.62	30
MPI-M	2.04	0.23	2.04	30
NCAR	2.05	0.28	2.05	30

 Table 4. 5: May-October annual means precipitation analysis from the weather stations and CMIP5 data for the period (1971-2000).

Analysis of the May-October annual daily mean precipitation is presented as boxplots in Figure 4.11 with the thick black line in the middle showing the median of the distribution. The median is used here as it is less sensitive to outliers, minimum and maximum values in the distribution.

As reported earlier, three out of the eight CMIP5 models showed mean and median (Table 4.6) similar to those of the weather stations that represent the sugar beet farming areas in the region. The three models that best replicated the May-October daily mean precipitation under the historical phase are: CCCma, EC-Earth and MOHC (see Figure 4.11). Excluding Manston, with a mean of 1.56 mm day-¹, the May-October mean precipitation for the rest of the stations ranged between 1.61 mm day-¹ and 1.78 mm day-¹, and similarly, the May-October mean precipitation for the three models that represented observations very well ranged from 1.61 mm day-¹ to 1.74 mm day-¹. Figure 4.11 also showed that the MOHC has a better representation of the range of precipitation event sizes and, therefore, the MOHC model was solely used in this thesis for the analysis of future precipitation.

The three models that best replicated observed Eastern England precipitation were selected and used for further analysis of precipitation projections in this thesis. Selection of models in this manner has been reported in past studies that not all models are suitable for all regions. This has led to the exclusion of some General Circulation Models (GCMs) in studies of future projections as a result of poor model performance of observed climate (McSweeney et al. 2014; Manning et al. 2009; Sexton et al. 2004). The remaining five models, namely: CSIRO-Mk3.6.0, IPSL-CM5A-LR, MIROC5, MPI-ESM-LR and the CCSM4 were not in line with the observations and were therefore considered undesirable and rejected. Furthermore, past studies such as (McSweeney et al. 2014; Brands et al. 2013) also reported that not all models perform well in every region but their studies also showed that the MOHC (HadGEM2-ES) outperformed other models in this region and the UK in general and, was considered best for calculating the watering regimes reported in chapter five of this thesis.



Figure 4. 11: Box-and-whisker plot displaying the characteristics of the station observations and the historical phase of the CMIP5 models from May-October for 1971-2000. The solid thick blue line indicates the median of the distribution and the solid thick black line within the box represents the median (2nd quartile) of the distribution. The extremes of the box represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and highest values of the distribution. The blue line running across the plot shows the combined mean precipitation of the stations.

Figure 4.12 showed that the stations data had multimodal and symmetrical shaped distributions compared to the CMIP5 models with mostly unimodal and in some cases skewed distribution. As previously mentioned, the range and distribution of observed precipitation from the stations data were much wider than the simulations from the Earth System Models (ESMs) used. This is not unexpected as models do not represent the extremes of variability well as pointed out by (Maraun et al. 2010). Again, this is not seen as a problem as the study here examined mean conditions rather than the extremes. Nonetheless, the distribution of precipitation from the MOHC model is much closer to that of the observations than the CCCma and EC-Earth models. Therefore, the MOHC model projections were prominently used in further precipitation calculations in this thesis (Section 4.5).



Figure 4. 12: Histogram distribution plots of the May-October daily mean precipitation for 1971-2000 from the historical phase of the daily weather stations observations and the CMIP5 models.

In summary, the response of the CCCma, EC-Earth and MOHC models to variability in May-October seasonal precipitation over the 30-year period (1971-2000) showed patterns consistent with the observations but has wider variations.

4.4 **Projections of future precipitations**

4.4.1 Analysis of the May-October daily mean precipitations under the "historical" and "future" phases of the CMIP5 models

This Section of the study investigated a core aspect of this research by examining precipitation projections for Eastern England. The study compared CMIP5 models that best replicated the May-October daily ensemble mean precipitation for the "historical" period (1971-2000) with the CMIP5 ensemble mean precipitation for the "future" scenarios under RCP45 and RCP85 for the period (2021-2050). This Section also investigated the changes in May-October daily mean precipitation by examining the two different time frames to detect for a trend in precipitation.

The aim was to find out if a change existed and if so, how much? Answers to these questions will be used to calculate the watering regimes to be implemented in the greenhouse plant experiment in chapter five of this thesis.

Analysis of the May-October daily mean precipitation (Table 4.7 and Figure 4.13) showed good agreement among the models of a reduction in May to October precipitation between the historical and future scenarios under RCP45 and RCP85. Comparison between the CMIP5 models "historical" phase and projections from the CCCma, MOHC and the EC-Earth showed that UK precipitation decreased in the three models, apart from the EC-Earth model which showed an increase in mean precipitation under the RCP85. Among the three models, the MOHC indicated the biggest negative changes under the medium and high greenhouse gas emission scenarios. This result is consistent with the studies of (Kendon et al. 2016; Jenkins et al. 2008; Osborne & Hulme 2002; Osborne et al. 2000) of a predicted decrease in summer precipitation. The distribution of the ensemble mean precipitation output for the historical phase of the MOHC data is the closest to the station observed mean reported in Section 4.3.1. Again, this characteristic added to the multiple reasons why the MOHC was heavily relied on for the greenhouse plant experimental watering regimes between the present and future scenarios. Moreover, precipitation calculations based on the MOHC provides a plausible but relatively extreme scenario that could be a potential source of threat to the UK sugar beet farming and the sugar industry.

I	Historical		RCP45		RCP85			
Models	Mean	Std	Mean	Std	Mean	Std		
CCCma	1.74	0.3	1.49	0.2	1.5	0.2		
EC-Earth	1.62	0.2	1.54	0.2	1.81	0.2		
MOHC	1.61	0.2	1.35	0.2	1.38	0.2		

 Table 4. 6: Ensemble means of the May-October mean precipitation from the best three performing models (CCCma, MOHC and EC-Earth).



Figure 4. 13: Boxplot of the May to October daily mean precipitation data from the a) historical phase (1971-2000), b) RCP45 (2021-2050) and c) RCP85 (2021-2050) output from the CCCma, MOHC and EC-Earth climate models. The boxplot details are the same as for Figure 4.3.

4.4.2 Changes in the May-October daily mean precipitations under the "historical" and "future" phases (RCP45 and RCP85) of the CMIP5 models

Result of the May-October daily mean precipitation changes based on Relative Precipitation Comparison test between the two temporal windows showed a reduction in precipitation between the two periods under investigation (1971-2000) and (2021-2050). This method was employed by Baker & Huang (2012) to compare absolute and relative climate changes in future 105 precipitation projections under different time frames. Table 4.8 and Figure 4.14 highlight the differences between the two time frames from the CCCma, EC-Earth and the MOHC models. The results showed a negative change in precipitation between the two time frames under CCCma and MOHC climate models. However, EC-Earth model in contrast indicated a positive change between the two time frames (see Table 4.8). The historical phase for example returned an ensemble mean of 1.62 (mm) compared to the RCP85 run with an ensemble mean of 1.38 (mm) which indicates a precipitation reduction of 14.9% between the historical period (1971-2000) and the future period (2021-2050) - see Table 4.8. Similarly, the CCCma showed a reduction of 14.3% reduction in May-October mean precipitation but the EC-Earth model indicated a contrasting increase of 3.72%. This type of difference is not unusual as different models perform differently under different conditions in different regions (McSweeney et al. 2014; Brands et al. 2013) - See Section 4.4.1.

Models			
CCCma	Period	Mean daily precipitation (mm day- ¹)	Difference from historical (%)
Historical	1971-2000	1.74	0
RCP45	2021-2050	1.49	-14.3
RCP85	2021-2050	1.5	-13.7
Mean of RCP45 and RCP85	2021-2050	1.49	-14.3
EC-Earth			
Historical	1971-2000	1.61	0
RCP45	2021-2050	1.54	-4.34
RCP85	2021-2050	1.81	12.4
Mean of RCP45 and RCP85	2021-2050	1.67	3.72
монс			
Historical	1971-2000	1.62	0
RCP45	2021-2050	1.35	-16.8
RCP85	2021-2050	1.38	-14.9
Mean of RCP45 and RCP85	2021-2050	1.36	-15.8

 Table 4. 7: Result of the CMIP5 climate models changes in May-October daily mean precipitation for the sugar beet growing season.

Figure 4.15 showed all the models had unimodal shaped distribution with the MOHC showing a much better distribution than the rest particularly under the historical phase and clearly showed a reduction in precipitation under RCP45 and RCP85. Figure 4.15 agrees with Figure 4.14 with the CCCma and EC-Earth also showing reduction in precipitation under RCP45 and RCP85. Only EC-Earth showed a noticeable increase in precipitation under RCP85 compared to the Historical phase.



Figure 4. 14: Time series plot from CCCma, EC-Earth and the MOHC HadGEM2-ES climate models illustrating the trend of the May to October precipitation for the historical (1971-2000), RCP45 and RCP85 (2021-2050) scenarios. The blue horizontal line running across the plot represents the stations mean line as per Figure 4.9 plotted against the CMIP5 models.



Figure 4. 15: May-October mean daily precipitation for the historical phase, RCP45 and RCP85 from the CCCma, EC-Earth and MOHC climate models for the periods 1971-2000 and 2021-2050.

The change indicated by this result implies that there may be different mechanisms (i.e. different

responses to increasing greenhouse gas emissions in the atmosphere) at play controlling precipitation delivery in the two time frames in the different models; historical (1971-2000) and future (2021-2050). Past and recent studies in the UK have indicated both high variability and a decrease in mean summer precipitation Met Office (2014); Jenkins et al. (2009) as illustrated in Figure 4.13 from the CCCma and MOHC climate models.

4.4.3 Total May-October wet days precipitation under the "historical" phase, "RCP45" and RCP85 scenarios using the CMIP5 models

In this Section of the study, CMIP5 climate models was used to evaluate the total May-October daily precipitation for the growing season under the historical phase for the period 1971-2000 and for the future scenarios under RCP45 and RCP85 for the period 2021-2050. The total May-October daily precipitation was calculated by summing up the total number of precipitation event days between May-October for each year of the study period for the historical phase (1971-2000) and the future scenarios under RCP45 and RCP85 (2021-2050). This enabled days with non-precipitation events to be removed from the analysis. Using the MOHC for example, the MOHC has a monthly calculation based on 30 day calendar for all months compared to the CCCma and EC-Earth based on normal calendar days for each month in a 365 day year. Therefore, the total May-October precipitation data points for the MOHC model was 5400 based on 180 days in each year (e.g., May-October) for the 30 year period whereas the CCCma and EC-Earth had 5520 data points based on 184 days from May and October for each year of the study period.

Analysis showed fairly good agreement among the models of a reduction in total May-October precipitation between the historical phase and the future scenarios. Results across models indicate that the three models agree on a reduction in precipitation between the historical phase and RCP45 scenario (see Table 4.9). Similarly, the CCCma and the MOHC also agreed on a reduction between the Historical phase and the RCP85 scenario but the EC-Earth model showed no reduction in precipitation between the historical period and RCP85. Table 4.9 showed the mean analysis and the number of wet days between the two time frames. Results here are consistent with results of the models for the May-October mean daily precipitation reported in Section 4.4.2. Figure 4.16 illustrates the distribution of precipitation from the models between the historical phase and the future periods. The models displayed a unimodal shaped and

positively skewed distribution with indications of a reduction in precipitation between the historical phase of the models and the future scenarios with the exception again being RCP85 from the EC-Earth which indicated an increase in precipitation compared to the historical.

Results of the total May-October daily precipitation analysis carried out in this Section compares well to the total May-October daily mean precipitation in Section 4.2.2. The historical phases from the CMIP5 models and the future scenarios under RCP45 and RCP85 exhibited similar patterns of precipitation distribution as the weather stations total May-October daily precipitation (also see Figure 4.4). Furthermore, the range and distribution of observed precipitation from the weather stations were much wider than those of the CMIP5 models which is not unexpected as models do not represent extremes of precipitation variability well (Maraun et al. 2010).

Models					
CCCma	Mean	Std	Total # of days	Wet days precipitation	Non-wet days precipitation
Historical	2.45	2.3	5520	3929	1591
RCP45	2.37	2.4	5520	3698	1822
RCP85	2.43	2.5	5520	3563	1957
EC-Earth					
Historical	2.44	2.3	5520	3658	1862
RCP45	2.37	2.3	5520	3582	1938
RCP85	2.44	2.3	5520	3664	1856
MOHC					
Historical	2.49	2.6	5400	3669	1731
RCP45	2.28	2.8	5400	3177	2223
RCP85	2.38	2.9	5400	3133	2267

Table 4. 8: Mean analysis of total May-October distribution of daily precipitation between the two differenttime frames (historical phase 1971-2000 and the RCP45 and RCP85 scenarios 2021-2050).



Figure 4. 16: Distribution of the total May-October precipitation for the historical period (1971-2000) and RCP45 and RCP85 scenarios (2021-2050) from the CCCma, EC-Earth and MOHC climate models.

4.4.4 Changes in total May-October wet day precipitation under the "historical" phase, "RCP45" and RCP85 scenarios

Changes in the total May-October wet days precipitation were reflected in the precipitation event sizes between the historical phase (1971-2000) of the models and future scenarios under RCP45 and RCP85 (2021-2050). Comparison of the total wet day precipitation for the two temporal windows using the ensemble means indicated varying results across the models. Tables 4.10 and 4.11 highlight the differences in mean wet day precipitation and number of wet day precipitation between the two different time frames from the CCCma, EC-Earth and MOHC models. Results represent projected changes in future wet days precipitation which varies from one model to another. All the models agree on a reduction (See Table 4.10) in the total May-October wet days precipitation between the historical phase and RCP45 scenario while the models all indicated no significant difference between the RCP45 and RCP85 May-October wet days precipitation. Although, the number of wet day precipitation has been presented in Section 4.4.3, for ease of reference in this Section, the percentage change in total wet day precipitation and change in the number of wet days between the historical phase and future scenarios are added to the analysis in Tables 4.10 and 4.11.

Results across the models varied with the CCCma indicating a -3.27% reduction in precipitation between the two time frames under RCP45 and -0.82 under RCP85. The EC-Earth indicated no change (0%) between the two periods under RCP85 and -2.87 while the MOHC model indicated -8.43% and -4.42 under RCP45 and RCP85 respectively (Table 4.10). The MOHC again showed the biggest negative changes under future scenarios with a much larger change under RCP45. In short, the CCCma and MOHC showed similar precipitation patterns indicating a decrease in total May-October wet day precipitation under RCP45 and RCP85. Similarly, the number of wet day (precipitation) counts showed a reduction in all the models between the historical phase and RCP45 scenario while only CCCma and MOHC showed a reduction between the historical phase and RCP85, the EC-Earth differed with a slight increase in the number of wet days (See Table 4.11).

Again, Table 4.11 show that the MOHC model has the largest negative change in the number of wet day precipitations further supporting the use of the data as the basis for the future precipitation calculations so that a relatively extreme scenario could be investigated with regards

to water stress that may potentially affect the UK sugar beet industry. Furthermore, this result implies that different distribution of precipitation events and sizes prevailed under the two different scenarios. The result showed a larger decline in precipitation under the RCP85 emission scenario than the RCP45.

Models			
CCCma	Period	Mean daily precipitation event size (mm per day)	Difference from historical (%)
Historical	1971-2000	2.45	0
RCP45	2021-2050	2.37	-3.27
RCP85	2021-2050	2.43	-0.82
EC-Earth			
Historical	1971-2000	2.44	0
RCP45	2021-2050	2.37	-2.87
RCP85	2021-2050	2.44	0
MOHC			
Historical	1971-2000	2.49	0
RCP45	2021-2050	2.28	-8.43
RCP85	2021-2050	2.38	-4.42

Table 4. 9: Result of the CMIP5 climate models mean wet days for the growing season total May-Octoberdaily precipitation. Changes between the historical phase (1971-2000) and future scenarios under RCP45and RCP85 (2021-2050) are presented.

 Table 4. 10: Projected changes in the number of wet day precipitation from the models for the total May

 October daily mean precipitation between the historical phase and future scenarios (RCP45 and RCP85).

Models				
CCCma	Period	Number of observations	Number of wet days	Change in number of wet days
Historical	1971-2000	5520	3929	0
RCP45	2021-2050	5520	3698	-231
RCP85	2021-2050	5520	3563	-366
EC-Earth				
Historical	1971-2000	5520	3658	0
RCP45	2021-2050	5520	3582	-76
RCP85	2021-2050	5520	3664	6
MOHC				
Historical	1971-2000	5400	3669	0
RCP45	2021-2050	5400	3177	-492
RCP85	2021-2050	5400	3133	-536

4.4.5 CMIP5 models individual month's May-October daily precipitation

This Section provides model response to individual month's (May-October) daily precipitation for the study area covering the historical period (1971-2000) and the future scenarios (2021-2050). The monthly precipitation was derived by summing up daily totals for each month from May-October for the 30 year period. This provides a general view of the monthly precipitation characteristics of the area and the comparison between the two different time frames will help in identifying trends and variability in precipitation during the summer growing season. In reality, changes in precipitation vary from one month to another meaning that some months will likely have more or less precipitation than others and it also provides an indication of a "realistic distribution of future precipitation scenario" that is important in determining the monthly watering regimes for individual months.

Results of the mean statistical analysis conducted for the monthly May-October precipitation using ANOVA multiple mean comparison (Table 4.12) revealed large variations in mean and standard deviation of precipitation amongst the models. The highest monthly mean precipitation occurred in October under RCP85 from the EC-Earth model with a value of 4.14 mm per day (Std 2.4). The lowest monthly mean precipitation occurred in July also under RCP85 with a value of 1.66 mm per day (Std 2.4) from the EC-Earth model. It is noteworthy to point out that the EC-Earth model showed the biggest dispersions amongst the models especially in the months of May and October (Figure 4.17). This result however, does show a striking similarity with the weather stations observation data (Figure 4.9) with indications of high monthly variations. Table 4.12 shows the extent of dispersion among and between the models allowing for illustration of changes in monthly precipitation. In this respect, more attention is focused on the negative changes in precipitation in order to identify months with the biggest negative impacts.

The CCCma indicated a decline in July, August and September mean precipitation under RCP45 and RCP85. The model also showed a negative change in mean precipitation in August and September, and a marginal decrease in October under the RCP85. The EC-Earth showed negative changes in August and September monthly mean precipitation under RCP45 and in July under RCP85. The MOHC model reported a decrease in monthly mean precipitation in June, July and August under both the RCP45 and RCP85 scenarios. Overall, negative precipitation changes were identified in the JJA months under both future scenarios (RCP45 and RCP85).

Similarly, the number of wet days showed a general reduction in precipitation in all the months from the CCCma model except in June where it reported an increase under RCP45. The EC-Earth model reported a one day decrease in the number of wet days under RCP45 and an increase under RCP85 in May. June and July also showed increases in the number of wet days under RCP45 and RCP85. August, September and October showed decreases in the number of wet days under days under RCP45 and RCP85. The MOHC model on the other hand, showed a reduction in the number of wet days across all the months from May-October. The question now is, "does the reduction in the number of wet days translate to reduced precipitation over the summer season growing months?

Past studies such as Jenkins et al. (2008), Osborn et al. (2000), Wigley & Jones (1987) provide evidence of precipitation trends towards drier summers as a result of reduction in the frequency of wet days and a reduction in mean wet day precipitation amounts. Similarly, Simpson & Jones (2012); Jones & Conway (1997) confirmed that there was a general reduction in summer July and August precipitation across the UK and it is consistent with results in this study. Likewise, Perry & Hollis (2005) in their study comparing two different time frames 1971-2000 and 1961-1990 reference period also found a reduction in summer precipitation and the number of wet days precipitation nationwide.

CCCma		н	listorical				RCP 45				RCP85	
Months	Mean	Std	Wet days	s Total counts	Mean	Std	Wet days	Total counts	Mean	Std	Wet days	Total counts
May	3.19	1.9	635	930	3.23	2	599	930	3.41	2	582	930
June	2.42	1.7	543	900	2.55	1.8	557	900	2.61	1.9	522	900
July	1.93	1.5	629	930	1.87	1.4	561	930	1.94	1.8	529	930
August	2.12	1.8	726	930	1.7	1.6	657	930	1.85	1.5	591	930
September	2.17	1.9	670	900	2.1	1.6	616	900	2.02	2.1	626	900
October	2.88	2.1	726	930	3	2.4	708	930	2.86	2.3	713	930
EC-Earth												
May	2	1.4	572	930	2.16	1.6	571	930	2.14	1.4	633	930
June	1.93	1.5	548	900	1.96	1.7	567	90	2.1	1.6	566	900
July	1.81	1.3	486	930	1.86	2.2	551	930	1.66	1.4	513	930
August	2.05	1.6	575	930	1.92	1.6	536	930	2.1	1.5	549	930
September	2.91	1.9	686	900	2.76	2.1	592	900	2.92	2	634	900
October	3.95	2.3	791	930	4.05	2.3	765	930	4.14	2.4	769	930
MOHC												
May	2.55	1.9	606	900	2.63	2	552	900	2.68	2.1	537	900
June	2.67	1.9	540	900	2.5	1.9	523	900	2.29	1.9	495	900
July	2.46	1.8	601	900	2.29	1.8	516	900	2.23	1.7	489	900
August	2.43	2	551	900	2.15	1.9	466	900	2.41	2.3	473	900
September	2.34	2	622	900	2.12	2.2	480	900	2.54	2.4	496	900
October	2.74	2.4	749	900	2.68	2.3	640	900	2.94	2.5	643	900

 Table 4. 11: Models analysis of monthly May-October precipitation under the historical phase (1971-2000) and future scenarios, RCP45 and RCP85 (2021-2050).



Figure 4. 17: Boxplots showing the distribution of monthly precipitation from the models under the historical phase (1971-2000) and future scenarios (2021-2050). The boxplot details are the same as for Figure 4.3.

4.4.6 Changes in individual month's May-October daily precipitation

In this Section, individual models were evaluated for changes in individual month's May-October daily precipitation for the historical phase (1971-2000) and the future scenarios under RCP45 and RCP85 (2021-2050). Relative precipitation comparison test of the two temporal windows using the ensemble means indicated varying results across models. Table 4.13 highlights the differences in mean monthly precipitation between the two different time frames from the CCCma, EC-Earth and MOHC models, however focus here is also on negative changes as mentioned in Section 4.4.5. Results indicated that projected changes in future precipitation during the season showed high monthly variability across the models. The CCCma model for instance indicated negative changes in precipitation in July, August and September under RCP45. The RCP85 scenario showed negative changes in August and September and a small change of -0.7 in October. The largest negative changes in precipitation were recorded in August with a value of -19.8% recorded under RCP45. The second largest negative change was recorded again in August under RCP85 with a value of -12.7 followed by September with a value of -6.9 under RCP85. Other negative changes were also recorded in September and July under RCP45 for -3.2 and -3.1 respectively.

The EC-Earth model equally showed negative changes with the greatest change recorded in July with a value of -8.3. August and September recorded values of -6.3 and -5.6 respectively under RCP45 (See Table 4.13). The MOHC model indicated the most negative changes in monthly May-October precipitation across the models. The model indicated reduced precipitation in June, July, August, September and October under RCP45. The RCP85 also showed reduced precipitation in the months of June, July and August. The biggest change in precipitation was calculated for June with a reduction of -14.2% under RCP85 followed by August with -11.5%. Other negative changes were also calculated across different months (e.g. June with a value of -6.4%, July -6.9% and September with a value of -9.4% (Table 4.13) under RCP45. The RCP85 indicated negative changes of -9.3% in July, -0.8% in August (Table 4.13).

Models			Mont	hly means			Differe	nce from h	nistorical	(%)		
CCCma	May	June	July	August	September	October	May	June	July	August	September	October
Historical	3.19	2.42	1.93	2.12	2.17	2.88	0	0	0	0	0	0
RCP45	3.23	2.55	1.87	1.7	2.1	3	6.2	5.4	-3.1	-19.8	-3.2	4.2
RCP85	3.41	2.61	1.94	1.85	2.02	2.86	6.9	7.9	0.5	-12.7	-6.9	-0.7
EC-Earth												
Historical	2.01	1.93	1.81	2.05	2.91	3.95	0	0	0	0	0	0
RCP45	2.16	1.96	1.86	1.92	2.76	4.05	7.5	1.6	2.8	-6.3	-5.6	2.5
RCP85	2.14	2.1	1.66	2.1	2.92	4.14	6.4	8.8	-8.3	2.4	0.34	4.8
монс												
Historical	2.55	2.67	2.46	2.43	2.34	2.74	0	0	0	0	0	0
RCP45	2.63	2.5	2.29	2.15	2.12	2.68	3.1	-6.4	-6.9	-11.5	-9.4	-2.2
RCP85	2.68	2.29	2.23	2.41	2.54	2.94	5.1	-14.2	-9.3	-0.8	8.5	7.3

 Table 4. 12: Analysis of changes in Monthly May-October daily mean precipitation between the historical phase (1971-2000), RCP45 and RCP85 (2021-2050) from the CMIP5 models.

Overall, the months of July and August featured in all the models as showing a reduction in precipitation. The MOHC is the only model that showed negative changes across the June, July and August months as shown Table 4.13. These changes further justify the use of the MOHC model as a worst case scenario for the UK sugar beet farming in this research. Moreover, it is the most similar to the observations in terms of event size distribution. The lowest mean precipitations across the models were recorded in June and July which are two of the core summer months (June, July, August - JJA) that are projected to have the biggest changes in future precipitation (Jenkins et al. 2008) of which this research is most interested in. Furthermore, the JJA months are the period in the summer that temperature and evapotranspiration are likely to peak with potential impacts on available water resources and therefore an important aspect of this research.

In summary, the models showed high variations in the individual month's precipitation similar to the high variations exhibited by the weather stations for the period 1971-2000. Therefore, understanding the mechanisms associated with variations in precipitation patterns could be an important aspect in addressing future uncertainties in precipitation projection studies but is however not addressed in this study.

4.4.7 Summer (June, July and August; JJA) precipitation from the CMIP5 models

Results of the JJA daily mean precipitation varied between individual months from the three models for the historical period (1971-2000) compared to the future scenarios under RCP45 and RCP85. The CCCma model showed a significant reduction in monthly JJA mean daily precipitation between the historical and future scenarios under RCP45 and RCP85 in June. There was however, a significant decrease in mean daily precipitation in July and August between the historical phase and the two future scenarios (See Table 4.14 and Figure 4.18). It is important to look at the months of June, July and August because the biggest changes in precipitation are predicted to occur during these period and it coincides with the time of year that temperature values are also at its peak.

The EC-Earth model displayed an increase in monthly JJA mean precipitation in June under RCP45 and RCP85; showed a reduction in July under RCP45 but an increase under RCP85. In August, it showed a decrease under RCP45 and a marginal increase under RCP85. The MOHC (HadGEM2-ES) model on the other hand showed a decrease in monthly JJA mean daily precipitation across all the months under both the RCP45 and RCP85 compared to the other models. The overall performance of the models shown in Table 4.14 and illustrated in Figure 4.18 showed that the MOHC model is in broad agreement with changes in the seasonal and total May-October precipitation analysis. Comparison of the models JJA means precipitation for the period 2021-2050 with the stations JJA means precipitation for the period 1971-2000 (see Table 4.5) showed a reduction in mean precipitation in all the months under RCP45 and RCP85 (See Figure 4.11).

Models	June		Ju	ly	August		
CCCma	Mean	Std	Mean	Std	Mean	Std	
Historical	2.42	1.7	1.93	1.5	2.12	1.8	
RCP45	2.55	1.8	1.87	1.4	1.94	1.6	
RCP85	2.12	1.9	1.7	1.8	1.85	1.5	
EC-Earth							
Historical	1.93	1.5	1.81	1.3	2.05	1.6	
RCP45	1.96	1.7	1.86	2.2	1.92	1.6	
RCP85	2.1	1.6	1.66	1.44	2.1	1.5	
MOHC							
Historical	2.67	1.9	2.46	1.8	2.43	2	
RCP45	2.5	1.9	2.29	1.8	2.15	1.9	
RCP85	2.29	1.9	2.23	1.7	2.41	2.3	

 Table 4. 13: JJA monthly mean precipitation analysis for the historical phase and the RCP45 and RCP85 scenarios.



Figure 4. 18: Distribution of monthly JJA daily mean precipitation for the historical phase (1971-2000), RCP45 and RCP85 (2021-2050). The boxplot shows the 95% confidence interval for the JJA precipitation from the CCCma, EC-Earth and MOHC models.

4.5 Event size distribution of monthly May-October precipitation

Precipitation is one element of climate that is extremely variable in terms of amount, duration and intensity, and its distribution is uneven seasonally or annually causing year-to-year variability in different places and regions. This Section of the study analyses and reports on the distribution of precipitation events and sizes for individual months from May-October for a 30 year period under the historical phase (1971-2000), RCP45 and RCP85 scenarios (2021-2050). Previous simulation results from the three models used in this study revealed large variations in May-October precipitation among the models but the MOHC (HadGEM2-ES) model showed the biggest negative changes in the May-October precipitation (i.e. a worst case scenario precipitation change for the UK) between the historical phase and the future scenarios under RCP45 and RCP85. This behaviour encouraged the single use of the MOHC model for the analysis of event size distribution which was used for calculating the watering regime rates in the plant experiment reported in Chapter 5.

This Section of the study also analysed non-zero precipitation days from a 30 year time series under the historical phase (1971-2000) and future scenarios, RCP45 and RCP85 (2021-2050) from the MOHC model. The analysis was conducted on a monthly basis taking into account the frequency and size of precipitation events (i.e. distribution of wet days and dry days) in individual months so that a more realistic watering regime could be implemented for the plant experiment in Chapter 5. For the purpose of this study, the thesis only dealt with wet days precipitation >0.05 mm per day thereby eliminating precipitation days that are <0.05 to maintain consistency in the experiment.

4.5.1 Analysis of monthly May-October wet day distribution

Results from the MOHC model for the two different time frames in this study so far has been broadly consistent in indicating reduced trends in precipitation under future scenarios. Analysis of the wet day event size distribution of precipitation under the historical phase, RCP45 and RCP85 showed high monthly variations and also revealed that monthly mean precipitation declined in the months of June, July and August (See Table 4.15). Mean precipitation increased in May and October but was dependent on the scenario being considered. In September, mean

precipitation increased under RCP85 but reduced under RCP45 (See Table 4.15). There is considerable reduction in the total number of future non-zero precipitation days under RCP45 and RCP85 from May to October. Although, May and October showed reduction in the number of non-zero precipitation days, precipitation events actually increased. The highest number of precipitation days was recorded in October under the historical phase with a value of 2796 while the lowest number of precipitation days was recorded in August under RCP85 with a value of 1694. Table 4.15 shows a comparison of the total number of precipitation days between historical and future scenarios as well as the mean and standard deviation from the data analysis for the two time frames.

The mean precipitation from this analysis showed remarkable similarity to the individual month's daily mean precipitation (See Table 4.12, Sections 4.4.5-4.4.6). Both results revealed large monthly variations particularly in July and August. In the current analysis, reduction in the distribution of precipitation events in June, July and August gives indication of likely episode of dry spells particularly in July under future scenarios (RCP45 and RCP85). Table 4.15 also showed the percentage change in precipitation between the historical phase and the future scenarios (RCP45 and RCP85). The largest negative changes were recorded in June under RCP85 followed by August, September and July. May showed an increase under RCP45 and RCP45.

	May		June		July		August		September		October	
Category	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std	Mean	Std
Historical	2.55	3.5	2.69	3.92	2.46	3.9	2.43	4.2	2.37	3.9	2.73	4.3
RCP45	2.6	3.7	2.55	3.8	2.3	3.5	2.11	3.6	2.15	4.3	2.71	4.5
RCP85	2.74	4.1	2.27	3.3	2.27	3.3	2.34	4.1	2.55	4.8	2.92	4.9
Change from historical (%)												
Historical	0		0		0		0		0		0	
RCP45	2		-5.2		-6.5		-13.1		-9.3		-0.7	
RCP85	7.5		-15.6		-7.7		-3.7		7.6		7	
Non-zero precipitation days	May		June		July		August		September		October	
Historical	2254		2109		2327		2155		2345		2796	
RCP45	2108		1988		1941		1777		1795		2500	
RCP85	2003		1754		1890		1694		1871		2520	
Difference from historical												
Historical	0		0		0		0		0		0	
RCP45	146		121		386		378		550		296	
RCP85	251		355		437		461		474		276	

 Table 4. 14: Comparison of monthly May-October event size distribution and non-zero precipitation days (wet days) between the historical phase (1971-2000RCP45 and RCP85 from the MOHC model.

4.5.2 Changes in monthly May-October distribution of event sizes

Results of the changes in the distribution of May-October precipitation events and size for the two temporal periods (1971-2000 and 2021-2050) also showed large month-to-month variability. Table 4.15 shows a mean comparison of the distribution of precipitation events under the historical phase and future scenarios (RCP45 and RCP85). The comparison revealed that the highest mean precipitation event occurred in October with a value of 2.92 mm day-¹ under RCP85. The lowest mean precipitation occurred in August with a value of 2.11 mm day-¹ under RCP45 (See Table 4.15 for further details).

Relative changes in the monthly distribution of precipitation events between the two periods showed June to have the biggest negative changes with a -15.6% reduction in precipitation under RCP85 and a lesser reduction of -5.2% under RCP45 (Table 4.15). This was followed by August with a negative change of -13% under RCP45 and a lesser change of -3.7% under RCP85. The month of July also showed a decline in precipitation with -7.7% and -6.5% under RCP85 and RCP45 respectively. The model response in May was positive with an increase of 7.5% and 1.9% under RCP85 and RCP45 respectively. The model indicated a mixed response in September with

an increase of 7.6% under RCP85 and a reduction of -9.3% under RCP45. There was also a mixed response in October with an increase of 7% under RCP85 and a marginal decrease of - 0.7% under RCP45.

Overall, the core summer months (JJA) have shown considerable negative changes in summer precipitation and supports earlier results in this thesis (See Table 4.12, Section 4.4.5). In the same vein, the model simulation results also revealed a considerable reduction in the number of precipitation counts (i.e. wet days precipitation >0.05 mm day-¹) in future scenarios (2021-2050) compared to the baseline period of 1971-2000. The biggest changes in the number of precipitation counts occurred in September under RCP45 and RCP85 with 550 and 474 counts less than historical respectively. Other big changes also occurred in June, July and August (See Table 4.15).

Figure 4.19 illustrates the monthly variations of precipitation events and further suggests likely episodes of dry spells and heavy decline in precipitation in July and August particularly under RCP45.



Figure 4. 19: Variations in the monthly distribution of precipitation events and sizes under the historical phase, RCP45 and RCP85 from the MOHC (HadGEM2-ES) model.

4.6 Watering regimes (Irrigation)

4.6.1 Introduction

Generally, Eastern England precipitation like most of the UK is highly variable in terms of duration, distribution and events sizes. The analysis of precipitation is usually on a daily, monthly, seasonal or annual basis and generally uneven from place-to-place and region-to-region making precipitation forecasts more challenging and the agricultural industry is no exception. Therefore, understanding the precipitation characteristics and watering of crops in Eastern England is an important factor in the growth and development of agricultural crops in the field and under controlled conditions.

For sugar beet and other crops in general, precipitation or irrigation timing is very important in order to maximise yield. In the first season of the experiment, irrigation was carried out every other day based on climatological precipitation changes between the historical and future time windows. In the second season, irrigation was implemented to represent a more realistic distribution of precipitation events and sizes from the historical and future time windows. In this section of the thesis, details of the calculations for the watering regimes are presented. In the first season, the watering regime was based on the result of changes in the overall May-October mean precipitation; whilst the second season was based on the distribution of precipitation events and sizes for the two temporal windows. Irrigation of the plants was normally carried out in the mornings to enable plants to maximise available water when temperature and evapotranspiration are low. This is critical as crop yield can be adversely affected by under or over irrigation. Therefore, to maintain a healthy plant population, timely and adequate amount of water is vital. So, this Section deals with the amount of water administered to the plants over the two growing seasons.

4.6.2 Results of the climatological watering regimes calculations

For ease of reference, in the first season of the experiment, calculation of the "control" watering amount was based on the water requirements of sugar beet plants and the mean number of sugar beet growing days, as reported by (Brouwer & Heibloem 1986). The mean daily water value was
calculated in terms of mm day-¹, which was then converted into a volume in litres that would be applied to the plants by multiplying the area of the compost at the level of the surface (mm²) by the precipitation value (mm day-¹) to get a volume per day. The pots used were approximately cylindrical and during the experiments, the plants were watered every other day with two times this volume. The values in mm³ were converted to litres by dividing the results by 1,000,000. Using this method, the "control" watering regime was calculated as 0.230 L day-¹ multiplied by two to give a value of 0.46L every other day. However, for ease of measuring, a 1000 millilitre (ml) burette was used to deliver the water to the plants making the control watering regime 460ml.

Model results presented in earlier sections of this chapter of the May-October precipitation for the historical phase, RCP45 and RCP85 (e.g. Section 4.4.1 and Figure 4.13) identified the ensemble means of three CMIP5 climate models as representing observations very well. All the projections indicated that UK precipitation decreased in the models apart from RCP85 in the EC-Earth model which showed an increase. Of these models, MOHC (HadGEM2-ES) showed the largest negative changes in precipitation (See Tables 4.7 and 4.8). Therefore, and further to the reasons outlined above, the MOHC data was used as the basis for the "future" precipitation calculations so that a plausible but relatively extreme scenario is being investigated - this is a scenario that may stress the sustainability of the UK sugar industry and so, it is worth investigating.

Section 4.4.2, Tables 4.7, 4.8 and Figure 4.13 showed that the difference between the MOHC RCP45 and RCP85 experiments was statistically insignificant and is further displayed in Figures 4.20 and 4.21. As a result, two watering regimes were used for this season: "historical", or "control"; and "future" (i.e. the mean of RCP45 and RCP85). Statistical analysis showed a significant difference (reduction) of -15.8% between the "control" (1971-2000) and the "future" (2021-2050) regimes (see Table 4.8). This -15.8% reduction in precipitation from 1971-2000 compared to 2021-2050 was applied to the calculated watering amount for the "control" group to obtain the value for the "future" watering regime as 0.195 L day⁻¹, or 0.39 L every other day. These watering quantities were applied to the two watering regime groups from 7 June 2014 (i.e. when the plants reached their 10-12 leaf stage) until harvesting on 23 November 2014 (i.e. growing day 223, which was used in the calculation of the watering regime).



Figure 4. 20: Result of the daily May-October data from the historical (1971-2000), RCP45 (2021-2050) and RCP85 (2021-2050) from the MOHC model. The boxplot details are the same as in Figure 4.3.

95% family-wise confidence level



Figure 4. 21: Tukey multiple comparison of means test for the daily May-October precipitation (mm per day) for the historical, RCP45 and RCP85.

4.6.3 Results of the realistic distribution of monthly watering regimes calculations

The monthly precipitation data are of high information value when evaluating weather events during the growing season. The data are sufficient enough to robustly describe possible changes or trends from a long-term (i.e. 30 years) climate point of view. This section analysed the non-zero precipitation days (wet days) from the MOHC (HadGEM2-ES) four member ensemble model. The aim was to apply a realistic distribution of watering events to the plants based on the calculation of precipitation events for the control experiment as well as the RCP45 and RCP85 scenarios. Results indicated that there are no statistical significant differences between the precipitation totals for RCP45 and RCP85 (See Section 4.6.2) but the distribution of those events

differed significantly thereby making it important to assess whether the distribution of precipitation events affected the growth and yield of plants.

The first step was to calculate the number of watering days per month for each of the three experimental categories (Control, RCP45 and RCP85). This was done by calculating the percentage of wet days for each month in each of the categories and applying that to the number of days in each month. Thereafter, the amount of water to be applied on each watering event day was based on the distribution of precipitation from each category. Then, the data was split into percentiles which then allowed wet days to be allocated in the correct proportion to replicate the overall monthly precipitation distribution. Please see Appendix 1-5 for details. Results from this analysis showed that the control category had a higher number of smaller watering events while the RCP85 had a lower but larger watering events with RCP45 somewhere in between.

The distribution of these events meant that there are days when water is applied (i.e. wet days) and days when there is no application (i.e. dry days). Therefore, to determine the specific days when these differing watering events were applied, a random number generator was applied. This give rise to wet and dry periods that appeared randomly in the watering regime, similar to the real climate system. Appendix 1-5 shows the watering regime calculations used in the second growing season for the month of October. Only October is presented here as an example of the work and procedure of the analysis because of the largeness of the files. Moreover, it is worth pointing out that the analysis commenced in July with 24 days left in the calendar month because of delay in the planting date.

Chapter 5 – The Greenhouse plant experiments

5.1 Introduction

The impact of precipitation on growth and yield of agricultural crops generally can be described in two ways. Firstly, the state of the soil and the amount of water available for plants use. Secondly, excessive water demand from the leaves induced by the state of the atmosphere (i.e. high temperatures) can influence the growth and yield of crops. This chapter uses the results from the precipitation analysis in Chapter 4 to report on the application of the watering regimes in the greenhouse sugar beet plant experiments. The analysis enabled the evaluation of how changes in both seasonal and monthly precipitation affected sugar beet plants development and yield over two growing seasons. Changes in the plants' group for the "control" experiment (1971-2000) were compared to the plants' categories under future scenarios of RCP45 and RCP85 for 2021-2050.

In the first season, plant seeds were sown on the 15th April 2014 and in the second season seeds were sown on the 25th May 2015. The experiment in the first season entailed growing 150 sugar beet plants in the greenhouse and was categorised into two: historical and future experiments with 75 replicates each. It's noteworthy that for this season, the study only investigated the impacts of climatological changes on the plants yield by delivering the total growing season precipitation from May to October in a series of regular and equal watering events for the two different categories (See Section 4.6.2).

In the second season, 201 sugar beet seeds were sown in the greenhouse and the experiment was categorised into three: Control, RCP45 and RCP85 with 67 replicates each. The watering regimes were designed to replicate realistic distribution of precipitation events. The impacts of the watering regimes on plants' parameters were assessed over the two seasons and results are presented here for the above-ground and below-ground plant parameters for the different categories of plant experiments. The growth of the plants was measured with a tape rule and observations showed that the plants' leaf formation occurred early (see also Kenter et al. (2006); Scott & Jaggard (2000)) but increased steadily in multiples of two throughout the growing

season. The results in this chapter give an indication of the ultimate effect of the different watering regimes on the different plant categories.

Over the two growing seasons, plants were sown under the same condition (See Section 3.8.5) and thinned at their 4-6 leaf stages from two to one seedling per pot to encourage uniform establishment. Plants were late allocated into the control and future watering regimes and the different watering regimes were implemented after the plants had reached their 10-12 leaf growth stage and had started forming tubers. To account for natural variability in plant sizes, the plants assigned to each watering regime were selected to result in an equal distribution of plant sizes in each watering regime. Allocation of plants at this time was done to coincide with rainfall analysis from May to October for the study periods and also because the biggest precipitation changes are projected for the summer. Changes in precipitation from the analysis conducted were imposed on the plants in the future categories, which had a reduction in precipitation amount.

Subsequently, a number of non-destructive parameters were used to assess the yield potential of the plants over the growing season. The measures parameters include: the number of leaves on each plant, the height of the plants, the growth ratio and the leaf width. These parameters were measured every two weeks to enable the examination of the effect of water reduction on the plants' development and productivity; this can place yield in the context of the growing season examined. At the end of the experiment, destructive measurements were taken to determine the mean fresh weight of the tubers as harvested and when dried.

5.2 Climatological watering regime measurements

5.2.1 Above-ground measurements

A number of non-destructive parameters were used to assess the yield potential of the plants over the growing season including: the number of leaves; height of the plants (i.e. height of the tallest stem); the growth ratio of the plants (i.e. height divided by number of stems); leaf width (i.e. width of the widest leaf). Only the final values are presented here because these data give an indication of the ultimate effect of the different watering regimes (See Table 5.1 and Figure 5.1). In all cases, the "control" group had higher values than the "future" group but the difference between the groups was not statistically significant (See sub-Sections 5.2.1.1-5.2.1.4) over the two growing seasons.

5.2.1.1 Highest tip of the plants

The measurement of the highest tip of the plants also known as the height to node ratio (HNR) is used to determine the characteristics of plant growth and development. It measures the total height of the plants starting from the cotyledonary node (i.e. nodes beside each other at the base of the plants) to the terminal of the plant (i.e. highest tip on the plant). Result of the comparison of plants height between the control and future experiments did not show any significant difference (p-value 0.94). Table 5.1 shows the means of the final set of non-destructive measurements taken. The mean comparison between the experiments returned a mean of 51.0 cm \pm 5.5 for the control experiment and 50.9 cm \pm 5.2 for the future experiment with a p-value of 0.94 suggesting that there was no statistical significant difference between the two experiments (Table 5.1).

5.2.1.2 The number of leaves

The different growth rate of the leaves within and between experiments at different stages of development made it impossible to select leaves of a single plant as representative of the entire population. The growth rate of the canopies differed across the entire population with some plants producing more leaves than others. The control experiment had a greater number of leaves than the future experiment but this difference was not statistically significant as shown in Table 5.1 detailing the final set of non-destructive measurements taken. The mean number of leaves from the control experiment returned a mean of 21.8 ± 3.2 compared to the future experiment with a mean of 21.6 ± 3.0 and a p-value of 0.69.

5.2.1.3 Growth ratio

The growth ratio of the plants varies within and between the groups. It gives an indication of the vegetative and reproductive capacity of the plants. In this experiment, there was no statistical

significant difference between the growth ratio of the control experiment and the growth ratio of the future experiment. Results shown in Table 5.1 retuned a mean value of 2.6 cm \pm 0.3 for the control experiment and 2.6 cm \pm 0.3 for the future experiment and a p-value of 0.68.

5.2.1.4 Leaves width

The position of the leaves surface influences light interception, evaporation, photosynthesis, response to irrigation and plant growth (Blanco & Folegatti 2005). The leaves' width provides comparison of the plants sensitivity to environmental changes between the control and future experiments. Result here did not show any significant difference between the different experiments with the control experiment returning a mean of 13.9 cm \pm 1.8 and the future experiment with a mean of 14.0 cm \pm 1.6 with a p-value of 0.68 (Table 5.1).

In summary, the plants above-ground parameters measured under the climatological watering regime did not show any significant difference between the growth and development of the plants in the two different experiments.

	Control		Fut			
Parameters	Mean	Std	Mean	Std	P-value	Population
Highest tip of plants	51.0	5.5	50.9	5.2	0.94	75
Number of leaves	21.8	3.2	21.6	3.0	0.69	75
Growth Ratio	2.6	0.3	2.6	0.3	0.68	75
Leaves width	13.9	1.8	14.0	1.6	0.68	75

 Table 5. 1: Mean and Std. of the final measurements of non-destructive measurements from the control and future seasonal watering regimes.



Figure 5. 1: Boxplot of the climatological watering regime data from the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extreme of the plots represents the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and highest values in the data.

5.3 Below-ground measurements for the climatological experiment

5.3.1 Wet weight of yield under the climatological watering regimes

At the end of the season, all the sugar beet tubers were harvested on 23rd November 2014 (Day 223) and destructive measurements were taken to determine the mass of each tuber as harvested and when dried. During harvest, plant tubers were uprooted from the soil, washed, dried and weighed individually. Then, the leaves were cut off from the tubers which are of the most interest in this research. The tubers were then weighed individually without the leaves and the measurement reported as the "wet" weight. The mean "wet" weights of all plants were calculated for both experiments with the "control" having a mean tuber wet weight of 359.5g and the

"future" with 318.5g. Figure 5.2 shows the boxplot of the wet weight data and Figure 5.3 shows a histogram of the complete data set, which clearly have different distributions.



Figure 5. 2: Results of the tuber wet mass data analysis. Boxplot showing the tuber wet mass data from the control and future categories. The boxplot details are the same as for Figure 5.1.

An independent sample t-test was performed on the control and future data with the hypothesis that there was no difference in the mean tuber wet weight of the "control" and "future" watering experiments. Results showed that a statistical significant difference existed in the yield of the "control" and "future" experiments with a p-value of 0.03. The future category showed a reduction in yield compared to the control experiment.



Figure 5. 3: Histogram showing the distribution of the tuber wet mass data for the control and future categories.

5.3.2 Dry weight of yield under the climatological watering regimes

The dry mass was obtained through a laboratory experiment by cutting the tubers into small but similar sized pieces and drying them in an oven at 80°C to remove moisture from the tubers. Figure 5.4 shows a boxplot of the dry weight matter and Figure 5.5 show histogram of the control and future datasets. Statistical analysis of the dry weight showed that the control category had a mean of 95.2g (73.5% reduction from the wet weight) and the future category with a mean of 88.2g (72.3% reduction from the wet weight).

This result indicated a p-value of 0.11 with the null hypothesis that there was no difference between the groups. This implies that the statistical significance of this result is just outside the 10% level often applied to determine significance. This by implication suggests that the difference in mass is mostly a result of the moisture content in the tubers between the two watering regimes. Despite the lack of statistical basis for rejecting the null hypothesis, there are still differences worthy of comment. In particular, the largest tubers from the control category (i.e. greater than 150g) are absent from the future category and the mean for the future category is noticeably smaller.



Figure 5. 4: Results of the tuber dry mass data analysis. Boxplot showing the tuber dry mass data from the control and future categories. The boxplot details are the same as for Figure 5.1.



Figure 5. 5: Histogram showing the distribution of the tuber dry mass data for the control and future categories.

5.3.3 Soil moisture

Soil moisture is generally described as water that is held in spaces between soil particles and is usually determined by the water table and it also helps in the transportation of important nutrients through the plant (Arnold, 1999). Nutrients are drawn from the soil and used by the plants without which plant cells show evidence of stress through wilting of the leaves and stem. In the case of such events, the plants regain turgidity through irrigation. So, considering the variability in precipitation, it is important to assess soil moisture in this study. By measuring the soil moisture, comparative assessment was made between the control and future watering regimes.

The changes in precipitation represented by the two watering regimes indicated a statistical significant difference in soil moisture between the control and future watering regimes. It was observed that soil moisture increased with irrigation and is depleted by crop growth. The mean growing season (May-October) soil moisture data collected during the growing season is shown in Figure 5.6 and the mean monthly soil moisture data are presented in Figure 5.7. Result of the mean soil moisture comparison showed that the control soil moisture had a mean of $18.1\% \pm 2.7$ compared to the future with a mean of $16.1\% \pm 2.7$ (Table 5.2).

The percentage (%) difference between the two watering regimes was further assessed using the null hypothesis that there was no difference in the two groups. The result of the independent sample t-test conducted using a 95% confidence level showed a significant reduction in the level of soil moisture in the future category with a p-value of 8.7×10^{-6} . In short, the analysis revealed that the future group had a significant reduction in soil moisture. Table 5.3 shows the June-October analysis of the soil moisture measurement between the two categories and Figure 5.7 shows a line plot displaying the mean monthly soil moisture. Results reveal that July had the lowest soil moisture percentage in both the control and future watering regimes. In fact, both datasets showed the same trend indicating July as the month with the lowest soil moisture that may potentially have an impact on yields.

Soil moisture (%)						
Category	Mean	Std	Population			
Control	18.1	2.7	75			
Future	16.1	2.7	75			

 Table 5. 2: June-October soil moisture data of the growing season for the control and future watering categories calculated from the measurements taken every two weeks.

 Table 5. 3: Mean monthly soil moisture data for the control and future experiments during the growing season.

	Control		Fut	ure
Month	Mean	Std	Mean	Std
June	18.5	3.7	16.3	3.8
July	14.9	4.9	12.0	4.6
August	18.3	2.4	16.7	2.7
September	20.5	2.1	19.7	2.6
October	19.3	1.6	18.0	2.1



Figure 5. 6: Results of the soil moisture data analysis. The boxplot shows soil moisture data from the control and future categories.



Figure 5. 7: Line graph showing the mean monthly soil moisture measurement for the control (blue line) and future (maroon line) categories.

5.3.4 Correlation between soil moisture and wet weight of yield

In order to further examine the impact of soil moisture on wet weight of yield, the relationship

between soil moisture and wet tuber mass was examined using the Pearson Correlation test. Figure 5.8 displays the results which indicate that 43% of the variability in wet mass in the control category could be explained by the variability in soil moisture. Conversely, 57% of the variability in wet mass in the future category could be explained by the variability in soil moisture. In summary, as the soil moisture in both categories increases, the value of the tuber wet mass decreases. This suggests that there was a strong negative linear relationship between the yields and soil moisture in the experiment.



Figure 5. 8: Scatter plot showing the wet mass for individual tubers from the control (red circles; solid line) and future (blue triangle; solid line) categories plotted against the mean soil moisture data for each replicate.

5.3.5 Correlation between soil moisture and dry weight of yield

A further examination of the impact of soil moisture on dry weight of yield was carried out using Pearson Correlation test. Results indicate that 41% of the variability in dry mass of the control category could be associated with variability in soil moisture. Figure 5.9 displays the plot which shows that as the soil moisture in both categories increases, the value of the dry weight decreases suggesting a strong negative linear relationship between soil moisture and dry weight of the crop.



Figure 5. 9: Scatter plot showing the dry mass for individual tubers from the control (red circles; solid line) and future (light green triangle; solid line) categories plotted against the mean soil moisture data for each replicate.

5.4 Realistic distribution of precipitation watering regimes parameters measurement

5.4.1 Above-ground measurements

In the second season of the experiment, using the distribution of precipitation events and sizes watering regimes, a similar approach as the first season experiment was employed to measure plant parameters to keep the experiment unbiased. A number of non-destructive parameters were used to assess the yield potential of the plants over the growing season including: the number of leaves; height of the plants (i.e. height of the tallest stem); the growth ratio of the plants (i.e. height divided by number of stems); leaf width (i.e. width of the widest leaf). Only the final values are presented here because these data give an indication of the ultimate effect of the different watering regimes. The experiment implemented three watering regimes for the three categories (Control, RCP45, and RCP85) of plant experiment undertaken unlike the climatological experiment with two watering regimes (See Section 4.5, Table 4.15, Figures 4.21 and 4.22). Results in this Section varied among the different parameters measured but with no statistical significant difference except in the number of leaves where there was a significant difference between the control and future categories. Tables 5.4 and 5.5 show the analysis and Figure 5.9 show the distribution of the measured parameters from the dataset.

5.4.1.1 Highest tip of the plants

Result of the comparison of the highest tip between the control, RCP45 and RCP85 experiments did not show any significant difference. Table 5.4 shows the means of the final set of non-destructive measurements taken. The mean comparison between the experiments returned a mean of 35.5 cm \pm 6.9 for the control experiment, 33.9 cm \pm 6.4 and 33.3 cm \pm 6.2 for RCP45 and RCP85 experiments respectively (Table 5.4). The result returned a p-value of 0.15 suggesting that there was no statistically significant difference between the highest tips of the plants from the three experiments (Figure 5.9).

5.4.1.2 The number of leaves

The growth rate of the canopies differed across the entire population with some plants producing more leaves than others. Results indicate that the control experiment had more leaves than the RCP45 and RCP85 groups. Results of the analysis, shown in Table 5.4, indicate that there was a statistically significant difference between the number of leaves in the control category and the future experiments (RCP45 and RCP85). Multiple means tests using ANOVA returned a mean of 17.7 ± 3.0 for the control category, 16.6 ± 2.5 and 16.5 ± 2.5 for the RCP45 and RCP85 groups respectively with a p-value of 0.02 (See Table 5.4). This result suggests that there was a significant difference in at least, one of the categories between the three experiments. Therefore, Tukey multiple contrast of means test was carried out to confirm which categories are different from the other/s. Results shown in Table 5.5 confirm that there was a significant difference between the number of leaves in the control and RCP45 categories and Figure 5.9 shows the distribution from the dataset. Similarly, the control experiment also differed from RCP85 with a p-value of 0.02. There was however no significant difference between the number of leaves in RCP45 and RCP85.

5.4.1.3 Growth ratio

The growth ratio of the plants varied within and between the groups. Results showed that there was no statistical significant difference between the overall growth ratio of the control watering regime and the growth ratio of the RCP45 or RCP85 watering regimes. Result of the multiple means test shown in Table 5.4 retuned a mean value of 2.0 cm \pm 0.3 for the control watering regime, 2.1 cm \pm 0.4 for RCP45 and 2.0 cm \pm 0.4 for RCP85 and a p-value of 0.5. This result suggests that there was no statistical significant difference in the growth ratio of the three experimental categories.

5.4.1.4 Leaf width

Result of the leaves width measurements (i.e. widest point of the biggest leaf) did not show any statistical significant difference between the three different watering experiments. The control experiment returned a mean of 9.8 cm \pm 2.3, RCP45 with a mean of 9.4 cm \pm 2.1 and RCP85 143

had a mean of 9.2 cm \pm 2.0 and a p-value of 0.3 (Table 5.4).

In summary, the plants above-ground parameters measured under the monthly watering regimes did not show any statistical significant difference between the highest tips of the plants, the growth ratio and leaf width. However, there was a statistically significant difference between the numbers of leaves in the different categories.

 Table 5. 4: Mean and Std of the final non-destructive measurements from the control, RCP45 and RCP85

 Realistic precipitation distribution watering regimes.

	Con	trol	RCP45		RCP85			
Parameters	Mean	Std	Mean	Std	Mean	Std	Population	P-value
Highest tip of plants	35.5	6.9	34.0	6.4	33.3	6.2	67	0.15
Number of leaves	17.7	3.0	16.6	2.5	16.5	2.6	67	0.02
Growth Ratio	2.0	0.3	2.1	0.4	2.0	0.4	67	0.5
Leaves width	9.8	2.3	9.4	2.1	9.2	2.0	67	0.3

 Table 5. 5: Results of the Tukey analysis of contrasts means test from the parameters measured for the control, RCP45 and RCP85 experiments also illustrated in Figure 5.9.

 Control Data

Growth Ratio				
Category	Estimate	Std Error t	value	P-value
RCP45 - Control == 0	0.07	0.06	1.18	0.46
RCP85 - Control == 0	0.03	0.06	0.45	0.89
RCP85 - RCP45 == 0	-0.04	0.06	-0.73	0.75
Height of leaves				
RCP45 - Control == 0	-1.54	1.13	-1.36	0.36
RCP85 - Control == 0	-2.12	1.13	-1.88	0.15
RCP85 - RCP45 == 0	-0.58	1.13	-0.52	0.86
Leaf width	_			
RCP45 - Control == 0	-0.4	0.37	-1.11	0.52
RCP85 - Control == 0	-0.58	0.37	-1.59	0.25
RCP85 - RCP45 == 0	-0.18	0.37	-0.49	0.88
Number of leaves				
RCP45 - Control == 0	-1.13	0.46	-2.46	0.04
RCP85 - Control == 0	-1.19	0.46	-2.59	0.02
RCP85 - RCP45 == 0	-0.06	0.46	-0.13	1.01



Figure 5. 10: Boxplot of the Realistic precipitation distribution watering regime results from the control and future experiments. The thick black line represents the median (2nd quartile) of the distribution. The extremes of the boxes represent the 1st (bottom) and 3rd quartiles (top). The whiskers indicate the lowest and highest values in the data.

5.5 Below-ground measurements

5.5.1 Wet weight of yield under the distribution of watering events and sizes

The second season harvest followed the same procedure as the first season. All the sugar beet tubers were harvested on 23rd December 2015 (Day 255) and destructive measurements were taken to determine the mass of each tuber as harvested and when dried. During harvest, plant tubers were uprooted from the soil, washed, dried and weighed individually. Then, the leaves were cut off from the tubers and the tubers were then weighed individually without the leaves

and the measurement reported as the "wet" weight. The mean "wet" weight of all plants were calculated for all the experiments with the "control" group having a mean tuber wet weight of 153.4g, RCP45 with a mean of 130.8g and RCP85 with a mean of 113.3g (See Table 5.6). Figure 5.10 shows the distribution of the data from the Control, RCP and RCP85 and Figure 5.11 shows a histogram distribution of the complete datasets with different distributions. Figure 5.12 shows the contrast in means from the datasets.

Multiple means tests was performed on these datasets with the hypothesis that there was at least one difference in the mean tuber wet weight of the "control" and "future" experiments under RCP45 and RCP85. The calculations were based on a mean statistics and normality of data with 95% confidence interval. Results indicate a p-value of 0.01 which suggest that there is at least one statistically significant difference in the mean tuber wet weight from the three experiments.

In order to ascertain the difference that existed between the experimental categories, Tukey multiple comparison of means test was conducted and results show in Table 5.7 that there was no statistical significant difference between the wet weight of the control and RCP45 categories with a p-value of 0.22. Similarly, there was no difference between the wet weights of RCP45 and RCP85 experiments with a p-value of 0.42. However, there was a statistical significant difference between the wet weight of the control and RCP85 categories with a p-value of 0.01.

Category	Wet weight	Std	Population
Control	153.4	96.1	67
RCP45	130.5	76.5	67
RCP85	113.3	61.5	67

 Table 5. 6: Results of mean tuber wet weight for the growing season from the Control, RCP45 and RCP85 experiments.

Wet weight				
Category	Estimate	Std Error t	value	P-value
RCP45 - Control == 0	-22.95	13.69	-1.68	0.22
RCP85 - Control = 0	-40.08	13.69	-2.93	0.01
RCP85 - RCP45 == 0	-17.13	13.69	-1.25	0.42

 Table 5. 7: Tuber wet weight data analyses showing results of the Tukey multiple comparison of means test for the control, RCP45, and RCP85.



Figure 5. 11: Results of the tuber wet weight data analysis. Boxplot showing the tuber wet weight data from the control, RCP45 and RCP85 categories. The boxplot details are the same as for Figure 5.1.



Figure 5. 12: Histogram showing the distribution of the wet weight data for the control, RCP45 and RCP85 categories of the realistic precipitation distribution experiment.



95% family-wise confidence level

Figure 5. 13: Results of the Tukey test showing that the mean wet weight between RCP85 and the Control experiment is significantly different from the others.

5.5.2 Dry weight of yield under the distribution of watering events and sizes

Similar to the first season experiments, the dry mass was obtained through a laboratory experiment by cutting the tubers into small but similar pieces and drying them in an oven at 80°C to remove moisture from the tubers. The mean dry mass of all plants were calculated for all the experiments with the control having a final mean weight of 48.9g (68.1% reduction from the wet weight), RCP45 with a mean of 42.7g (67.3% reduction from the wet weight) and RCP85 with a mean of 38.3g (66.2 % reduction from the wet weight) - See Table 5.8. Figure 5.13 shows a distribution of the data from the Control, RCP45 and RCP85 and Figure 5.14 shows a histogram distribution of the complete datasets with different distributions. Figure 5.15 shows the contrast in means from the datasets.

Multiple mean tests were performed on these datasets with the hypothesis that there was at least one difference in the mean dry weight of the control, RCP45 and RCP85 experiments. The calculations were based on a mean statistics and normality of data with 95% confidence interval. Results indicate a p-value of 0.03 which suggest that there is at least one difference in the mean tuber dry weight from the three experiments.

Therefore, to confirm which category or categories that were different, Tukey multiple comparison of means test was conducted and results show in Table 5.8 that there was no statistical significant difference between the dry weight of the control and RCP45 categories with a p-value of 0.26. Similarly, there was no difference between the dry weights of RCP45 and RCP85 experiments with a p-value of 0.51. However, there was a statistically significant difference between the dry weight of the control and RCP85 categories with a p-value of 0.02.

Again, this result by implication suggests that the difference in dry weight mass is a result of the different moisture contents in the tubers of the different watering regimes. In comparison to the first season wet weight yield, the tubers in the second season were relatively smaller. The smallest tuber had a value of 105g from the future category in the first season, which was much bigger than the largest minimum tuber of 34.1g in the second season. Also, the smallest maximum tuber value of 547g from the first season was much higher than the highest tuber value of 437.1g in the second season. These results highlight the vulnerability of yield to changes in precipitation patterns that are likely under future scenarios in addition to the interannual

variability of yield patterns which are also likely to increase annual yield gaps and make yields less predictable.

Category	Dry weight	Std	Population
Control	48.9	27.9	67
RCP45	42.7	21.7	67
RCP85	38.3	18.3	67

 Table 5. 8: Results of mean tuber dry weight for the growing season from the Control, RCP45 and RCP85 experiments.

 Table 5. 9: Tuber dry weight data analyses showing results of the Tukey multiple comparison of means test for the control, RCP45, and RCP85.

Dry	weight				
Cat	egory	Estimate	Std Error	t value	P-value
RCP45 - 0	Control == 0	-6.2	3.97	-1.56	0.26
RCP85 - 0	Control == 0	-10.58	3.97	-2.67	0.02
RCP85 - 1	RCP45 == 0	-4.37	3.97	-1.1	0.51
Mean dry mass (g)			RCP45	RCP85	

Figure 5. 14: Results of the tuber dry weight data analysis. Boxplot showing the tuber dry weight data from the control, RCP45 and RCP85 categories. The boxplot details are the same as for Figure 5.1.



Figure 5. 15: Histogram showing the distribution of the tuber dry mass complete data for the control, RCP45 and RCP85 categories.



Figure 5. 16: Results of the Tukey test showing that the mean dry weight between RCP85 and the Control experiment is significantly different from the others.

5.5.3 Soil moisture

The changes in precipitation represented by the three watering regimes indicated a statistically significant difference in soil moisture between the control, RCP45 and RCP85 experiments. It was observed again that soil moisture increased with irrigation and is depleted by crop growth. Results of the July-October mean soil moisture measurements (Table 5.10) show that the control soil moisture had a mean of $20.7\% \pm 1.1$ compared to RCP45 with a mean of $19.6\% \pm 1.7$ and RCP85 with a mean of $19.5\% \pm 1.4$. The result of the mean monthly soil moisture data collected during the growing season is presented in Table 5.12 and Figure 5.16 shows the distribution of the soil moisture data while the mean monthly soil moisture data is presented in Figure 5.17.

The difference between the three watering regimes was further assessed using the null hypothesis that there was no difference in the soil moisture between the groups using ANOVA. Results of the mean July-October soil moisture indicated a significant difference in the soil moisture from the three watering regimes. Therefore, Tukey multiple comparison of means test was conducted using a 95% confidence level and the result showed a significant reduction in the level of soil moisture under future scenarios of RCP45 and RCP85 compared to the control. However, there was no difference between the soil moisture of RCP45 and RCP85. Table 5.11 shows the result of the multiple mean comparison test of the growing season (July-October) soil moisture analysis for the control, RCP45 and RCP85 categories, and Figures 5.17 - 5.18 show a line plot of the mean monthly soil moisture and the result of the Tukey multiple means test.

These results reveal monthly variability in soil moisture with the control category showing the highest monthly soil moisture in all the months from July-October. The highest monthly soil moisture for RCP45 was in October followed by July with August and September having the biggest monthly reduction in soil moisture. The RCP85 category showed that July had the lowest soil moisture value followed by August and September with October having the highest soil moisture. In all cases, and similar to the first season experiment, July and August showed the lowest soil moisture values.

Category	Soil moisture (%)	Std	Population
Control	20.7	1.1	67
RCP45	19.6	1.7	67
RCP85	19.5	1.4	67

 Table 5. 10: Result of the mean monthly (July-October) soil moisture data collected during the growing season for the control, RCP45 and RCP85 experiments.

 Table 5. 11: Soil moisture data analyses showing results of the Tukey multiple comparison of means test for the control, RCP45, and RCP85 watering regimes.

Soil moisture (%)				
Category	Estimate	Std Error	t value	P-value
RCP45 - Control == 0	-1.16	0.24	-4.82	1.00E-04
RCP85 - Control == 0	-1.29	0.24	-5.33	1.00E-04
RCP85 - RCP45 == 0	-0.12	0.24	-0.51	0.87

 Table 5. 12: Result of the mean monthly soil moisture data collected during the growing season for the control, RCP45 and RCP85 experiments.

Soil moisture (%)							
Category July August September October							
Control	19.5	19.5	20.2	22.2			
RCP45	19.4	17.9	17.9	20.6			
RCP85	18.1	18.7	19.5	20.8			



Figure 5. 17: Results of the soil moisture data analysis. Boxplot showing soil moisture distribution from the control, RCP45 and RCP85 categories.



Figure 5. 18: Line graph showing the mean monthly soil moisture measurement for the control (blue line), RCP45 (maroon line) and RCP85 (green line) categories.

95% family-wise confidence level



Figure 5. 19: Results of the Tukey test showing the differences in mean soil moisture between the three experiments.

5.5.4 Correlation between soil moisture and wet weight of yield

In order to further examine the impact of soil moisture on yield, the relationship between soil moisture and wet tuber mass was examined using the Pearson Correlation test. Figure 5.19 shows the correlation results which indicate that 37% of the variability in wet mass in the control category could be explained by the variability in soil moisture. Conversely, 55% of the variability in wet mass in the RCP45 category could be explained by the variability in wet mass could be explained by the variability in wet mass could be explained by the variability in wet mass could be explained by the variability in wet mass could be explained by the variability in soil moisture. Likewise, the RCP85 showed that 46% of the variability in wet mass could be explained by the variability in soil moisture. In summary, as the soil moisture in all the categories increases, the value of the tuber wet mass decreases. This suggests that there was a strong negative linear relationship between wet weight yields and soil moisture in the experiments. The implications of these results on wet weight have been described in Sections 5.3.2 and 5.5.2.



Figure 5. 20: Scatter plot showing the wet mass for individual tubers from the control (red circles; solid line), RCP45 (green triangle; solid line) and RCP85 (blue squares; solid line) categories plotted against the mean soil moisture data for each replicate.

5.5.5 Correlation between soil moisture and dry weight of yield

A further examination of the impact of soil moisture on yield was carried out using Pearson Correlation test. Results indicate that 14% of the variability in dry mass of the control category could be associated with variability in soil moisture. Conversely, 28% and 20% of the variability in dry mass of the RCP45 and RCP85 categories respectively, could be associated with variability in soil moisture. Figure 5.21 shows that as the soil moisture increases, the value of the dry mass decreases. This suggests that there was a strong negative linear relationship between soil moisture and dry mass of the crop.



Figure 5. 21: Scatter plot showing the dry mass for individual tubers from the control (red circles; solid line), RCP45 (green triangle; solid line) and RCP85 (blue squares; solid line) categories plotted against the mean soil moisture data for each replicate.

5.6 Heathrow weather station

5.6.1 Temperature

The inter-annual temperature obtained from the greenhouse weather station contained some gaps in the collected data and was therefore rendered unusable for this study. As a result, temperature data was obtained and extracted from Heathrow weather station which was the closest station to Brunel University London, the location of the greenhouse used in this study. This Section focused on mean monthly temperatures rather than daily scale temperature data because monthly mean temperatures reasonably capture the dynamics of daily scale in that months with many hot days tend to have high monthly mean temperatures. However, future work in this regard may wish to explore the daily temperature occurrence above certain critical threshold which was not addressed in this study.

In this Section, mean and maximum daily temperatures were calculated for 2014 and 2015 from January to December but with particular focus on March to October which is the focus of the analyses in this thesis. Overall, there are no significant differences in average temperatures between the two years with 2014 having a mean annual temperature of 12.9 °C (Std 5.2) and 2015 with 12.8 °C (Std 4.9). However, there are sharp variations in monthly temperatures throughout the year which is not unusual because of the different seasons during the year. Results in Tables 5.13 and 5.14 indicate high monthly variations in the boxplots from January to December. The lowest monthly temperature values in both years were recorded in the months of December, January and February (DJF) which were the coldest months with mean daily temperatures of 6°C to 7°C (Table 5.13) and daily maximum temperatures ranging from 8°C to 11°C (See Table 5.14).

In both years, the highest mean daily temperatures were recorded in the JJA months with temperatures ranging from 21°C to 26°C. These values are roughly comparable to the summer period in Eastern England which has temperatures ranging from 20°C to 23°C (Met Office 2016). The highest temperature recorded over the two year period (2014 and 2015) occurred in July 2014 which is consistent with observations of when the greenhouse plants were most challenged. The impact of high temperatures in the JJA months is further discussed in Section 6.4.1

Temperatures in both years showed seasonal and diurnal variations. Figures 5.20 and 5.21 show the variations in mean daily and daily maximum temperatures on a month by month basis with the highest and lowest temperatures recorded. The median, range and distribution of temperature values were much wider in 2015 than 2014 suggesting that temperatures were more consistent in 2014. This is equally consistent with daily research observations of lesser temperatures in 2015 characterised by long periods of wet spell and cloudy weather thereby reducing temperature

especially between January and March. Temperature improved in April but wider variations in between days.

Comparison of temperature in both years showed a sharp contrast at the beginning of the growing season in March with 2014 showing much higher temperatures than 2015. March, April and May (MAM) of 2015 was characterised by long periods of wet and cloudy weather and consequently low temperatures.

	2014		2015	
Months	Mean	Std	Mean	Std
January	7.14	1.6	4.44	3.6
February	7.3	2.1	4.63	2.4
March	9.73	2.8	7.28	1.8
April	11.22	1.7	11.73	2.4
May	13.49	1.7	12.9	1.9
June	17.18	1.6	17.38	2.5
July	21.07	2.3	18.24	1.8
August	17.1	1.1	18.1	2.3
September	18	1.1	14.08	1.4
October	14.41	2.5	12.4	1.6
November	8.06	2.8	10.79	3.7
December	6.3	4.6	11.8	2.1

 Table 5. 13: Mean analysis of the daily temperature for the two years of study.

	2014		2015	
Months	Mean	Std	Mean	Std
January	9.71	1.6	7.76	2.7
February	10.58	1.8	7.87	2.5
March	14.52	3.3	11.72	2.1
April	16.31	1.7	16.6	3.4
May	18.95	3.1	17.77	2.2
June	22.37	2.7	22.16	3.6
July	25.99	2.8	23.54	3.6
August	21.71	2.5	22.16	3.2
September	21.32	1.8	18.77	1.7
October	17.96	2.6	16.14	2.1
November	12.9	2.1	13.6	3.2
December	8.8	2.7	13.43	1.5

Table 5. 14: Mean analysis of the maximum temperature for the two years of study.



Figure 5. 22: Daily mean temperature distribution for 2014 (orange colour) and 2015 (light blue colour) from Heathrow weather station.


Figure 5. 23: Monthly mean temperature distribution for 2014 (orange colour) and 2015 (light blue colour) from Heathrow weather station.

5.7 Sugar Content Analysis

5.7.1 Total sugar content

Three samples from each dried sugar beet group (Control, RCP45 and RCP85) obtained from the laboratory experiment in 2015 were sent for testing for total sugar content of each dried sample. The test was carried out by ALS Food and Pharmaceuticals. ALS is a UK registered company with a high reputation for conducting a broad range of analytical testing services in areas of food & drinks, microbiology, pesticides, contaminants, dairy, water and pharmaceutical products. The company's wide accreditation and approval ratings suggest that their results could be relied upon for accuracy and integrity.

Generally, sugar percentage in freshly harvested sugar beet roots is between 14-20% of the fresh root weight (Rinaldi 2012) and is heavily influenced by the amount of water/ moisture content in the root. Since the determination of sugar is measured by the amount of sugar concentration in the tuber, it is important therefore, to remove the water/ moisture in the root leaving only the sugar for analysis. A similar approach to the one used in this study was conducted by Mohammadzadeh & Hatamipour (2010) who investigated the effect of drying conditions on dried sugar beet. Findings from their study showed that removing 90% of the initial water content from the beet was important in conserving the sugar content and essential properties during storage over a long period of time.

Selection of samples was based on similar dry weight values of samples from each group. The analysis was carried out using Ion Exchange Chromatography (IEC) method and results (Table 5.15) showed fairly uniform values ranging from the lowest in the control group with 69.6% (C14) to the highest in the RCP85 group with a value of 75.3%.

Table 5. 15: Dried sugar beet analysis for total sugar content from the Control experiment represented byC, RCP45 represented by M and RCP85 represented by H experimental groups.

Control			RCP45			RCP85			
Plant ID	Dry weight	Total sugar %	Plant ID	Dry weight	Total sugar %	Plant ID	Dry weight	Total sugar %	
C1	75.5	74.5	M26	74.8	72	H18	75.5	75.3	
C9	43.3	71.6	M64	43.4	74.4	H61	43.4	71.2	
C14	61.5	69.6	M17	61.2	73.5	H19	61.7	73.8	

Statistical analysis of the three samples showed control with a mean of 71.9% (Std 2.5), RCP45 with a mean of 73.3% (Std 1.2) and RCP85 with a mean of 73.4 (Std 2.1), and a P-value of 0.6 using a 95% confidence interval. The result indicates a difference between the percentage of sugar in the control and RCP45 and RCP85 but it is not significant at 95% confidence interval while there was no difference between the percentage of sugar in RCP45 and RCP85. Although, the small number of dried samples analysed could be argued to be unrepresentative of the entire population, it nonetheless provides an important indication on the percentage of sugar content in future sugar beet yield. This result implies that the difference in the experiments came from the wet weight of tubers as a result of the moisture content which can be inferred from the different watering regimes. The results in this thesis indicates that likely changes in future precipitation will not have an impact on the sugar content in future yields but will have a significant impact on the size and weight of individual tubers.

Results of the sugar content analysis are particularly relevant to sugar beet processors in terms of storage as most processors store beets for several months before processing. The challenge with long-term storage is the potential for decay before processing commences. This is because if tubers are peeled or scraped, it quickly oxidises and changes colour from white to black. This causes decay to the roots and a reduction in sugar level. An effective way to avoid this sort of decay is by drying the beet to remove the water content (Mohammadzadeh & Hatamipour 2010). For example, the dried beets used in this study were stored for a period of almost 10 months before sending them to the laboratory for further analysis. Additionally, dried beets will not decay nor reduce in its sugar content making it a win-win situation for the processors.

Chapter 6 - Discussion

6.1 Introduction

This study assessed how precipitation patterns changed in Eastern England from 1971-2000 and how they are likely to change between 2021-2050 using a suite of CMIP5 climate models to simulate future climate scenarios under RCP45 and RCP85. Furthermore, the study examined how the resultant changes in future precipitation may affect sugar beet yield in Eastern England by 2050. The findings from this study suggest that a potential change in future precipitation, as interpreted from the medium and high greenhouse gas emissions scenarios (RCP45 and RCP85) in an ensemble of MOHC (HadGEM2-ES) daily mean precipitation data, is likely to reduce sugar beet yield in the UK by 2050. Past (1971-2000) and future (2021-2050) modelled precipitation analyses of Eastern England suggest a reduction in future annual May-October precipitation, and therefore the threats to, and opportunities for, UK sugar beet farming are discussed in relation to possible precipitation reduction. More so, the study considered the fate of UK sugar beet farming in view of recent unusual seasonal precipitation events and how this will affect UK sugar beet farming that is solely 95% rain-fed and one of the principal aims of this thesis.

On a national scale, reduction in future precipitation will put a strain on water resources and this will become even more challenging with increasing water demands due to global population growth, economic development and increased urbanisation. Variations in precipitations such as the Cumbria floods in 2014 highlights the challenges faced by farmers whose crops and farmland were damaged, causing significant losses in terms of yield and income. Likewise, a persistent deficiency in the volume of precipitation can potentially lead to low Available Water Capacity (AWC) which will also be challenging for crop yields. Additionally, variability and changes in precipitation pattern makes it difficult for farmers to plan ahead for future growing season. These factors highlight an urgent need for the consideration of adaptation strategies that could be beneficial to farmers (Dorling 2014). In this regard, creative and innovative solutions for the management of water resources now and in future to enable farmers to be better prepared for adverse weather events will be a relevant approach. However, in order to develop such effective

management strategies, it is vital to understand the impact of changes in precipitation on the farmlands that provide society with food.

This study employed the use of climate models to predict the nature of future UK precipitation, the results of which could then be used to reduce the risk of crop yield losses by implementing new management strategies that will maximise yield potentials and farmers' income in future growing seasons. Model output was used in this study as a basis for quantitative comparison of past and future precipitation analyses using the growing season mean precipitation and changes in the distribution of precipitation events and frequency from CMIP5 climate models. This approach differed from previous field and controlled studies in that it was the first time that CMIP5 climate model ensembles have been used to inform a greenhouse experiment in controlled manner. In this section, the characteristics of past (1971-2000) and projected (2021-2050) Eastern England precipitations and the impact of changes in future precipitation on the sugar beet experiment conducted over two seasons (2014 and 2015) are discussed.

6.2 Eastern England climate

6.2.1 Weather stations observations

The ability to observe trends in precipitation data over a long period of time (i.e. 30 years) might be helpful in predicting periods of precipitation deficit or even, high/extreme precipitation. However, to understand the characteristics of Eastern England precipitation, daily mean precipitation data obtained from six weather stations in the sugar beet producing areas for the period 1971-2000 were analysed. Results from the stations data showed high variations in May-October daily mean precipitation over the 30 year period of study (See Figures 4.2 and 4.6). These types of variations make trends hard to detect (Met Office 2014), however, mean statistical analysis of the weather stations data showed that the precipitation characteristics of the stations showed good agreement across the region and the precipitation pattern from the stations exhibited similar annual trends. This simply means that the area under consideration was governed by the same regional precipitation mechanism and data could be relied on.

Excluding Manston (which showed a much smaller mean and median - See Section 4.2.1, Table 165

4.1), the rest of the stations were generally in broad agreement with the highest May-October mean precipitation recorded in Santon Downham with a mean of 1.78 mm day⁻¹ and the lowest mean of 1.61 mm day⁻¹ recorded in Writtle. In fact, two of the stations returned similar mean values of 1.76 mm day⁻¹ for Stansted Mountfichet and Coltishall. These results closely match the findings of Gregory et al. (1991) who reported a daily mean precipitation of 1.77 mm per day for Eastern England.

The ability to detect trends in precipitation records is quite an important one as it could help in predicting periods of deficient precipitation that could potentially lead to drought or intense precipitation that could potentially cause floods. The number of years necessary to detect any significant trends in precipitation data is a subject of continued debate but varies from one region to another due to the unpredictable nature of precipitation. Moreover, in some regions, this is compounded further due to poor data or equipment failure. For example, arid and semi-arid regions may require longer-term records as a result of poor data or the infrequent nature of precipitation in such areas. Eastern England however, does not have this problem because of the availability of continuous and long-term data provided by the UK Met Office.

The overall statistical analysis of the daily May-October mean precipitation observed data for the 30-year period (1971-2000) of study did not reveal any significant trend, positive or negative. This raises two questions: Is it possible that trends could be revealed if more years were included in the analysis? Or, could trends be detected over time if two different time frames were assessed? Results reported in Chapter 4 of this thesis showed a significant trend in daily May-October mean precipitation when two different time frames — one historical (i.e. 1971-2000) and a future prediction (i.e. 2021-2050) were assessed.

6.2.2 CMIP5 climate model simulation of precipitation under the historical phase

The assessment of annual May-October mean precipitation under the historical phase was very important in this study in order to evaluate model performance of Eastern England precipitation. Results of the annual May-October mean precipitation reported in Section 4.3.1 and illustrated in Figure 4.9 established that three out of the eight models used in this study showed broad agreement with the stations annual May-October mean precipitation data for the same period

(1971-2000). The aim of assessing model skill in order to quantitatively estimate and derive their weights is difficult according to Tebaldi & Knutti (2007). However, the comparison of historical precipitation data with observations have been reported to perform better than equal weighting of individual models (Krishnamurti et al. 2000). The use of models in this way have been reported in other studies to select desirable models for analysis (e.g. McSweeney et al. 2014; Brands et al. 2013) and gives credence for its use in this study.

The distribution of precipitation events from the models under the historical phase showed patterns consistent with observations and capturing key elements of Eastern England precipitation but with much wider variations. This suggest that the geographical patterns of extreme precipitation were not captured by the model which is normal as models do not simulate extreme precipitations (Maraun et al. 2010; Fowler & Ekström 2009). This issue was not considered to be a problem as the study here examined mean conditions rather than the extremes. Nevertheless, the distribution of the ensemble mean precipitation output for the historical phase of the MOHC (HadGEM2-ES) data is the closest to the stations observed mean reported in Section 4.3.1 and illustrated in Figure 4.9. This characteristic motivated the single use of the MOHC for future precipitation analysis and the greenhouse plant experimental watering regimes.

6.3 **Projected changes in future precipitation**

6.3.1 Changes in past and future Eastern England precipitation

In order to gain an understanding of likely changes to future precipitation, mean daily precipitation data from the best three models were analysed from May-October for the two different time slices under historical phase (1971-2000) and the future scenarios under RCP45 and RCP85 (2021-2050) were analysed. The ensemble means of individual models under the two different time frames were compared against each other (e.g. ensemble mean for the historical phase is compared with the ensemble mean under RCP45 and RCP85). Results indicate that Eastern England precipitation is projected to experience decreasing trends in May-October daily mean precipitation based on the two different approaches used in this study (i.e. seasonal mean precipitation and monthly distribution of events and sizes). This result supports similar

findings from previous studies of a reduction in UK summer precipitation (Met Office 2014; Fowler & Ekström 2009; Jenkins et al. 2009).

The three models used in this study replicated Eastern England precipitation fairly well. However, comparative annual May-October precipitation analysis between the historical phase and future scenarios of the models indicated differing results which simply implies that the magnitude of change depends on the scenario being assessed and the skill of the models used. This is simply because model performances differ under different conditions in different regions. The models captured key elements of Eastern England precipitation but the large variations in the distribution of precipitation events suggest that the geographical patterns of extreme rainfall were not captured by the models. However, this issue was not a problem because the study only considered mean precipitation. Among the three models used, the MOHC showed the biggest negative changes of 16.8% and 14.9% in precipitation under RCP45 and RCP85 respectively. This result could be viewed as a plausible but relatively extreme scenario that could be a potential source of threat to UK sugar beet farming and the sugar industry by extension. This result fulfils one part of the research hypothesis for a likely reduction in future Eastern England precipitation as explained in Section 1.6.

Although the output from the individual ensemble members showed slight differences but when combined together they clearly reflect a reduction in future May-October precipitation. This result is consistent with the result reported by Jenkins et al. (2009); Jenkins et al. (2008) of future reduction in UK summer precipitation and a reduction in the number of wet days. Perry & Hollis (2005) also reported similar reductions in summer precipitation nationwide. This is an important development for research into sugar beet farming and other agricultural crops in general in which their primary growing season is in the spring/summer time. The methods used in this study are also applicable to a range of agricultural crops whose growing season is in the summer and this study represents the first time that CMIP5 climate model data has been used to inform a greenhouse experiment.

Model projections of future climate scenarios in this study provide plausible information about what may be expected in the study region. It is noted in several past studies and successive IPCC reports that certain regions are particularly prone to the adverse effects of changes and variability in precipitation. Review of literature in this study identified Eastern England as one of such areas vulnerable to high variations in precipitation as a result of the mountains bordering Eastern England from the western parts of Scotland. Given the projection of continued increase in greenhouse gas emissions in the atmosphere, it is very likely that future Eastern England precipitation will continue to experience high variations in both annual and monthly precipitation. These sort of variations, when they occur could cause intensification of the hydrological cycle with associated changes in the frequency and magnitude of precipitation events such as extended periods of dry and wet spells (Williams et al. 2007; Huntington 2006). The scenario described here could be very challenging for agriculture particularly during the growing season. Analysis in Section 4.4.7 of this study suggests that this challenge will be more serious in the months between June and August, further analysed in Section 6.3.3.

6.3.2 Analysis of future total May-October wet day precipitation

Results of the relative comparison of Total May-October precipitation test for the two time windows (1971-2000 and 2021-2050) showed a decrease in total mean precipitation and a reduction in the number of wet days simulated from the models. The CCCma and MOHC indicated a reduction in total mean May-October precipitation but the EC-Earth showed no change (See Table 4.9). The reduction in number of wet day precipitation (See Table 4.10) suggests a trend towards drier summers. Changes from the models showed that CCCma had 231 and 366 less wet days than historical data under RCP45 and RCP85, respectively. The EC-Earth had 76 wet days less than historical under RCP45 and 6 wet days more than historical under RCP85. The MOHC also indicated reduction in the number of wet day events with 492 days less under RCP45 and 536 days less under RCP85 compared to historical wet day precipitation events.

Again, the MOHC showed the biggest negative changes in precipitation further justifying its use in this thesis. Evidence in Figure 4.13 again answers the research question in this study and clearly illustrate the reduction between historical and future precipitation from the models and raises questions about the future and sustainability of UK sugar beet production that is 95% dependent on precipitation. One notable factor in this study is that, the reduction in total May-October precipitation during the growing season (i.e. mean precipitation over the growing season) does not translate to a reduction in precipitation on a monthly basis during the growing season. This raises further questions of which months are most likely to be impacted by changes in precipitation? And could this change impact on the yield of crops grown during the season? The former question is addressed in the next section under the monthly distribution of precipitation while the later question is addressed in Section 6.4.

6.3.3 Monthly distribution of precipitation events and sizes.

As mentioned in the previous section, changes in precipitation is evident in the analysis of total May-October precipitation and there is need to identify the particular months that are likely to be affected by the changes. The assessment of monthly distribution of precipitation events provides a way to address this issue. Result in this section showed good agreement with monthly distribution of precipitation events from observations (See Figures 4.9 and 4.17) and the analysis under future climate scenarios (RCP45 and RCP85) also indicated large monthly variations shown in Figure 4.17. Likewise, Table 4.12 showed that the biggest changes occurred in the months from June to August. This is consistent with periods when temperature is predicted to be highest and further incidents of reduced or deficient precipitation will help to exacerbate the stress factors in relation to water resources.

The MOHC differed slightly from the other models and showed the biggest negative changes. Table 4.11 show the mean and standard deviation from the historical phase and future scenarios. The highest monthly change in precipitation was reported for October followed by June, and the lowest change in precipitation was reported for July (See Table 4.12). Overall, the biggest negative changes in precipitation from the models for the May-October monthly distribution of events and sizes indicated that July will be most challenging in terms of precipitation decrease.

Table 4.12 shows the level of changes for the individual months from the three models. Further examination revealed that there will be more dry spells under future scenarios especially in July and to a lesser extent in August. July and August showed the lowest mean precipitation values suggesting that these months will most likely be challenging for agricultural crops in terms of

water resources and water uptake for plants growth and development. This sort of scenario will be very challenging for farmers and water resource managers. By and large, water table will be affected which in turn impact on plants ability to uptake water from the soil for growth and development. The robustness of this result re-enforces the findings from past and current studies of a reduction in UK summer precipitation reported in Section 4.4.5 (See Simpson and Jones 2012; Wigley and Jones 1987). These studies specifically reported a reduction in July and August precipitation across the UK which is consistent with results in Section 4.4.7 and Table 4.14 of this study. July had the least mean precipitation in the growing season between the months of June to August. Likewise, Section 4.5.1 and Table 4.15 revealed large monthly variations with July and August indicating likely episode of dry spells under future scenarios. The likely implications this will have on future growing seasons is further discussed in Section 6.4.

6.3.4 Model simulation of precipitation changes

In Chapter 4, the study employed the use of eight CMIP5 climate models to simulate Eastern England precipitation by comparing the CMIP5 historical phase May-October mean precipitation data to the stations May-October mean precipitation data for the period 1971-2000. Three out of the eight CMIP5 climate models showed good agreement with mean precipitation data from the stations. This type of evaluation enabled the selection of realistic and desirable models that can replicate Eastern England precipitation well. Assessment of model performance in this manner demonstrate how and why less realistic models are excluded from precipitation analysis and have been used in previous studies such as McSweeney et al. (2014) who reported that models perform differently in different regions. In short, the analysis used the ensemble mean of individual models to determine how close the means were to observations. The results showed that the ensemble means of the CCCma, EC-Earth and MOHC models were the closest to the observations growing season (May-October) precipitation data and therefore considered the best performing models selected for use in this study.

Model projections of future May-October mean precipitation for the study region was conducted by comparing the ensemble means of the CMIP5 model historical phase (1971-2000) to the CMIP5 model future scenarios under RCP45 and RCP85 to determine if a trend or pattern existed between the two time frames. Model analysis revealed a negative trend in both annual and monthly precipitation (i.e. reduction in precipitation) in future scenarios. This simply means that there will be fewer wet day events and increased dry spell events under future climate change. The months of July and August were particularly identified as vulnerable to increases in temperature and deficient precipitation which most likely may affect summer grown crops. Moreover, the continued increases in temperature by 2050 will increase evaporation during the growing season, especially in the months of July and August thereby increasing the challenges of water availability for crop and other uses.

Result in this study raises questions regarding the viability of the sugar beet industry in the UK, particularly in terms of water resources. This is against a background of European Union (EU) policy changes that potentially undermine the economic model for the industry (Burrell et al. 2014). The combination of these challenges raises questions about the future of particular agricultural practices and, therefore, calls for creative and innovative adaptation strategies which are further discussed in Section 6.4. However, this will also depend on the impacts of climate change in other key growing regions, which have not been considered in this thesis.

In addition, and as pointed out in Chapter 4 of this study, future UK summer precipitation is likely to decline and is consistent with studies widely reported of a decline in UK summer precipitation (Met Office 2014; Christensen et al. 2010; Jenkins et al. 2008). However, model outputs showed much wider variations in the precipitation distribution compared to the historical phase and observations because they do not reproduce extreme events very well. In order to address this issue, the current study only considered seasonal means and mean distribution of precipitation events. The point suggested here however, is that future analysis should also attempt to include extreme events that were not addressed in this study. This could help reduce the variations between the model projections and the historical/ observations and also reveal changes in the summer that mean precipitation alone could not unveil.

Climate model provide the best examination of model performance and have shown great improvements over the decades, however, there are still some challenges in model simulation of some aspects of the climate system, such as clouds and aerosol, and uncertainty in the direction and magnitude of future greenhouse gas emissions (CCRA 2016; IPCC, Flato et al. 2013). In spite of this seemingly drawback, models have improved significantly from the time of the First Assessment Report (FAR) to the latest Fifth Assessment Report (AR5), and from the use of the

various MIP's to the latest suite of the CMIP5 climate models (See Section 3.4). This improvement has increased confidence in the use of models to simulate future climates based on models ability to reproduce past and current climates, although confidence is higher for temperature than it is for precipitation (IPCC, Flato et al. 2013). Figure 6.1 for example, illustrates the improvement in the various CMIP models for precipitation which shows an improvement in the correlation of annual mean precipitation between observations and model output. Precipitation output increased from 0.68 in the CMIP2 models to 0.77 in the CMIP3 and to 0.82 in the CMIP5 models (IPCC, Flato et al. 2013). The Figure further provides evidence of improvement in model performance by the increase in correlation for successive models. Model successes in this regard provide justification for its use in this thesis and it has proven reliable in successive IPCC reports including the AR5.



Figure 6. 1: Model capability in providing annual mean temperature and precipitation illustrated from the different phases of the CMIP2 (2000); CMIP3 (2005) and CMIP5 (2013). "The Figure shows the correlation (a measure of pattern similarity) between observed and modelled temperature (top panel) and precipitation (lower panel). Larger values indicate better correspondence between modelled and observed spatial

patterns and the black symbols indicate correlation coefficient for individual models. The large green symbol indicates the median value". Source: (IPCC, Flato et al. 2013). See -

https://www.ipcc.ch/pdf/assessment-report/ar5/wg1/WG1AR5_Chapter09_FINAL.pdf

Furthermore, the work carried out in this thesis has produced some firm conclusions but also thrown open some questions that subsequent studies may address. For example, the analysis of model performance could be extended to a wider range of models because the best performing models in one region may not necessarily perform well in other regions or may not even be the best when compared to some models not considered in this thesis. Nonetheless, the models have produced results that support previous studies in this field and also fulfil the first research hypothesis and answer the first research question in Section 1.6. For ease of reference, the research hypothesis and research question are re-stated here:

Hypothesis:

- Changes in future Eastern England precipitation patterns will impact on sugar beet productivity
- Reduction in future Eastern England precipitation will reduce sugar beet yield by 2050.

The same hypothesis may be approached differently using the following research questions:

- What are the potential impacts that changes in precipitation patterns will have on sugar beet yield in Eastern England by 2050?
- How will the sugar beet plant respond to the changes in precipitation?
- What types of mitigation and/ or adaptation measures can be implemented?

Analyses indicate that under future climate change scenarios, annual and monthly precipitation will decrease in Eastern England by 2050. The next Section addresses how reduction in May-October mean precipitation will affect future sugar beet yield in Eastern England.

6.4 Greenhouse Experiment

The previous Section reported a reduction in future Eastern England precipitation under future climate change scenarios (RCP45 and RCP85) and high potential for increased variability in inter-annual and inter-monthly precipitation that could potentially threaten sugar beet productivity in Eastern England. A degree of precipitation variability in Eastern England have been presented in the previous Section (Section 6.3), but more critically, substantial changes to the inter-annual and inter-monthly precipitation variability over time between the two frames (1971-2000 and 2021-2050) were found to be significant leading to the submission that inter-annual and inter-monthly variability in the growing season precipitation may threaten sugar beet and agricultural productivity in the UK. The following Sections present a discussion on the impacts of different watering regimes on the growth and yield of sugar beet under baseline conditions and future scenarios of RCP45 and RCP85.

Although, the method employed in this study is focused on sugar beet, its applicability cuts across a range of other agricultural crops that are particularly grown during the summer. For this reason, some parts of this discussion will cover agriculture in general as this is a very important sector in the UK, currently representing 71% of UK land-use (LWEC 2016), and producing 50% of the food consumed in the UK (LWEC 2016).

6.4.1 Sowing and emergence

The sowing, growing and harvesting of all sugar beet plants over the two growing seasons was carried out under the same environmental conditions, but with different watering regimes. In the first season, plant seeds were sown on the 15th April 2014 and the watering regimes were devised based on FAO recommendations of average water need for a sugar beet plant during the growing season (Brouwer & Heibloem 1986) and precipitation observations from weather stations in Eastern England, which is the dominant production region of sugar beet in the UK.

In the first season, under seasonal mean watering regimes, emergence and establishment was excellent with 298 pairs of cotyledonary leaves emerging out of 300 seeds sown. The first set of leaves emerged in pairs and served as a source of temporary food for the developing seedling.

The next sets of leaves are the true leaves which also emerged in pairs but alternate to each other. As the plants growth continued, the leaves began to emerge from the crown in pairs and in an alternate pattern. The leaves growth rate differs greatly from plant to plant which is why one plant cannot be used as representative of the entire population and could be attributed to sunlight variations in the greenhouse. All plants were sown under the same water management regime until the plants were categorised into different treatments; this occurred when they started forming tubers. Ideally, temperature and humidity would also have been controlled but, given that all the plants experienced the same conditions, the experimental design is sound in its aim to test the impact of different watering regimes.

In the second season, plant seeds were sown on the 25th May 2015 and the watering regimes were based on monthly distribution of precipitation events and sizes during the May-October growing season. Emergence and establishment was not as good as the first season with about 70% emerging cotyledonary leaves out of a total of 402 seeds sown. The growth of the leaves were not different from the first season experiment but general observations showed from the first season experiment that early sowing, adequate watering and radiation capture aided full canopy development with the leaves completely shading the pot circumference. Kenter et al. (2006) pointed out that achieving full canopy cover is likely to have helped improve plant and tuber growth and is quite consistent with observations in this study.

Comparison of plants' emergence and growth over the two seasons showed that plants in the second season did not quite achieve full canopy cover compared to the plants in the first season experiments. In the first season, the plants most likely benefitted from earlier sowing date, average spring time temperature and adequate rainfall enabling plants to capture much needed sunlight and radiation which enhanced full canopy development. This supports the result of Hoffmann & Kluge-Severin (2011) who compared the growth analysis of autumn and spring sown sugar beets and found an improvement in spring sown sugar beet as a result of higher temperatures in April. The second season saw spring time temperatures unusually cool with lots of rainfall and not enough sunshine and radiation inevitably affecting full canopy development during the season. The summer JJA months only witnessed a couple of extreme temperature events which did not really affect the plants because of their small sizes compared to the previous season. The impacts from the second season experiment demonstrate the importance of temperature, sunlight and early sowing in order to capture the necessary radiation needed to

attain full canopy cover. Earlier sowing on the 14th April 2014 captured more spring time radiation that aided leaf canopy cover than the late sowing on the 25th May 2015.

Studies such as Richter et al. (2006) showed that early sowing date during the months of March and early April resulted in yield gains of 0.2-0.3 t/ha, whereas, delay in sowing until the end of the month can cause yield loss of up to 1.4 t/ha. Given that future precipitation projections indicate a reduction in the summer, the combination of this with high summer temperatures will likely bring about shifts in the onset and length of the growing season by 2050 which in itself encourages earlier sowing. Results from this study have shown that earlier sowing in the first season yielded better than in the second season. This is supported by the findings of Crespo et al. (2011) who used maize production to demonstrate the possibility of adapting to projected climate shifts by 2050 by changing planting dates.

Overall, during the two growing seasons, plants in the different watering regimes were exposed to the same environmental conditions with plants in each watering regime evenly distributed throughout the greenhouse (See Figures 3.5 and 3.6). The amount of sunlight on different sides of the greenhouse varied, for example, but the systematic distribution of the members of each watering regime meant that there was minimal bias in such uncontrolled variables. Moreover, the parameter measurements only commenced after the plants had started forming tubers after their juvenile stages. Therefore, the real progress of the tubers can be estimated from the changes in the tubers from the different watering regimes and places yield in the context of the mean growing season conditions.

6.4.2 Event based impacts

Event based impacts resulting from changes in weather patterns such as high temperatures, had negative impacts on the plants. During high temperature events, the leaves wilted and went into early senescence; Lambers et al. (1988) reported that such water stress affects the growth and productivity of sugar beet and would have affected the plants in this study. The high temperatures in the month of June and July drove this problem, with leaves from the bigger plants wilting at the first sign of stress and the leaves from the smaller plants wilting later (Okom et al. 2017). This was reported by Hsaio (2000) in a separate study that showed large leaves are usually the

first to diminish at the first sign of water stress. Importantly, the wilting of the leaves did not affect one watering regime more than the other and therefore, the results of the experiments were not biased by the extreme weather events. In spite of this, plants from the different categories exhibited remarkable characteristics of adaptability in their high rate of recovery after watering following each stress episode. Figure 6.2 shows the impact of heat wave on the plants on the 17th July 2014 and Figure 6.3 shows the plants adaptive capacity after irrigation had been administered. The sort of impact described here in Figure 6.2 is reflected in Figures 5.7 and 5.17 which showed the impact that high temperature had on soil moisture in July and August.

Stressed Crops

Fri. 17 July Heatwave – Day 95 Day 95



Figure 6. 2: Plants from the different groups under stressful water condition as a result of heatwave.

After Irrigation

Day 97

Day 97



Figure 6. 3: Plants from the different groups showing remarkable recovery from stress after irrigation.

It is important to discuss wilting because the leaves capture the energy that is converted to sugar and, in so doing, play a key role in the final yield of the crops (Okom et al. 2017). Observations during the two growing seasons have shown that deficient precipitation and high summer temperatures had a negative impact on some plant leaves. Hsiao (2000) reported that a number of plant functions are affected under water stress conditions but the leaves are usually the first to be affected by wilting. Milford & Lawlor (1976) claimed that the younger leaves remain turgid until the stress becomes severe which is supported by observations from the current study. Other studies have shown that sugar beet can exhibit signs of retardation of leaf area increase emanating from temporary drought during the different stages of development. Choluj et al. (2004) reported a 6% reduction in relative water content of young and old leaves while Mohammadian et al. (2005) reported a loss of 14.1% in leaf area index of sugar beet plants as a result of water stress. Scott & Jaggard (1993) indicated in their study that one of the components to determine sugar beet yield is the amount of radiation it intercepts through the leaves and Choluj et al. (2014) more recently observed a 60% and 70% decrease in the leaf area index of some sugar beet genotypes as a result of water deficit compared to their control experiment.

The impact of water stress will be further compounded by predicted increase in temperature and rising levels of carbon dioxide (CO₂). By 2050, the atmospheric CO₂ concentration is likely to

exceed 500 ppm (IPCC 2013b), and all other things being equal, this increase may result in an increase in yields of C3 crops, including a 13% increase for sugar beet (Jaggard et al. 2010). However, the continued increase of CO_2 and its impact on other variables will, after a point, cause a decrease in the quality of the sugar beet (Myers et al. 2014). Additionally, future predicted increases in temperature by 2050 will increase evaporation during the growing season, especially in the months of June, July and August (JJA), which will be challenging for sugar beet production and will require further research into water resource management to maintain and sustain productions in order to maximise yields. Again, more complex experimental procedures, with more variables being controlled, could answer more complex questions but the results reported in this study are robust and address a fundamental issue in a controlled way.

6.4.3 Disease and pest

Over the two seasons, there were incidents of Powdery Mildew Disease (PMD) and Armyworm pest which attacked all plants regardless of categories. Plants in this study became infected with PMD through their leaves towards the end of summer. The outset of the disease appeared as tiny white spots on the front and back of the leaves which soon grew bigger and covered the surface of affected leaves. The disease spread quickly to other leaves around through wind dispersal and on each measurement recording date, the number of affected leaves was recorded. However, the severity of the disease was not much between the time of infection and harvest, and had no impact on the yield of the plants and was not presented in the results in this thesis.

The Armyworm pest attacked the plants in both seasons but were effectively monitored and controlled over the two seasons by manually removing the eggs and juvenile armyworms from the leaves as soon as they are noticed. As a result of close monitoring, the impact on the plants was considerably insignificant and had no bearing on yields and was not reported in the results. This sort of control measure may not be the case under field conditions with large acres of land and may therefore require more technical approach.

6.4.4 Above-ground parameter measurements

Results from the different watering regimes on above-ground plant parameters such as the height of plants (i.e. height of the tallest stem); the growth ratio of the plant (i.e. height divided by the number of stem); leaf width (i.e. width of the widest leaf) did not show any significant difference between the different watering regimes of the control experiment (1971-2000) and future scenarios (2021-2050). This indicates that the reduction in future May-October mean daily precipitation did not have a significant effect on the above-ground parameters measured in this study. Tables 5.1, 5.4 and 5.5 show the analysis of the different parameters.

6.4.5 Below-ground parameter measurements: Yield

The differential watering regimes is a condition that distinctly represents different levels of water availability in the experiments which is expressed in the yield. Over the two growing seasons, the yield of the different watering regimes showed a statistical significant difference in the mean wet tuber mass of the groups compared. In the first season, the yield of the different watering regimes showed a statistical significant (p<0.05) reduction in future wet tuber mass. Figures 5.2.and 5.3 show the boxplot and the distribution of the mean wet tuber mass from the control and future watering regimes respectively. The different watering regimes did result in a statistical significant impact on the root yield between the two groups with the control group having a mean tuber wet weight of 359.5g compared to the future group with 315.5g.

In the second season, the watering regimes were categorised into three watering groups of control, RCP45 and RCP85. The yield of the three watering regimes showed a significant reduction in the mean tuber wet mass of the RCP85 watering group compared to the control group; there was also a reduction in the mean tuber wet mass of the RCP45 watering group compared to the control but it was not significant at 95% confidence interval. However, there was no difference between the mean tuber wet mass of the RCP45 and RCP85 watering groups. Figures 5.10 and 5.11 showed the boxplot and the distribution of the mean tuber wet mass from the control, RCP45 and RCP85. The control group returned a mean tuber wet weight 153.4g compared to RCP45 with 130.5g and RCP85 with 113.3g. These results confirm the robustness

of the precipitation results in Chapter 4 of a reduction in precipitation between the historical phase and the future scenarios under RCP P45 and RCP85, and also confirming the lack of statistical significant difference in precipitation between RCP45 and RCP85. This difference in precipitation is reflected in the yield of the crop at the end of the season. This implies that under a reduced future precipitation, sugar beet yield will reduce by 2050 thereby fulfilling the second research hypothesis and answering the second and third research questions in this thesis.

Over the two growing seasons, the different watering regimes did result in statistical significant impact on the root yield (wet weight) between the different groups. The dry weight results are also in line with Kenter et al. (2006) who showed that dry root matter in field studies towards the end of a growing season depended on the availability of water in the soil. These results confirm that the experimental design had a direct impact on the growing environment, which was then reflected in the wet weight data. This is consistent with Richter et al. (2006), who modelled the response of UK sugar beet under climate change and found that water will be a major stress factor in future and relative soil moisture will be reduced under a high greenhouse gas emission scenario.

6.4.5.1 Dry yields of the plants

The dry mass of plants in both groups during the first season of the experiment did not indicate a significant difference (p=0.11). The control group returned a mean dry weight of 95.2g compared to the future group with 88.2g. This result implied that the difference in the tuber mass of both groups was to a certain extent, the result of water retention. There was nonetheless, a noticeable difference in the mean of the two groups, which would have been mostly linked to sugar content because once the water has been removed from the tuber; the majority of the remaining mass will be sugar. The dry mass of plants from the second season of the experiment showed a statistical significant reduction in the dry mass of RCP85 category compared to the control whereas, the RCP45 showed a reduction in dry weight but not statistically significant at 95% confidence level. There was no difference in the dry mass of the RCP45 and RCP85. Again, the differences indicated from the mean dry mass of the different groups suggest that the difference in yields emanates from the different watering regimes and as such, a reduction in future precipitation without a supplementary water input will be very challenging for sugar beet production in Eastern England.

6.4.6 Below-ground parameter measurements: Soil moisture

Shifts in weather patterns, annual and monthly precipitation variability, and changing soil conditions makes yield anything but certain. In this study, differential watering regimes was used to assess water supply, demand and availability for plants growth and yield, and results showed different impacts on soil moisture in the different experiments over the two growing seasons. It was found that soil moisture increased by application of irrigation and is depleted by plant growth processes. Variations in water supply to the plants in the different experiments depict actual distribution of precipitation events and sizes which is reflected in the percentage of soil moisture in each plant pot.

In the first season of the experiment, mean soil moisture measurement for the control group measured at 18.3% compared to 16.5% for the future group and in the second season, the control measured at 20.7%, RCP45 with 19.5% and RCP85 with 19.4%. A detailed examination of the monthly soil moisture between the groups revealed a reduction in future soil moisture under RCP45 and RCP85. Figures 5.7 and 5.17 reveal differences between the control experiments and the future experiments over the two seasons. The impact of the different watering regimes on soil moisture is the likely result of precipitation variability and high summer temperatures with associated increase in evapotranspiration and drying out of the soil.

This impact makes further statistical analysis necessary to assess the knock-on effect on yield. Analysis of the bi-weekly soil moisture measurements helped determine that a linear relationship existed between soil moisture and wet yield. In both seasons, results showed a strong negative correlation between yield and soil moisture. It is logical to assume that soil moisture could not have impacted on yield every month therefore monthly comparisons was carried out to find out which months were negatively affected by soil moisture. In the first season, July was implicated as the month that had the biggest negative impact on soil moisture and in the second season, July and August showed the biggest negative reduction. In both seasons, impacts were asymmetric between yield and soil moisture from both experiments and showed how much of the variability in yield is associated with variability in soil moisture as shown in Figures 5.8 and 5.19. The control and future categories in the first season showed that 43% and 57% of the variations in yield could be explained by the variations in soil moisture respectively. Similarly, in the second season experiment, correlation analysis revealed that 37% of the variability in yield of the control group was influenced by the variability in soil moisture, while under RCP45 and RCP85, 55% and 46% respectively showed that variability in yield was influenced by variability in soil moisture.

This result gives an indication that reduction in soil moisture level is consistent with reduction in future precipitation as illustrated from the different watering rates administered to the plants under future scenarios. This is quite practicable as the group with more water retained more soil moisture over the growing season. Therefore, going by future climate projections of reduced summer precipitation reported in such studies as Met Office (2014) and Jenkins et al. (2008), sugar beet farmers may need to consider other methods of supplementary water supply especially during the months of June, July and August when the plants are more likely to be challenged by high temperatures. This is particularly important because sugar beet productions in the UK is 95% rain-fed and therefore, presents a real need to consider adaptation options into how to make sugar beet productions sustainable and profitable in future.

Past study by Richter et al. (2006) on modelling the variability of UK sugar beet under climate change agreed that water will be a major stress factor in future and relative soil moisture will be reduced under high greenhouse gas emissions scenario. Although, this study was conducted 11 years ago, its findings are still very relevant, however, it did not consider the removal of EU capped quota system on sugar beet production. Therefore, after October 2017 when the capped system becomes effective, new approaches may be needed to maximise and sustain production in the UK.

A recent study conducted by El Chami et al. (2015) on the economics of irrigating wheat in Eastern England reported that the use of supplementary irrigation by farmers will be justified by increase in yields. The study asserts that the increment in yield from irrigation will be more beneficial in dry years and in reducing inter-annual yield gaps. Results from the current study aligns with the findings of El Chami (2015) in considering irrigation as a management option for

sugar beet farmers in order to remain viable in future growing seasons. However, no statistical evidence is presented here that suggests sugar content would increase with the implementation of irrigation but it does answer the third research question of what type of adaptation measure that could be implemented.

In summary, results in this study show that under a future warmer and drier summer, all other things being equal, yield will reduce unless other alternatives such as irrigation are considered. Investigations into the effect of other variables are also required. Nonetheless, the observations from the experiments also show that sugar beet is relatively resilient to increased temperatures and that the overall sugar content of the crop is not particularly sensitive to a moderate (16%) decrease in seasonal water availability.

6.5 Effects of climate change

6.5.1 Possible implications of future reduction in precipitation

The effect of reduced future May-October mean precipitation has implications for surface and underground water availability, groundwater levels and soil moisture. The variability and unpredictable nature of precipitation creates an added disadvantage if precipitation deficiency persists for long. The amount of precipitation that results in underground water for plants water intake varies throughout the year and recharge rates of underground water are at its lowest during the summer when temperatures are high resulting in more water being evaporated. This is supported by observations in this thesis during the high temperature events in July 2014 (See Figure 6.2 and 6.3) which showed the soils in the plant pots drying out causing the plants to wilt. The wilting of the plants in this study observed in the month of July particularly indicates the onset of water stress and a signal to commence irrigation immediately.

Generally, reduced precipitation will cause lower recharge rates and continuing discharge will cause the water table to drop making it more distant for plants to extract water for use. The persistence of this sort of situation will eventually cause soils to dry out with consequent impact on soil moisture. Future precipitation projections by the UK government estimates that 27-59 million people could be living in areas affected by water supply shortages by 2050s and that is 185

without considering increasing populations and rising water demands (DEFRA 2012). This type of situation can lead to drought and it is a known fact that all types of drought emanate from deficient precipitation which could have significant economic, environmental and social impacts, both direct and indirect. Some of the direct and indirect impacts are listed below:

Direct impacts that could emanate from deficient precipitation include but not limited to the following:

- Reduced crop yields
- Reduced water levels
- Increased chances of erosion
- Increased chances of plants diseases
- Loss of biodiversity

Indirect impacts include but not limited to the following:

- Reduced or loss of income to farmers
- Increased prices of food
- Creates unemployment
- Increase migration
- Increased crime and financial insecurity

A lot of these impacts occur in agriculture and related fields because of the reliance of agriculture on surface and sub-surface water, and this is against a backdrop of UK sugar beet farming which depends on 95% rain-fed production (British Sugar 2016). It is therefore important for sugar beet farmers in the UK, to not only consider alternative water management measures, but to also take advantage of possible opportunities that could reduce their vulnerability and exposure to climate risks with appropriate farm management practices that will ensure a successful yield at the end of the season.

In view of this sort of impacts, adapting to predicted changes in future precipitation is no more an option but an essential part of ensuring the sustainability of future yields. This thesis shows the nature and magnitude of future reduction in precipitation and the impacts it will have on future sugar beet yields and also highlights the need to develop effective adaptation strategies.

6.5.2 Possible implications of reduced precipitation on sugar beet plants

Results in this study focused on sugar beet but it is applicable to agriculture in general, particularly, crops grown in the spring through summer time. This study finds that a reduction in May-October mean precipitation have a number of implications for sugar beet farming and indeed agricultural productivity in Eastern England and the UK at large. Results here are significant for plants available water levels and soil moisture which is important for agricultural crops to flourish. Changes in future precipitation had a significant impact on future soil moisture which consequently impacted on future yields compared to the control experiment. This result is equally important for UK sugar beet farmers and the sugar beet processing industries because it identifies areas of potential challenges in order for them to adapt their management practices to ensure maximum future crop yield.

For any plant, water is important for growth and development, and variations in the delivery of precipitation have a significant impact on plant productivity (Pervin & Islam 2014; Gornall et al. 2010). Precipitation results indicated decrease in the months of July and August especially which affected soil moisture in these months as shown in Figures 5.7 and 5.17. Generally, Crop yield in any geographic area is determined by a number of factors including soil moisture which was found to have a negative linear relationship with wet yield. The amount of water required by sugar beet can be obtained as a cumulative water demand for the effective growing season because the water demand and supply have a bearing on the soil moisture positively or negatively. In this thesis, the soil moisture was found to have a significant negative effect on yield over the two growing seasons of the experiment.

The most significant physical indicator with regards to water stress on soil moisture are commonly identified through plant observation method such as the assessment of changes in plants physical characteristics such as the colour of the plants, curling of the leave and ultimately wilting. The wilting of plants as shown in Figure 6.2 is an indication of stress and a signal to commence irrigation immediately whether in the greenhouse or field conditions (Okom et al. 2017). With the sugar beet plants, as soon as irrigation is administered, the plants regain their turgidity showing the plants adaptive capacity to the event of water stress (See Figure 6.3). In reality however, if plants are not irrigated immediately at the onset of water stress, this may result in irreversible damage and hence yield loss. The work in this thesis provides farmers with

knowledge of what is possible in future climates in order to reduce risks, improve their decision making and enhance management practices in preparation for future growing seasons.

The current study through the use of irrigation to revive crops from water stress supports previous study on the use of irrigation as a relevant approach. Richter et al. (2006) considered the use of irrigation to be unprofitable as an alternative management option while El Chami et al. (2015), on the other hand, evaluated the economics of irrigating wheat in Eastern England and reported that the use of supplementary irrigation by farmers increased the yield of wheat. The study asserts that the increment in yield from irrigation will be more beneficial in dry years and in reducing inter-annual yield gaps. Results from the current study align with the result of El Chami et al. (2015) study in considering irrigation as a management option for sugar beet farmers in order to remain viable in future growing seasons.

In summary, results in this study show that under a future of warmer and drier summer, all other things being equal, yield will reduce unless other alternatives such as irrigation are considered. Moreover, further analysis of total sugar content in the experiments show that sugar beet is relatively resilient to increased temperatures and that the overall sugar content of the crop is not particularly sensitive to a moderate (16%) decrease in seasonal water availability as reported in Section 5.7.1. Therefore, investigations into the effect of other variables are also required given that the current study did not evaluate extreme events and temperature and radiation were not controlled. It is also suggested that future precipitation and agricultural crop assessment research incorporate these variables as input into agricultural crop model. Integrating climate model forecast data in crop models have been used in previous studies to predict crop yield in future growing seasons (Hansen et al. 2009; Mishra et al. 2008). Crop models simulate crop development, growth and yield, and could be used to control temperature, radiation and other climatic variables as well as investigate extreme events. This approach will help extend the results in this work (e.g. reveal the extent of impacts from extreme events on crop yield), and advance the relationship between crop yield and controlling climatic variables such as temperature and radiation.

6.5.3 Implications for food security

The review of literature showed that any change in climate will most likely affect agriculture positively or negative and the effects are already being felt in different parts of the world (Menzel et al. 2006). Mostly negative impacts have been detected where and when food prices increase following incidents of extreme events in food producing areas thereby affecting food security through food availability, access, utilization and stability (Ericksen 2008). Long-term incremental changes in climate are likely to alter productivity in food producing regions which are important for global food production. The UK generally does not fall under the category of countries that will be affected by food insecurity, however, the effects of climate change in other food producing regions could have likely impacts in the UK through international productions, trade and supply chains which could cause volatility in food prices (CCRA 2016).

Thornton et al. (2014) provides an analysis and evidence of tentative links between increasing climate variability and food security. Climate variability is the major cause of changes in precipitation patterns with adverse consequences for crop production and food security. Future trajectories of global food prices and food security are closely linked to future crop yields in food producing regions (Lobell et al. 2011; Lobell et al. 2009). Projections using crop models indicate a yield level-off or decline in many regions over the next few decades as a result of climate change and variability and, thus cause food security problems. Chapter 1, Figure 1.2 illustrates the relationship between changes in annual precipitation and annual sugar beet yield. The results indicated increasing yield gaps in the later years from 2007 to 2015 which is consistent with the high annual and monthly variations reported in Chapter 4. This type of relationship is important to investigate for two reasons:

- The economic importance of sugar beet to the UK rural economy providing about 13,000 jobs and producing 60% of the sugar consumed locally
- UK sugar beet production is 95% dependent on precipitation (British Sugar 2016). Therefore, a relative 16% reduction in precipitation is worth investigation.

Apart from the benefit to society, the economic motivation for sugar beet farmers generally is to fulfil the terms of their contract and earn income. So, for farmers involved in the business of sugar beet productions, their main concern then will be the size of the tubers and the percentage of sugar content in the tubers which are the main terms of their contract. However, shift is weather patterns, and annual/ monthly precipitation variability makes yield anything but certain as a result of precipitation variations, climatic impacts, soil conditions, pest and diseases. Analysis in this study has shown that both reduction in precipitation and precipitation variability have impacted negatively on sugar beet yield evident from the use of the different watering regimes. Furthermore, water supply and demand to the plants was based on availability of water to the plants in the different experiments and results showed similar impacts on soil moisture over the two growing seasons.

CCRA (2016) reported that there is the potential for domestic production to increase under a warmer climate but this opportunity may be constrained unless action is taken to address projected water deficit in most UK productive region. Eastern England is a good example because of the dry climate and the low precipitation rate in the region. An important economic opportunity for UK businesses may however, arise due to the increase demand for adaptation related goods and services.

The Committee on Climate Change – UK Climate Change Risk Assessment (CCRA) in their synthesis report for 2017, identified shortages in agricultural and public water supply as one of the top six areas of climate change risk in the UK. The report concludes that more action is needed in these areas now and in the long-term future to ensure sustainable supply in the driest months. The impacts of warmer temperatures on crop production will vary across the UK. For example, warmer, drier summers may have negative impacts on crop production in the east and south of the UK than the wetter areas of the north and west (LWEC 2016). Some studies have reported that warmer temperatures will accelerate the growth and development of plants, such as early flowering and increased length of the growing season (Thornton & Herrero 2010). However, there is the likelihood over time when the benefits of higher summer temperatures and longer growing season becomes outweighed by reductions in water availability in such a way that can affect agricultural productions negatively.

In addition to that, increasing CO_2 concentration in the atmosphere is predicted to generally increase the yield of certain C3 crops such as sugar beet although, these increases may vary depending on other factors such as water availability. This situation could very easily be escalated as a result of changes in precipitation patterns, increased evapotranspiration and reduced water availability for irrigation which can all threaten agricultural production particularly in dry areas such as Eastern England where precipitation is already in short supply. Furthermore, high temperatures during the growing season are also expected to increase evaporation and transpiration on agricultural farmlands which will affect the soils by making them to dry out until the underground water is recharged or irrigation is administered. So, under rain-fed conditions, as practiced here in the UK, deficiency in precipitation will be very challenging for plants because it can potentially lead to drought. For example, the first season analysis showed a 16% decrease in projected precipitation which may lead to an approximate reduction of 11% in wet weight yield. However, this level of reduction can be reasonably expected to be addressed by the use of irrigation, improvement in plant breeding and technological advancement.

The sustainability of future crop yields globally may depend on the ability of farmers to narrow the yield gap of crops between actual yield and potential. Possible adaptation strategies for the UK farmer will include irrigation and the use of reliable climate forecast to predict possible future climate change in order to target adaptation. This will help to reduce the problems on food production arising from climate change and hence food security.

Chapter 7 – Conclusions

7.1 Summary

Eastern England has high annual and monthly precipitation variability and the lowest amount of precipitation in the UK. Therefore, an evaluation of the present and future precipitation changes in the region was conducted using climate model simulations to provide a robust precipitation analysis. The main focus of this study was to firstly, understand the nature and characteristics of the climatological changes in Eastern England precipitation and secondly, investigate the potential impact of those changes in precipitation on sugar beet yield in the region; Eastern England being the primary production region for sugar beet in the UK.

Analysis of precipitation data from six weather stations in the study area showed high annual variations with all the stations exhibiting similar annual trends suggesting that similar precipitation regime prevailed over the region. Figure 7.1 show the same pattern of histogram distribution for the May-October daily mean precipitation (1971-2000) from the stations, further supporting the reliability of the data being representative of the region.



Figure 7. 1: Stations May-October daily mean precipitation (mm per day) from 1971-2000

Thereafter, seven CMIP5 climate model data under the historical phase was used to simulate observed precipitation for the period 1971-2000. This was important in order to verify model ability to simulate observed precipitation changes in the study area. Three out of the seven CMIP5 models (CCCma, EC-Earth and MOHC) reproduced observed precipitation very well and indicated high monthly and annual precipitation variations consistent with observations, and further contributing to the sensitivities and multiple stresses already present in the region. This makes it important to assess such variability and changes in precipitation characteristics when planning adaptation and mitigation measures for relevant sectors. More importantly, the precipitation results revealed a reduction in precipitation under future scenarios of RCP45 and RCP85.

Among the three reliable models used in this study, the ensemble mean of the MOHC was the closest to the observed means and was used in the calculation of the watering regimes used in the greenhouse plant experiment. The calculations were carried out using the MOHC (HadGEM2-ES) CMIP5 daily precipitation field of the mean of the medium and high greenhouse gas emissions scenarios (RCP45 and RCP85; 2021-205) with model output from the historical phase (1971-2000). The use of the latest suite of CMIP5 climate model ensemble to inform a greenhouse experiment in a novel way such as this has never been previously addressed in the literature. The key findings of this study are:

7.2 Wet weight yield

Irregularity in precipitation appeared to be the major source of variations in annual and monthly precipitation, as widely reported in the literature and consistent with results derived in this thesis (See Figures 4.1, 4.6 and 4.18). The experimental implementation of a 16% (i.e. mean of RCP45 and RCP85 in the first season of the experiment) water reduction to the sugar beet plants grown in the greenhouse implies that a reduced summer (May-October) precipitation will have a significant impact on wet sugar beet yield of the plant. The control plants had a mean tuber wet weight of 360g and the future with 319g which translates to about 11% decrease in wet yield with a p-value of 0.03. Table 7.1 shows the percentage (%) change in precipitation and reduction

in the number of wet day precipitation, and the impacts it had on yield of the crop at the end of the season.

Under the monthly distribution of precipitation events and sizes carried out in the second season of the experiment, Tables 4.15 and 4.18 reflect and show the variability in monthly precipitation. Analysis revealed that the months of June, July, August (JJA) are mostly impacted under the two future scenarios (RCP45 and RCP85). This is indeed consistent with projections predicting that the biggest changes will occur in the summer months (Met Office 2014; Jenkins et al. 2008). Table 7.2 shows the impact that changes in monthly precipitation had on the wet yield of the crop at the end of the season. Analysis showed that there was significant difference between the mean tuber wet mass of the crops in the control category and the mean tuber wet mass of the crops under RCP85.

In contrast, there was no significant difference between the mean tuber wet mass of crops in RCP45 and RCP85 categories re-enforcing the suggestion that water had a direct impact on yield. The yield results here indicated that the insignificant difference in the initial precipitation analysis calculated between RCP45 and RCP85 reported in Section 4.6.2 and displayed in Figures 4.19 and 4.20 supports the yield results (i.e. no significant difference between the wet weight yields of RCP45 and RCP85). Similarly, there was also no significant difference in the mean tuber wet mass between the control and RCP45 categories. Conclusively, significant decrease in precipitation between the two time frames had a significant impact on yield at the end of the season and where there are no significant differences in precipitation; yield appears not to be impacted. This result is very important when considering adaptation strategies for future crop yields

 Table 7. 1: Analysis of annual May-October change in mean precipitation between the historical phase (1971-2000) and RCP45 and RCP85 (2021-2050), and its impact on seasonal yield.

			Yields (g)		
Scenarios	Difference from historical (%)	Change in the number of wet day precipitation	Control	Future	
Historical	0%	0			
RCP45	-16.8	-492			
RCP85	-14.9	-536			
Mean of RCP45 & RCP85	-15.8	-514	359.5	318.5	

 Table 7. 2: Analysis of monthly distribution of precipitation events and sizes between the historical phase (1971-2000) and RCP45 and RCP85 (2021-2050), and its impact on seasonal yield.

	Difference from historical (%)			Change in the	Yields (g)				
Months	Historical	RCP45	RCP85	Historical	RCP45	RCP85	Control	RCP45	RCP85
May	0	2	7.5	0	-146	-251			
June	0	-5.2	-15.6	0	-121	-355			
July	0	-6.5	-7.7	0	-386	-437			
August	0	-13.1	-3.7	0	-378	-461			
September	0	-9.3	7.6	0	-550	-474			
October	0	-0.7	7	0	-296	-276	153.4	130.8	113.3

Furthermore, results of annual UK sugar beet yield measured against annual UK rainfall derived from DEFRA (2015) data showed that 2015 had less rainfall compared to 2014 and reflects the experimental yield variation in the two years. It's worth noting that this analysis supports the yield results obtained in this study with higher yields in 2014 than 2015. The results also confirm the importance and impacts of precipitation on sugar beet yield. The relationship between precipitation and annual yield of sugar beet illustrated in Chapter 1, Figure 1.2 show a good example of annual yield gap predicted to increase in future. This type of relationship is important to assess with regards to sugar beet in Eastern England for two reasons:

- Its economic importance to the UK economy, providing jobs and producing 60% of the sugar consumed locally (British Sugar 2016)
- UK sugar beet production is 95% dependent on precipitation (British Sugar 2016) and therefore very important to plan ahead in response to a constantly changing climate in order to remain viable and competitive particularly with the removal of the EU subsidy and capped quota system in addition to UK leaving the European Union.

The method employed in this study using CMIP5 climate models to inform a greenhouse crop experiment in a novel way can also be applied to a range of crops in other parts of the world. It can also benefit farming systems in other regions, particularly in less developed countries where reliable climate forecast is either limited or non-existent. The method of analysis used in this study can provide such regions with trust-worthy and reliable climate forecast that has the potential to identify vulnerable farming systems, minimise risks, improve management practices and put in place adaptation and mitigation measures that will help protect against climate change and variability. Brunet & Jones (2011) identified the lack of long-term data in some parts of the world as limiting a more robust climate assessment and the current study is robust enough to fill that gap.

7.3 Dry weight yield

Analysis of the dry weight yield obtained from the laboratory by removing moisture from the tubers showed that there was no significant difference in the control dry mass and future dry mass under the climatological watering regimes. The control category had a mean dry mass of 95.2g (73.5% reduction from the wet weight) and the future category had a mean dry mass of 88.2g (72.3% reduction from the wet weight). Statistical analysis indicated a p-value of 0.11 and seems to suggest that the difference in mass weight of the tubers is mostly a result of the moisture content in the tubers resulting from the different watering regimes. In essence, the plants that had more water produced bigger wet mass than plants with lesser water. In spite of this difference in watering regimes, there was no statistically significant difference in the dry mass of the plants between the control and future categories.

Contrastingly, analysis of the dry mass of the plants under the realistic distribution of monthly precipitation showed a significant difference between the dry mass of the control and RCP85 categories (See Section 4.6.3). There were no significant differences between the dry mass of Control and RCP45, and RCP45 and RCP85 categories respectively. Results are repeated here for ease of reference – The Control plants had a mean dry mass of 48.9g (68.1% reduction from the wet weight), RCP45 had a mean dry mass of 42.7g (67.3% reduction from the wet weight) and RCP85 had a mean dry mass of 38.3g (66.2% reduction from the wet weight). This result is key to understanding how the UK sugar beet industry will need to adapt to future climatic changes and work to determine what proportion of the yield that are linked or not linked to sugar content in the tubers.

Analysis of the total sugar content reported in Section 5.7.1 and Table 5.15 implies that there will be no difference in the total sugar content of sugar beet by 2050 in spite of the decrease in future precipitation. This result however is far from conclusive because of the small sample size examined. Future studies may use larger samples the size of the plant population in this study (i.e. 67 samples from each category) rather than the 3 samples from each category used in this study due to financial constraints. Although, the result here re-enforces the work of Choluj et al (2004) who reported that water stress did not affect the sucrose concentration in their study in spite of a 16-52% reduction in taproot and sugar yield.
7.4 Impacts on soil moisture

Changes in precipitation applied to the plant experiment over the two seasons showed significant negative impacts on soil moisture (See Sections 5.3.3 and 5.5.3). Variations in monthly precipitation indicates that soil moisture will be most challenged in the month of July (Figure 5.6) and supports experimental observations reported in Section 6.4.2 and shown in Figures 6.2 and 6.3.

Furthermore, results of soil moisture reported in Section 5.3.4 and Figure 5.8, Section 5.5.4 and Figure 5.19 respectively showed a strong negative linear relationship with yield and consistent with studies of Richter et al. (2006) who examined the impact of future drier climates on sugar beet yield under low and high emissions scenario. Results here reflects how changes and variation in precipitation are intertwined with changes in soil moisture and yield of sugar beet plants and is quite applicable to a range of other agricultural crop yields in any part of the world.

Reduction in future crop yield as reported in this study has important implications for global food security. Future trajectories of global food prices and food security have been closely associated with future crop yields in the food producing regions of the world (Lobell et al 2011, 2009).

7.5 Suggestions and future studies

- Results of projected precipitation simulation suggests that under a future warmer and drier summer, with all things being equal, water will be a stress factor for sugar beet production in Eastern England unless other alternatives such as irrigation are considered. However, further investigation will be required to assess the impact of irrigation on crop yield and the environment. The use of supplementary irrigation in this manner if economically viable will be justified by increase in yields
- The ability to adapt to changes and variations in precipitation by farmers will help to narrow the gap between actual and potential yield of crops. This will go a long way in addressing the issue of food security thereby making adaptation an important requirement

with regards to future crop yields

- Future research may focus on investigating other climatic variables not investigated in this study. For example, the use of agro-climatic model to investigate changes in precipitation and expand on the climatic variables to be investigated to include temperature, radiation and extreme events. The use of agro-climatic crop model in this way could provide a more detailed assessment of multiple climatic variables and its impacts on annual crop yields. This approach will help to extend the results in this thesis to reveal the extent of impacts from other variables including extreme events such as floods and drought on crop yield
- Increasing the sample size in the assessment of total sugar content in future experiments in order to carry out a detailed analysis of the impacts of changes in precipitation on sugar content in beets.

Glossary

AMIP	Atmospheric Model Intercomparison Project
AOGCM	Atmosphere Ocean General Circulation Models
AR4	Fourth Assessment Report
AR5	Fifth Assessment Report
AWC	Available Water Content
BADC	British Atmospheric Data Centre
BBRO	British Beet Research Organisation
Brexit	UK Exiting the European Union
CAP	Common Agricultural Policy
CC	Correlation Coefficient
CCCma	The Canadian Centre for Climate Modelling and Analysis
CCGCM	Canadian Coupled Global Climate Model
CCRA	Climate Change Risk Assessment
CET	Central England Temperature
CGCM	Couple General Circulation Model
CI	Couple General Circulation Model
CMIP	Coupled Model Intercomparison Project
CMIP3	Coupled Model Intercomparison Project Phase Three
CMIP5	Coupled Model Intercomparison Project Phase Five
СМР	Common Market Policy
CSIRO	Commonwealth Scientific and Industrial Research Organisation
DEFRA	Department for Environment, Food and Rural Affairs
EC-Earth	European Centre of Medium Range Weather Forecast Model
EFRA	Environment, Food and Rural Affairs Committee
ESM	Earth System Models
EU	European Union
FAO	Food and Agriculture Organisation of the United Nations
FAR	First Assessment Report
GCM	Global Climate Models
GHG	Greenhouse Gas
GPCP	Global Precipitation Climatology Project

HadCM2	Hadley Centre Coupled Model Version Two
HadGEM2-ES	Hadley Global Environment 2 Earth System
HadRM3	Hadley Centre Regional Model Version 3
HNL	Hertfordshire and North London
HNR	Height to Node Ratio of Plants
IEC	Ion Exchange Chromatography
IMechE	Institute of Mechanical Engineers
IPCC	Intergovernmental Panel on Climate Change
JJA	June, July and August
kg	Kilogram
LWEC	Living With Environmental Change
MAFF	Ministry of Agriculture, Fisheries and Food
MAM	March, April and May
Met Office	United Kingdom Meteorological Office
MIDAS	Met Office Integrated Data Archive System
MIP	Model Intercomparison Project
mm	Millimetres
MME	Multi Model Ensemble
MOHC	Met Office Hadley Centre
OECD	Organisation for Economic Co-operation and Development
PMD	Powdery Mildew Disease
ppm	Parts Per Million
RCM	Regional Climate Models
RCP	Representative Concentration Pathway
SRES	Special Report on Emission Scenarios
UK	United Kingdom
UKCP	United Kingdom Climate Projections
WFP	World Food Programme
WG1	Working Group 1 of the IPCC

References

- Ainsworth, E. A & Ort, D.R., 2010. How do we improve crop production in a warming world? *Plant physiology*, 154 (2), pp.526–530.
- Alexanderl, L. V., Zhange, X., Petersonn, T. C., Caesarl, J., Gleasonn, B., Klein Tankq, A., Haylockg, M., Collinsb, D., Trewinb, B., Rahimzadeha, F., Tagipoura, A., Ambenjek, P., Rupa Kumarm, K., Revadekarm, J., Griffithso, G., Vincente, L., Stephensonh, D., Burnh, J., Aguilard, E., Brunetd, M., Taylorp, M., Newf, M., Zhaic, P., Rusticuccii, M., J. L. Vazquez-Aguirrej, J. L., 2006. Global observed changes in daily climate extremes of temperature and precipitation. *Journal of Geophysical Research Atmospheres*, 111(5).
- Alexander, L. V & Jones, P. D., 2001. Updated precipitation series for the U.K. and discussion of recent extremes. *Atmospheric Science Letters*, 1(2), pp.142–150.
- Allan, R. P. & Soden, B. J., 2008. Atmospheric Warming and the Amplification of Precipitation Extremes. *Science*, 321(5895), pp.1481–1484.
- Allen, R. G., Pereira, L.S., Raes, D. & Smith, .D., 1998. Crop evapotranspiration: Guidelines for computing crop requirements, Available at: http://www.fao.org/docrep/X0490E/X0490E00.htm
- Allison, I., Bindoff, N. L., Bindschadler, R. A., Cox, P. M., de Noblet, N., England, M. H., Francis, J. E., Gruber, N., Haywood, A. M., Karoly, D. J., Kaser, G., Le Quéré, C., Lenton, T. M., Mann, M. E., McNeil, B. I., Pitman, A. J., Rahmstorf, S., Rignot, E., Schellnhuber, H. J., Schneider, S. H., Sherwood, S. C., Somerville, R. C. J., Steffen, K., Steig, E. J., Visbeck, M. & Weaver, A. J., 2009. The Copenhagen Diagnosis, 2009: Updating the World on the Latest Climate Science. The University of New South Wales Climate Change Research Centre (CCRC), Sydney, Australia, 60pp.
- Arnold, J. E., 1999. Soil moisture. Available at: https://weather.msfc.nasa.gov/landprocess/. [Accessed 17 June 2016].
- Asseng, S., Foster, I. & Turner, N. C., 2011. The impact of temperature variability on wheat yields. *Global Change Biology*, 17(2), pp.997–1012.
- Baker, N. C. & Huang, H. P., 2012. A comparison of absolute and relative changes in precipitation in multimodel climate projection. *Atmospheric Science Letters*, 13(3), pp.174–179.
- Barnett, D. N., Brown, S. J., Murphy, J. M., Sexton, D. M. H. & Webb, M. J., 2006.

Quantifying uncertainty in changes in extreme event frequency in response to doubled CO2 using a large ensemble of GCM simulations. *Climate Dynamics*, 26(5), pp.489–511.

- Bates, B. C., Kundzewicz, Z. W., Wu, S., & Palutikof, J. P., 2008. Climate Change and Water. Technical Paper of the Intergovernmental Panel on Climate Change, IPCC Secretariat, Geneva, 210 pp. Available at: https://www.ipcc.ch/pdf/technical-papers/climate-changewater-en.pdf
- BBRO, 2013. *BBRO Recommended varieties list 2013*, British Beet Research Organisation (BBRO), Norwich, UK.
- Blanco, F. F. & Folegatti, M. V., 2005. Estimation of leaf area for greenhouse cucumber by linear measurements under salinity and grafting. *International Water and Irrigation*, 25(4), pp.34–37.
- Brands, S., Herrera, S., Fern´andez, J. & Guti´errez, J. M., 2013. How well do CMIP5 Earth System Models simulate present climate conditions in Europe and Africa? *Climate dynamics*, 41, pp.803–817. Available at: http://link.springer.com/10.1007/s00382-013-1742-8%5Cnhttp://link.springer.com/article/10.1007/s00382-013-1742-8.
- British Sugar, 2016. British Sugar: A homegrown success story. Available at: https://www.britishsugar.co.uk/perch/resources/british-sugar-report15.0v2.pdf
- British Sugar, 2011. UK Sugar beet sustainability report. Available at: www.britishsugar.co.uk/../Beet_Sugar_Industry_Sustainability_Report_2011--update-06-14.pdf.
- Broecker, W. S., 1975. Climatic Change : Are We on the Pronounced Global Warming? *Science*, 189, pp.460–463.
- Brouwer, C. & Heibloem, M., 1986. Irrigation water management: Irrigation water needs. Food and Agriculture Organisation of the United Nations, Rome, Italy. http://www.fao.org/docrep/s2022e/s2022e00.htm
- Brown, J. R., Jakob, C. & Haynes, J. M., 2010. An evaluation of rainfall frequency and intensity over the Australian region in a global climate model. *Journal of Climate*, 23(24), pp.6504–6525.
- Brown, K. F., Messem, A. B., Dunham, R. J. & Biscoe, P. V., 1987. Effect of drought on growth and water use of sugar beet. *The Journal of Agricultural Science*, 109(3), p.421.
- Brunet, M. & Jones, P., 2011. Data rescue initiatives: Bringing historical climate data into the 21st century. *Climate Research*, 47(1–2), pp.29–40.
- Burrell, A., Himics, M., Doorslaer, B., Ciaian, P. & Shrestha, S., 2014. EU sugar policy: a

sweet transition after 2015? Report EUR 26530 EN, European Commission, Joint Research Centre: Sevilla.

- Cantelaube, P. & Terres, J. M., 2005. Seasonal weather forecasts for crop yield modelling in Europe. *Tellus, Series A: Dynamic Meteorology and Oceanography*, 57(3), pp.476–487.
- Carlson, L. & Bauder, J., (n/d). Sugar beet Agronomy 101. *Montana State University Press*. Available at: http://waterquality.montana.edu/farmranch/irrigation/other_crops/sugarbeet.html [Accessed July 15, 2015].
- Catto, J., Jakob, C. & Nicholls, N., 2013. A global evaluation of fronts and precipitation in the ACCESS model. *Australian Meteorological Oceanographic Journal*, 63, pp.191–203. Available at: http://users.monash.edu.au/~jcatto/publications/catto_lowres_AMOJ.pdf.
- CCRA, 2016. UK Climate Change Risk Assessment 2017. Synthesis Report. , pp.1–86. Available at: https://www.theccc.org.uk/wp-content/uploads/2016/07/UK-CCRA-2017-Synthesis-Report-Committee-on-Climate-Change.pdf.
- CGIAR, 2000. CGIAR Annual report 2000. The challenge of climate change: poor farmers at risk. Consultative Group on International Agriculture. Available at: https://cgspace.cgiar.org/bitstream/handle/10947/5437/cgar00txt.pdf?sequence=1&isAllo wed=y
- Chadwick, R., Boutle, I. & Martin, G., 2013. Spatial patterns of precipitation change in CMIP5: Why the rich do not get richer in the tropics. *Journal of Climate*, 26(11), pp.3803–3822.
- Chatterton, J., Viavattene, C., Morris, J., Penning-Rowsell, E. & Tapsell, S., 2007. Delivering benefits through evidence: The costs of the summer 2007 floods in protecting and improving the environment in England. Flood and coastal erosion risk management research and development programme. *Defra, Environment Agency*. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/291190/sch o1109brja-e-e.pdf
- Chen, C., Baethgen, W. E. & Robertson, A., 2013. Contributions of individual variation in temperature, solar radiation and precipitation to crop yield in the North China Plain, 1961-2003. *Climatic Change*, 116(3–4), pp.767–788.
- Chen, C. T. & Knutson, T., 2008. On the verification and comparison of extreme rainfall indices from climate models. *Journal of Climate*, 21(7), pp.1605–1621.
- Chmielewski, F. M. & Potts, J. M., 1995. The relationship between crop yields from an experiment in southern England and long-term climate variations. *Agricultural and Forest*

Meteorology, 73(1–2), pp.43–66.

- Chołuj, D., Wiśniewska, A., Szafrański, K. M., Cebula, J., Gozdowski, D. & Podlaski, S., 2014. Assessment of the physiological responses to drought in different sugar beet genotypes in connection with their genetic distance. *Journal of Plant Physiology*, 171(14), pp.1221–1230.
- Choluj, D., Karwowska, R., Jasińska, M. & Haber, G., 2004. Growth and dry matter partitioning in sugar beet plants (Beta vulgaris L.) under moderate drought. *Plant, Soil* and Environment, 50(6), pp.265–272.
- Chou, C., Chen, C., Tan, P. & Chen, K. T., 2012. Mechanisms for global warming impacts on precipitation frequency and intensity. *Journal of Climate*, 25(9), pp.3291–3306.
- Christensen, J. H., Kjellstrom, E., Giorgi, F., Lenderink, G. & Rummukainen, M., 2010. Weight assignment in regional climate models. *Climate Research*, 44(2–3), pp.179–194.
- Christensen, J. H., Hewitson, B., Busuioc, A., Chen, A., Gao, X., Held, I., Jones, R., Kolli, R.
 K., Kwon, W. T., Laprise, R., Magaña Rueda, V., Mearns, L., Menéndez, C. G.,
 Räisänen, J., Rinke, A., Sarr, A. & Whetton, P., 2007. Regional climate projections. In
 Climate change 2007: The physical science basis. Contribution of Working Group I to the
 Fourth Assessment Report of the Intergovernmental Panel on Climate Change. pp. 847–940.
- Christensen, O. B. & Christensen, J. H., 2004. Intensification of extreme European summer precipitation in a warmer climate. *Global and Planetary Change*, 44(1–4), pp.107–117.
- Colman, R. A., Moise, A. F. & Hanson, L. I., 2011. Tropical Australian climate and the Australian monsoon as simulated by 23 CMIP3 models. *Journal of Geophysical Research Atmospheres*, 116(10).
- Covey, C., AchutaRao, K. M., Cubasch, U., Jones, P., Lambert, S. J., Mann, M. E., Phillips, T. J. & Taylor, K. E., 2003. An overview of results from the Coupled Model Intercomparison Project. *Global and planetary change*, 37(1–2), pp.103–133.
- Cramer, W., Yohe, G. W., Auffhammer, M., Huggel, C., Molau, U., da Silva Dias, M. A. F., Solow, A., Stone, D. A. & Tibig, L., 2014: Detection and attribution of observed impacts. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., Barros, V. R., Dokken, D. J., Mach, K. J., Mastrandrea, M. D., Bilir, T. E., Chatterjee, M., Ebi, K. L., Estrada, Y. O., Genova, R. C., Girma, B., Kissel, E. S., Levy, A. N., MacCracken, S., Mastrandrea, P. R.,

& L.L.White, L. L. (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 979-1037.

- Crane, T. A., Roncoli, C., Paz, J., Breuer, N. E., Broad, K., Ingram, K. T. & Gerrit Hoogenboom, G., 2009. Seasonal climate forecasts and risk management among Georgia farmers. *Weather, climate and society*. Available at: https://journals.ametsoc.org/doi/pdf/10.1175/2009WCAS1006.1
- Crespo, O., Hachigonta, S. & Tadross, M., 2011. Sensitivity of southern African maize yields to the definition of sowing dekad in a changing climate. *Climatic Change*, 106(2), pp.267– 283.
- Croxton, P. J., Huber, K., Collinson, N. & Sparks, T. H., 2006. How well do the Central England Temperature and the England and Wales Precipitation series represent the climate of the UK? *International Journal of Climatology*, 26(15), pp.2287–2292.
- Dai, A., 2006. Precipitation characteristics in eighteen coupled climate models. *Journal of Climate*, 19(18), pp.4605–4630.
- DEFRA, 2015. Agriculture of the United Kingdom 2015. (Department for Environment, Food and Rural Affairs. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/557993/AU K-2015-05oct16.pdf
- DEFRA, 2014. Agriculture in the United Kingdom. (Department for Environment, Food and Rural Affairs. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/430411/auk -2014-28may15a.pdf.

- DEFRA, 2012. UK Climate Change Risk Assessment: Government Report, Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/69487/pb1 3698-climate-risk-assessment.pdf.
- DEFRA, 2004. Reassessing drought risks for UK crops using UKCIP02 climate change scenarios, Available at:

randd.defra.gov.uk/Document.aspx?Document=CC0368_2604_FRP.doc%0A.

- Deser, C., Phillips, A. S., Alexander, M. A. & Smoliak, B. V., 2014. Projecting North American climate over the next 50 years: Uncertainty due to internal variability. *Journal of Climate*, 27(6), pp.2271–2296.
- Dorling, S., 2014. Managing climate variability in agriculture: Using weather forecasting to support farming adaptation. *Outlook on Agriculture*, 43(1), pp.13–18.

- Duan, Q., Ajami, N., Gao, X. & Sorooshian, S., 2007. Multi-model ensemble hydrologic prediction using Bayesian model averaging. *Advances in water resources* (2007), 30 (5) 1371-1386.
- Durrant, M. J., Mash, S. J. & Jaggard, K. W., 1993. Effects of seed advancement and sowing date on establishment, bolting and yield of sugarbeet. *The Journal of Agricultural Science*, 121, p.333.
- EFRA Committee, 2014. *Winter Floods 2013/14*, London. Environment, Food and Rural Affairs Committee. Available at: https://www.publications.parliament.uk/pa/cm201415/cmselect/cmenvfru/240/240.pdf.
- Ehsanfar, S. & Modarres-Sanavy, S., 2005. Crop protection by seed coating. *Communications in agricultural and applied biological sciences*, 70(3), pp.225–9.
- El Chami, D., Knox, J. W., Deccache, A. & Weatherhead, E. K., 2015. The economics of irrigating wheat in a humid climate - A study in the East of England. *Agricultural Systems*, 133, pp.97–108.
- Emori, S. & Brown, S. J., 2005. Dynamic and thermodynamic changes in mean and extreme precipitation under changed climate. *Geophysical Research Letters*, 32(17), pp.1–5.
- Environment Agency, 2014. Environment Agency written evidence: Winter floods inquiry. Available at:

http://data.parliament.uk/WrittenEvidence/CommitteeEvidence.svc/EvidenceDocument/E nvironment, Food and Rural Affairs/Winter Floods/written/7992.html. [Accessed March 23, 2015].

- Ericksen, P. J., 2008. Conceptualizing food systems for global environmental change research. *Global Environmental Change*, 18(1), pp.234–245.
- Euractiv, 2013. Flooding cost farmers €1billion as damage tally continues. *Euractiv Network*. Available at: http://www.euractiv.com/Section/agriculture-food/news/flooding-costs-farmers-1-billion-as-damage-tally-continues/ [Accessed May 21, 2017].
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E., 2016. Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), pp.1937– 1958.
- FAO, 2016. Climate change and food security: risks and responses. Food and Agricultural Organisation of the UN, Rome. Available at: http://www.fao.org/3/a-i5188e.pdf
- FAO, 2009. Food and Agriculture Organisation, 2009. How to feed the World in 2050.

Proceedings of a technical meeting of experts, Rome, Italy, 24-26 June 2009. In *How to feed the World in 2050 Proceedings of a technical meeting of experts Rome Italy 2426 June 2009*.

- FAO, 1996. Rome Declaration on World Food Security and World Food Summit Plan of Action Rome Declaration on World Food Security.
- FAO, (n/d). Sugar beet: Crop water description and climate. Land and water. Food and Agriculture Organisation of the United Nations. Available at: http://www.fao.org/landwater/databases-and-software/crop-information/sugarbeet/en/
- Fernández-Donado, L., González-Rouco, J. F., Raible, C. C., Ammann, C. M., Barriopedro, D., GarcíaBustamante, E., Jungclaus, J. H., Lorenz, S. J., Luterbacher, J., Phipps, S. J., Servonnat, J., Swingedouw, D., Tett, S. F. B., Wagner, S., Yiou, P. & Zorita, E., 2013.
 Large-scale temperature response to external forcing in simulations and reconstructions of the last millennium' Climate of the Past, vol 9, no. 1, pp. 393-421. DOI: 10.5194/cp-9-393-2013.
- Fischer, E. M. & Knutti, R., 2016. Observed heavy precipitation increase confirms theory and early models. *Nature Climate Change*, 6(11), pp.986–991.
- Fischer, G., Sha, M., Tubiello, F. N. & van Velhuizen, H., 2005. Socio-economic and climate change impacts on agriculture: an integrated assessment, 1990-2080. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 360(1463), pp.2067–2083. Available at: http://rstb.royalsocietypublishing.org/cgi/doi/10.1098/rstb.2005.1744.
- Flato, G., Marotzke, J., Abiodun, B., Braconnot, P., Chou, S. C., Collins, W., Cox, P.,
 Driouech, F., Emori, S., Eyring, V., Forest, C., Gleckler, P., Guilyardi, E., Jakob, C.,
 Kattsov, V., Reason, C. & Rummukainen, M., 2013: Evaluation of Climate Models. In:
 Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to
 the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker,
 T.F., Qin, D., Plattner, G-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y.,
 Bex, V. & Midgley, P. M. (eds.)]. Cambridge University Press, Cambridge, United
 Kingdom and New York, NY, USA.
- Flowerdew A, 2012. *Calibration and combination of medium-range ensemble precipitation forecasts*, Available at: http://www.metoffice.gov.uk/media/pdf/h/6/FRTR567.pdf.
- Fowler, H. J. & Ekström, M., 2009. Multi-model ensemble estimates of climate change impacts on UK seasonal precipitation extremes. *International Journal of Climatology*, 29, pp.385– 416.

- Fowler, H. J. & Kilsby, C. G., 2003. Implications of changes in seasonal and annual extreme rainfall. *Geophysical Research Letters*, 30(13).
- Frei, C., Scholl, R., Fukutome, S., Schmidli, J. & Vidale, P. L., 2006. Future change of precipitation extremes in Europe: Intercomparison of scenarios from regional climate models. *Journal of Geophysical Research Atmospheres*, 111(6).
- Friedlingstein, P., Andrew, R. M., Rogelj, J., Peters, G. P., Canadell, J. G., Knutti, R., Luderer, G., Raupach, M. R., Schaeffer, M., Van Vuuren, D. P., 2014. Persistent growth of CO2 emissions and implications for reaching climate targets. *Nature Geoscience*, 7(10), pp.709–715. Available at: http://dx.doi.org/10.1038/ngeo2248%5Cnhttp://www.nature.com/doifinder/10.1038/ngeo2248.
- Gaetani, M. & Mohino, E., 2013. Decadal prediction of the sahelian precipitation in CMIP5 simulations. *Journal of Climate*, 26(19), pp.7708–7719.
- Garner, G., Hannah, D. M. & Watts, G., 2017. Climate change and water in the UK: Recent scientific evidence for past and future change. *Progress in Physical Geography*, 41(2), pp.154–170.
- Gates, W. L., Boyle, J. S., Covey, C., Dease, C. G., Doutriaux, C. M., Drach, R. S., Fiorino, M., Gleckler, P. J., Hnilo, J. J., Marlais, S. M., Phillips, T. J., Potter, G. L., Santer, B. D., Sperber, K. R., Taylor, K. E. & Williams, D. N., 1999. An Overview of the Results of the Atmospheric Model Intercomparison Project (AMIP I). *Bulletin of the American Meteorological Society*, 80(45), pp.29–55.
- Geng, S., Penning de Vries, F. W. T. & Supit, I., 1986. A simple method for generating daily rainfall data. *Agricultural and Forest Meteorology*, 36(4), pp.363–376.
- Giannini, A., Biasutti, M., Held, I. M. & Sobel, A. H., 2008. A global perspective on African climate. *Climatic Change*, 90(4), pp.359–383.
- Giorgi, F., Bi, X. & Pal, J. S., 2004. Mean, interannual variability and trends in a regional climate change experiment over Europe. I. Present-day climate (1961-1990). *Climate Dynamics*, 22(6–7), pp.733–756.
- Gleckler, P. J., Taylor, K. E. & Doutriaux, C., 2008. Performance metrics for climate models. *Journal of Geophysical Research Atmospheres*, 113(6).
- Godfray, C. H. J. & Garnett, T., 2014. Food security and sustainable intensification Food security and sustainable intensification. *Philosophical Transactions of the Royal Society:B*, 369(February), pp.6–11.

- Gornall, J., Betts, R., Burke, E., Clark, R., Camp, J., Willett, K. & Wiltshire, A., 2010.
 Implications of climate change for agricultural productivity in the early twenty-first century. *Philosophical Transactions of the Royal Society of London B: Biological Sciences*, 365(1554), pp.2973–2989. Available at: http://rstb.royalsocietypublishing.org/content/365/1554/2973
- Gregory, J. M., Jones, P. D. & Wigley, T. M. L., 1991. Precipitation in Britain: An analysis of area-average data updated to 1989. *International Journal of Climatology*, 11(3), pp.331– 345.
- Gregory, P. J., Ingram, J. S. I. & Brklacich, M., 2005. Philos Trans R Soc Lond B Biol Sci. 2005 Nov 29; 360(1463): 2139–2148. doi: 10.1098/rstb.2005.1745. Available at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1569578/
- Gu, G., Adler, R. F., Huffman, G. P. & Scott, C., 2007. Tropical rainfall variability on interannual-to-interdecadal and longer time scales derived from the GPCP monthly product. *Journal of Climate*, 20(15), pp.4033–4046.
- Halmstad, A., Najafi, M. R. & Moradkhani, H., 2013. Analysis of precipitation extremes with the assessment of regional climate models over the Willamette River Basin, USA. *Hydrological Processes*, 27(18), pp.2579–2590.
- Hansen, J. W., Mishra, A., Rao, K. P. C, Indeje, M. & Ngugi, R. K., 2009. Potential value of GCM-based seasonal rainfall forecasts for maize management in semi-arid Kenya. *Agricultural Systems*, 101(1–2), pp.80–90.
- Hansen, J., Johnson, D., Lacis, A., Lebedeff, S., Lee, P., Rind, D. & Russell, G., 1981. Climate impact of increasing atmospheric carbon dioxide. *Science (New York, N.Y.)*, 213(4511), pp.957–966.
- Hao, Z., AghaKouchak, A. & Phillips, T. J., 2013. Changes in concurrent monthly precipitation and temperature extremes. *Environmental Research Letters*, 8(3), p.34014.
- Harvey, L. D. D., 2004. Characterizing the annual-mean climatic effect of anthropogenic CO2 and aerosol emissions in eight coupled atmosphere-ocean GCMs. *Climate Dynamics*, 23(6), pp.569–599.
- Haugen, J. E. & Iversen, T., 2008. Response in extremes of daily precipitation and wind from a downscaled multi-model ensemble of anthropogenic global climate change scenarios.
 Tellus, Series A: Dynamic Meteorology and Oceanography, 60 A(3), pp.411–426.
- Hawkins, E. & Sutton, R., 2009. The potential to narrow uncertainty in regional climate predictions. *Bulletin of the American Meteorological Society*, 90(8), pp.1095–1107.

- Hennessy, K. J., Gregory, J. M. & Mitchell, J. F. B., 1997. Changes in daily precipitation under enhanced greenhouse conditions. *Climate Dynamics*, 13, pp.667–680.
- Hirota, N. & Takayabu, Y. N., 2013. Reproducibility of precipitation distribution over the tropical oceans in CMIP5 multi-climate models compared to CMIP3. *Climate Dynamics*, 41(11–12), pp.2909–2920.
- Hoffman, C. M. & Kluge-Severin, S., 2011. Growth analysis of autumn and spring sown sugar *beet. European Journal of Agronomy* 34(1): 1-9. doi:10.1016/j.eja.2010.09.001
- Houghton, J. T., Ding, Y., Griggs, D. J., Nouger, M., van der Linden, P. J., Dai, X., Maskell,
 K. & Johnson, C. A., 2001. Climate Change 2001: The Scientific Basis. *Climate Change* 2001: The Scientific Basis, p.881.
- Houghton, J. T., Jenkins, G. J. & Ephraums, J. J., 1990. Climate Change The IPCC Scientific Assessment. *Ipcc*, 1(1), p.414. Available at: https://www.ipcc.ch/ipccreports/far/wg_I/ipcc_far_wg_I_full_report.pdf.
- Howden, S. M., Soussana, J-F., Tubiello, F. N., Chhetri, N., Dunlop, M. & Meinke, H., 2007.
 Adapting agriculture to climate change. *Proceedings of the National Academy of Sciences*, 104(50), pp.19691–19696. Available at: http://www.pnas.org/content/104/50/19691.abstract
 http://www.pnas.org/content/104/50/19691.full.pdf.
- Hsaio, T. C., 2000. Leaf and root growth in relation to water status. Hortic. Sci. 35: 1051-1058
- Hulme, M. & Jenkins, G. J., 1998. Climate change scenarios for the UK: scientific report. UK Climate Impacts ProgrammeTechnical Report no. 1, Climatic Research Unit, Norwich.
- Huntington, T. G., 2006. Evidence for intensification of the global water cycle: Review and synthesis. *Journal of Hydrology*, 319(1–4), pp.83–95.
- IMechE, 2013. Global food: Waste not, want not. Institute of Mechanical Engineers (IMechE). Available at: https://www.imeche.org/policy-and-press/reports/detail/global-food-wastenot-want-not
- IPCC, 2014a. Summary for Policymakers. In: *Climate Change 2014: Impacts, Adaptation and Vulnerability.Part A:Global and Sectoral Aspects.Contributions of the Working Group II to the Fifth Assessment Report* of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L.White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp.1–32.

- IPCC, 2014b. Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Edenhofer, O., Pichs-Madruga, R., Sokona, Y., Farahani, E., Kadner, S., Seyboth, K., Adler, A., Baum, I., Brunner, S., Eickemeier, P., Kriemann, B., Savolainen, J., Schlömer, S., von Stechow, C., Zwickel, T.and Minx, J. C. (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC, 2014c. Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team, Pachauri, R. K. and Meyer, L. A. (eds)]. IPCC, Geneva, Switzerland, pp 151.
- IPCC, 2013a. IPCC, 2013: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. [Stocker, T. F., Qin, D., Plattner, G-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y., Bex. V. and Midgley, P. M. (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA. Available at: http://www.ipcc.ch/pdf/assessment-report/ar5/wg1/WG1AR5_SPM_FINAL.pdf
- IPCC, 2013b. IPCC, 2013: Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. [Stocker, T. F., Qin, D., Plattner, G-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y., Bex. V. and Midgley, P. M. (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.Available at:

http://www.climatechange2013.org/images/uploads/WGIAR5_Citations_FinalRev1.pdf

- IPCC, 2012: Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., Barros, V., Stocker, T. F., Qin, D., Dokken, D. J., Ebi, K. L., Mastrandrea, M. D., Mach, K. J., Plattner, G-K., Allen, S. K., Tignor, M. & P.M. Midgley, P. M. (eds.)]. Cambridge University Press, Cambridge, UK, and New York, NY, USA, 582 pp.
- IPCC, 2007. Climate change 2007 : impacts, adaptation and vulnerability : Working Group II contribution to the Fourth Assessment Report of the IPCC Intergovernmental Panel on Climate Change. Working Group II Contribution to the Intergovernmental Panel on Climate Change Fourth Assessment Report, 1(July), p.976. Available at:

http://www.ipcc.ch/publications_and_data/publications_ipcc_fourth_assessment_report_w g2_report_impacts_adaptation_and_vulnerability.htm.

- Jaggard, K. W., Qi, A. & Ober, S., 2010. Possible changes to arable crop yields by 2050. *Philos. Trans. R. Soc. B.* 365: 2835-2851.
- Jaggard, K. W., Qi, A. & Ober, E. S., 2009. Capture and use of solar radiation, water, and nitrogen by sugar beet (Beta vulgaris L.). *Journal of Experimental Botany*, 60(7), pp.1919–1925.
- Jaggard, K. W., Qi, A. & Semenov, M. A., 2007. The impact of climate change on sugarbeet yield in the UK: 1976–2004. *The Journal of Agricultural Science*, 145(4), p.367. Available at: http://www.journals.cambridge.org/abstract_S0021859607006922.
- Jenkins, G., Murphy, J., Sexton, D., Lowe, J., Jones, P. & Kilsby, C., 2009. UK Climate Projections: Briefing report. Met Office Hadley Centre, Exeter, UK. Available at: http://ukclimateprojections.metoffice.gov.uk/media.jsp?mediaid=87867
- Jenkins, G. J., Perry, M. C. & Prior, M. J., 2008. The climate of the UK and recent trends. Met Office Hadley Centre, Exeter, UK. Available at: http://ukclimateprojections.defra.gov.uk/images/stories/trends_pdfs/Trends.pdf.
- Jones, P. & Thornton, P., 2003. The potential impacts of climate change on maize production in Africa and Latin America in 2055. *Global Environmental Change*, 13(1), pp.51–59.
- Jones, P. D., Lister, D. H., Jaggard, K. W. & Pidgeon, J. D., 2003. Future climate impact on the productivity of sugar beet (Beta vulgaris L.) in Europe. *Climatic Change*, 58(1–2), pp.93–108.
- Jones, P. D. & Reid, P. A., 2001. Assessing future changes in extreme precipitation over Britain using regional climate model integrations. *International Journal of Climatology*, 21(11), pp.1337–1356.
- Jones, P. D. & Conway, D., 1997. Precipitation in the British Isles: An analysis of area-average data updates to 1995. *International Journal of Climatology*, 17(4), pp.427–438.
- Jones, P. D., Jonsson, T. & Wheeler, D., 1997. Extension to the North Atlantic oscillation using early instrumental pressure observations from Gibraltar and south-west Iceland. *International Journal of Climatology*, 17(13), pp.1433–1450.
- Karl, T. R., Melillo, J. M. & Peterson, T. C., 2009. Global Climate Change Impacts in the United States,
- Karoly, D. J. & Stott, P. A., 2006. Anthropogenic warming of central England temperature. *Atmospheric Science Letters*, 7(4), pp.81–85.

- Kelley, C., Mingfang, T., Seager, R. & Kushnir, Y., 2012. Mediterranean precipitation climatology, seasonal cycle, and trend as simulated by CMIP5. *Geophysical Research Letters*, 39(21).
- Kendon, M., McCarthy, M., Jevrejeva, S. & Legg, T., 2016. State of the UK Climate 2016. Met Office Hadley Centre, National Oceanography Centre, UK. Available at: https://www.metoffice.gov.uk/binaries/content/assets/mohippo/pdf/uk-climate/state-ofthe-uk-climate/mo-state-of-uk-climate-2016-v4.pdf
- Kendon, M. & McCarthy, M., 2015. The UK's wet and stormy winter of 2013/2014. *Weather*, 70(2), pp.40–47.
- Kendon, M., Marsh, T. & Parry, S., 2013. England and Wales drought 2010 to 2012. Weather 68, pp.88–95. Available at: http://www.metoffice.gov.uk/climate/uk/interesting/2012drought [Accessed May 11, 2017].
- Kenter, C., Hoffmann, C. M. & Märländer, B., 2006. Effects of weather variables on sugar beet yield development (Beta vulgaris L.). *European Journal of Agronomy*, 24(1), pp.62–69.
- Kharin, V. V. & Zwiers, F. W., 2005. Estimating extremes in transient climate change simulations. *Journal of Climate*, 18(8), pp.1156–1173.
- Kharin, V. V. & Zwiers, F. W., 2000. Changes in the extremes in an ensemble of transient climate simulations with a coupled atmosphere-ocean GCM. *Journal of Climate*, 13(21), pp.3760–3788.
- Kim, H-Y. 2014. Analysis of variance (ANOVA) comparing means of more than two groups. *Restor Dent Endod.* 39(1): 74-77
- Kim, T. K. 2015. T test as a parametric statistic. *Korean J Anesthesiol*. 68 (6): 540-546. doi:10.4097/kjae.2015.68.6.540
- Kittel, T. G. F., Giorgi, F. & Meehl, G. A., 1997. Intercomparsion of regional biases and doubled CO 2 -sensitivity of coupled atmosphere-ocean general circulation model experiments. *Climate Dynamics*, 14(1), pp.1–15.
- Klein Tank, A. M. G., Zwiers, F. W. & Zhang, X., 2009. Guidelines on Analysis of extremes in a changing climate in support of informed decisions for adaptation. *Climate Data and Monitoring*, (WCDMP-No. 72), p.52.
- Klein Tank, A. & Können, G., 2003. Trends in indices of daily temperature and precipitation extremes in Europe, 1949-99. *Journal of climate* (2003), 16 (22), 3665-3680
- Knutti, R., Furrer, R., Tebalsi, C., Cermak, J. & Meehl, G. A., 2010. Challenges in combining projections from multiple climate models. *Journal of Climate*, 23(10), pp.2739–2758.

- Kopparla, P., Fischer, E. M., Hannay, C. & Knutti, R., 2013. Improved simulation of extreme precipitation in a high-resolution atmosphere model. *Geophysical Research Letters*, 40(21), pp.5803–5808.
- Kosanic, A., Harrison, S., Anderson, K. & Kavcic, I., 2014. Present and historical climate variability in South West England. *Climatic Change*, 124(1–2), pp.221–237.
- Krishnamurti, T. N., Kishtawal, C. M., Zhang, Z., LaRow, T., Bachiochi, D. & Williford, E., 2000. Multimodel Ensemble Forecasts for Weather and Seasonal Climate. *Journal of Climate*, 13(23), pp.4196–4216.
- Kumar, S., Merwade, V., Kinter III, J. L. & Niyogi, D., 2013. Evaluation of temperature and precipitation trends and long-term persistence in CMIP5 twentieth-century climate simulations. *Journal of Climate*, 26(12), pp.4168–4185.
- Lambers, H., Chap-lin, F. S. & Pons, T. L., 1998. *Plant Physiological Ecology*. Springer@ New York, NY.
- Lauer, J. G., 1997. Sugar beet performance and interactions with planting date, genotype, and harvest date. *Agronomy Journal*, 89(3), pp.469–475.
- Lawlor, D. W. & Mitchell, R. A. C., 1991. The Effects of Increasing CO₂ on Crop Photosynthesis and Productivity - a Review of Field Studies. *Plant Cell and Environment*, 14(8), pp.807–818.
- Leblanc, M. J., Tregoning, P., Ramillien, G., Tweed, S. O. & Fakes, A., 2009. Basin-scale, integrated observations of the early 21st century multiyear drought in Southeast Australia. *Water Resources Research*, 45(4), pp.1–10.
- Lebsock, M., Morrison, H. & Gettelman, A., 2013. Microphysical implications of cloudprecipitation covariance derived from satellite remote sensing. *Journal of Geophysical Research Atmospheres*, 118(12), pp.6521–6533.
- Lewis, H., Mittermaier, M., Mylne, K., Norman, K., Scaife, A., Neal, R., Pierce, R., Harrison, D., Jewell, S., Kendon, M. & Saunders, R., 2015. From months to minutes-exploring the value of high-resolution rainfall observation and prediction during the UK winter storms of 2013/2014. *Meteorological Applications*, 22(1), pp.90–104.
- Liu, B., Asseng, S., Christoph Müller, C., Ewert, F., Elliott, J., Lobell, D. B., Martre, P., Ruane, A. C., Wallach, D., Jones, J. W., Rosenzweig, C., Aggarwal, P. K., Alderman, P. D., [...] Zhu, Y., 2016. Similar estimates of temperature impacts on global wheat yield by three independent methods. *Nature Climate Change*, doi: 10.1038/NCLIMATE3115. Available at: http://agri.ckcest.cn/ass/NSTL007-20161205001.pdf

- Lobell, D. B., Schlenker, W. & Costa-Roberts, J., 2011. Climate Trends and Global Crop Production Since 1980. *Science*, 333(6042), pp.616–620. Available at: http://dx.doi.org/10.1126/science.1204531.
- Lobell, D. B., Cassman, K. G. & Field, C. B., 2009. Crop Yield Gaps : Their Importance, Magnitudes, and Causes Crop Yield Gaps : Their Importance, Magnitudes, and Causes.
- Lobell, D. B., Burke, M. B., Tebaldi, C., Mastrandrea, M. D., Falcon, W. P. & Naylor, R. L., 2008. Prioritizing climate change adaptation needs for food security in 2030. *Science*, Vol. 319, Issue 5863, pp. 607-610. doi: 10:1126/science.1152339.
- MAFF, 1999. CAP reform: potential for effects on environmental impact of farming, House of Commons report, Boxworth and Cambs.
- Manning, L. J., Hall, J. W., Fowler, H. J., Kilsby, C. G. & Tebaldi, C., 2009. Using probabilistic climate change information from a multimodel ensemble for water resources assessment. *Water Resources Research*, 45(11), pp.1–13.
- Maraun, D., Wetterhal, F., Ireson, A. M., Chandler, R. E., Kendon, E. J., Widmann, M.,
 Brienen, S., Rust, H. W., Sauter, T., Themebl, M., Venema, V. .K. C., Chun, K. P.,
 Goodess, C. M., Jones, R. G., Onof, C., Vrac, M. & Thiele-Eich, I., 2010. Precipitation
 downscaling under climate change: Recent developments to bridge the gap between
 dynamical models and the end user. *Reviews of Geophysics*, 48(2009RG000314), pp.1–38.
- Maraun, D., Osborn, T. J. & Gillett, N. P., 2008. United Kingdom daily precipitation intensity: Improved early data, error estimates and an update from 2000 to 2006. *International Journal of Climatology*, 28(6), pp.833–842.
- McSweeney, C. F., Jones, R. G., Lee, R. W. & Rowell, D. P., 2014. Selecting CMIP5 GCMs for downscaling over multiple regions. *Climate Dynamics*, 44(11–12), pp.3237–3260. Available at: http://dx.doi.org/10.1007/s00382-014-2418-8.
- Meehl, G. A., Hu, A., Arblaster, J. M., Fasullo, J. & Trenberth, K. E., 2013. Externally forced and internally generated decadal climate variability associated with the interdecadal pacific oscillation. *Journal of Climate*, 26(18), pp.7298–7310.
- Meehl, G. A., Stocker, T. F., Collins, W. D., Friedlingstein, P., Gregory, J. M., Kitoh, A., Knutti, R., Murphy, J. M., Noda, A., Raper, S. C. B., Watterson, I. G., Weaver, A. J. & Zhao, Z. C., 2007. 2007: Global Climate Projections. *Climate Change 2007: Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change*, pp.747–846.
- Meehl, G. A., Arblaster, J. M. & Tebaldi, C., 2005. Understanding future patterns of increased

precipitation intensity in climate model simulations. *Geophysical Research Letters*, 32(18), pp.1–4.

- Meehl, G. A., Kar, T., Easterling, D. R., Changnon, S., Pielke Jr., R., Changnon, D., Evans, J., Groisman, P. Ya., Knutson, T. R., Kunkel, K. E., Mearns, L. o., Parmesan, C., Pulwarty, R., Root, T., Sylves, R. T., Whetton, P. & Zwiers, F., 2000a. An Introduction to Trends in Extreme Weather and Climate Events: Observations, Socioeconomic Impacts, Terrestrial Ecological Impacts, and Model Projections *. *Bulletin of the American Meteorological Society*, 81(3), pp.413–416. Available at: http://journals.ametsoc.org/doi/abs/10.1175/1520-0477%282000%29081%3C0413%3AAITTIE%3E2.3.CO%3B2.
- Meehl, G. A., Boer, G. J., Covey, C., Latif, M. & Stouffer, R. J., 2000b. The Coupled Model Intercomparison Project (CMIP). *Bulletin of the American Meteorological Society*, 81(2), pp.313–318. Available at: http://ams.allenpress.com/perlserv/?issn=1520-0477&issue=02&page=0313&request=get-abstract&volume=081.
- Meinshausen, M., Smith, S. J., Calvin, K., Daniel, J. S., Kainuma, M. L. T., Lamarque, J-F., Matsumoto, K., Montzka, S. A., Raper, S. C. B., Riahi, A., Thomson, A., Velders, G. J .M. & van Vuuren, D. P. P., 2011. The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. *Climatic Change*, 109(1), pp.213–241.
- Menzel, A., Sparks, T. H., Estrella, N., Koch, E., Aaasa, A., Ahas, R., Alm-Kübler, K.,
 Bissolli, P., Braslavská, O., Briede, A., Chmielewski, F. M., Crepinsek, Z., Curnel, Y.,
 Dahl, Å., Defila, C., Donnelly, A., Filella, Y., Jatczak, K., Måge, F., Mestre, A., Nordli,
 Ø., Peñuelas, J., Pirinen, P., Remišová, V., Scheifinger, H., Striz, M., Susnik, A., Van
 Vliet, A. J. H., 2006. European phenological response to climate change matches the
 warming pattern. *Global Change Biology*, 12(10), pp.1969–1976.
- Met Office, 2016. *Eastern England: Climate*, Meteorological Office, Exeter, UK. Available at: www.metoffice.gov.uk/climate/uk/regional-climate/ee.
- Met Office, 2014. Too hot, too cold, too wet, too dry: Drivers and impacts of seasonal weather in the UK., (March), pp.1–36. Available at: http://www.metoffice.gov.uk/media/pdf/4/8/Drivers_and_impacts_of_seasonal_weather_i n_the_UK.pdf.
- Met Ofice, 2012. Met Office Integrated Data Archive System (MIDAS) Land and Marine Surface Stations Data (1853-current). NCAS British Atmospheric Data Centre. Available at: http://catalogue.ceda.ac.uk/uuid/220a65615218d5c9cc9e4785a3234bd0 [Accessed

August 12, 2015].

- Met Office, 2011. Climate: Observations, projections and impacts, United Kingdom. Met Office, UK 2011. [October 2, 2017).
- Met Office (n/d). UK Climate summaries: UK and regional series. Yearly ordered statistics. Met Office Hadley Centre, UK. Available at: https://www.metoffice.gov.uk/pub/data/weather/uk/climate/datasets/Rainfall/date/UK.txt
- [Accessed: 19 Feb. 2018].
- Milford, G. J. F. & Lawlor, D. W., 1976. Water and physiology of sugar beet. Proceedings of the 39th Winter Congress of the International Institute for Sugar Beet Resaerch, Brussels, Belgium.
- Min, S.-K., Zhang, X., Zwiers, F. W. & Hegerl, G. C., 2011. Human contribution to moreintense precipitation extremes. *Nature*, 470(7334), pp.378–381.
- Mishra, A., Hansen, J. W., Dingkuhn, M., Baron, C., Traore, S. B., Ndiaye, O. & Ward, M. N., 2008. Sorghum yield prediction from seasonal rainfall forecasts in Burkina Faso. *Agricultural and Forest Meteorology*, 148(11), pp.1798–1814.
- Mitchell, T. D. & Jones, P. D., 2005. An improved method of constructing a database of monthly climate observations and associated high-resolution grids. *International Journal* of Climatology, 25(6), pp.693–712.
- Mohammadzadeh, A. & Hatamipour, M., 2010. Effect of drying conditions on properties of dried sugar beet. *Iranian journal of chemical engineering*, 7(3). Available at: http://www.sid.ir/en/vewssid/j_pdf/102420100306.pdf.
- Morison, J. & Matthews, R., 2016. Agriculture and Forestry Climate Change Impacts Summary Report, Living With Environmental Change., pp.1–24. Available at: http://www.nerc.ac.uk/research/partnerships/ride/lwec/report-cards/agriculture/.
- Moss, R. H., Edmonds, J. A., Hibbard, K. A., Manning, M. R., Rose, S. K., Van Vuuren, D. P., Carter, T. R., Emori, S., Kainuma, S., Kram, T., Meehl, G. A., Mitchell, J. F. B., Nakicenovic, N., Riahi, K., Smith, S. J., Stouffer, R. J., Thomson, A. M., Weyant, J. P. & Wilbanks, T. J., 2010. The next generation of scenarios for climate change research and assessment. *Nature*, 463(7282), pp.747–56.
- Muchan, K., Lewis, M., Hannaford, J. & Parry, S., 2015. The winter storms of 2013/2014 in the UK: hydrological responses and impacts. *Weather*, 70(2), pp.55–61.
- Myers, S. S., Zanobetti, S., Kloog, I., Huybers, P., Leaky, A. D. B., Bloom, A. j., et al. 2014. Increasing CO₂ threatens human nutrition. Nature 510: 139-142.

- Neher, O. T. & Gallian, J. J., 1947. Powdery Mildew on Sugar Beet Importance, identification, and control., University of Idaho, USA. pp.1–6.
- Nicholls, N., 1997. Increased Australian wheat yield due to recent climate trends. *Nature*, 387(6632), pp.484–485.
- Okom, S., Russell, A., Chaudhary, A. J., Schrimshaw, M. D. & Francis, R. A., 2017. Impacts of projected precipitation changes on sugar beet yield in eastern England. *Meteorological Applications*, 24(1), pp.52–61.
- Osborne, T. M. & Wheeler, T. R., 2013. Evidence for a climate signal in trends of global crop yield variability over the past 50 years. *Environmental Research Letters*, 8(2), p.24001.
- Osborn, T. J. & Hulme, M., 2002. Evidence for trends in heavy rainfall events over the UK. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 360(1796), pp.1313–1325. Available at: http://rsta.royalsocietypublishing.org/cgi/doi/10.1098/rsta.2002.1002.
- Osborn, T. J., Hulme, M., Jones, P. D. & Basnett, T. A., 2000. Observed trends in the daily intensity of United Kingdom precipitation. *International Journal of Climatology*, 20(4), pp.347–364.
- Osborn, T. J. & Hulme, M., 1998. Evaluation of the European daily precipitation characteristics from the atmospheric model intercomparison project. *International Journal* of Climatology, 18(5), pp.505–522.
- Pal, J. S., Giorgi, F. & Bi, X., 2004. Consistency of recent European summer precipitation trends and extremes with future regional climate projections. *Geophysical Research Letters*, 31(13), pp.20–23.
- Pall, P., Allen, M. R. & Stone, D. A., 2007. Testing the Clausius-Clapeyron constraint on changes in extreme precipitation under CO2 warming. *Climate Dynamics*, 28(4), pp.351– 363.
- Palmer, T. N. & Räisänen, J., 2002. Quantifying the risk of extreme seasonal precipitation events in a changing climate. *Nature*, 415(6871), pp.512–514.
- Parry, S., Marsh, T. & Kendon, M., 2013. 2012: From drought to floods in England and Wales. Weather, 68(10), pp.268–274.
- Pendergrass, A. G. & Hartmann, D. L., 2014. Two modes of change of the distribution of rain. *Journal of Climate*, 27(22), pp.8357–8371.
- Peng, S., Huang, J., Sheehy, J. E., Laza, R. C., Visperas, R. M., Zhong, X., Centeno, G. S., Khush, G. S. & Cassman, K. G., 2004. Rice yields decline with higher night temperature

from global warming. *Proceedings of the National Academy of Sciences*, 101(27), pp.9971–9975.

- Perry, M. & Hollis, D., 2005. The development of a new set of long-term climate averages for the UK. *International Journal of Climatology*, 25(8), pp.1023–1039.
- Pervin, L.& Islam, Md. S., 2014. Sytem dynamics approach for modelling of sugar beet yield considering the effects of climatic variables. *Journal of the Science of food and Agriculture*, 95 (3), 515-521.doi: 10.1002/jsfa.6762. Available at: http://onlinelibrary.wiley.com/doi/10.1002/jsfa.6762/full. [Accessed: 30 November 2014].
- Phillips, T. J. & Gleckler, P. J., 2006. Evaluation of continental precipitation in 20th century climate simulations: The utility of multimodel statistics. *Water Resources Research*, 42(3), pp.1–10.
- Pidgeon, J. D., Werker, A. R., Jaggard, K. W., Richter, G. M., Lister, D. H. & Jones, P. D., 2001. Climatic impact on the productivity of sugar beet in Europe, 1961-1995. *Agricultural and Forest Meteorology*, 109(1), pp.27–37.
- Pincus, R., Batstone, C. P., Hofmann, R. J. P., Taylor, K. E. & Glecker, P. J., 2008. Evaluating the present-day simulation of clouds, precipitation, and radiation in climate models. *Journal of Geophysical Research Atmospheres*, 113(14), pp.1–10.
- Poppy, G. M., Jepson, P. C., Pickett, J. A. & Birkett, M. A., 2014. Achieving food and environmental security: new approaches to close the gap. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 369(1639), pp.20120272–20120272.
- Porter, J. R., Liyong, X., Challinor, A. J., Cochrane, K., Howden, S. M., Iqbal, M. M., Lobell, D. B., Travasso, M. I., Netra Chhetri, N. C., Garrett, K., Ingram, J., Lipper, L., McCarthy, J., Smith, D., Thorton, P., Watson, J. & Ziska, L., 2014. Food security and food production systems. In *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*. pp. 485–533.
- Prudhomme, C., Young, A., Watts, G., Haxton, T., Crooks, S., Williamson, J., Davies, H., Dadson, S. & Allen, S., 2012. The drying up of Britain? A national estimate of changes in seasonal river flows from 11 Regional Climate Model simulations. *Hydrological Processes*, 26(7), pp.1115–1118.
- Prudhomme, C., Reynard, N. & Crooks, S., 2002. Downscaling of global climate models for flood frequency analysis: where are we now? *Hydrological Processes*, 16(6, Sp. Iss. SI), pp.1137–1150.

- Qi, A. & Jaggard, K. W., 2008. The impact of past and future climate change on sugar beet yield in the UK. *Aspects of Applied Biology*, (88), pp.153–159.
- Raisanen, J., 2007. How reliable are climate models? *Tellus, Series A: Dynamic Meteorology and Oceanography*, 59(1), pp.2–29.
- Räisänen, J., 2001. CO-2 induced climate change in CMIP2 experiments: Quantification of agreement and role of internal variability. *Journal of Climate*, 14(9), pp.2088–2104.
- Raisanen, J. & Joelsson, R., 2001. Changes in average and extreme precipitation in two regional climate model experiments. *Tellus Series A - Dynamic Meteorology and Oceanography*, 53(5), pp.547–566.
- Randall, D. A., Woods, R. A., Bony, S., Colman, R., Fichefet, T., Fyfe, J., Kattsov, V., Pitman, A., Shukla, J., Srinivasan, J., Stuffer, R. J., Sumi, A. & Taylor, K. E., 2007. Climate Models and Their Evaluation. Climate Change 2007: The Physical Science Basis.
 Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, pp.591–662.
- Ray, D. K., Gerber, J. S., MacDonald, G. K. & West, P. C., 2015. Climate variation explains a third of global crop yield variability. *Nature Communications*, 6, Available at: https://www.nature.com/articles/ncomms6989
- Richter, G. M., Qi, A., Semenov, M. A. & Jaggard, K. W., 2006. Modelling the variability of UK sugar beet yields under climate change and husbandry adaptations. *Soil Use and Management*, 22(1), pp.39–47.
- Rinaldi, M., 2012. Sugar beet . In FAO paper 66 " Crop Yield Crop yield response to water. , (October 2012), pp.2016–2018. Available at: https://www.researchgate.net/publication/258355057_Sugar_beet_In_FAO_paper_66_Crop_Yield_Response_to_Water.
- Rytter, R. M., 2005. Water use efficiency, carbon isotope discrimination and biomass production of two sugar beet varieties under well-watered and dry conditions. *Journal of Agronomy and Crop Science*, 191(6), pp.426–438.
- Santer, B. D., Mears, C., Doutriaux, C., Caldwell, P., Gleckler, P. J., Wigley, T. M. L.,
 Solomon, S., Gillett, N. P., Ivanova, D., Karl, T. R., Lanzante, J. R., Meehl, G. A., Stott,
 P. A., Taylor, K. E., Thorne, P. W., Wehner, M. F. & Wentz, F. J., 2011. Separating signal and noise in atmospheric temperature changes: The importance of timescale. *Journal of Geophysical Research Atmospheres*, 116(22).
- Schmidhuber, J. & Tubiello, F. N., 2007. Global food security under climate change.

Proceedings of the National Academy of Sciences, 104(50), pp.19703–19708.

- Schröter, K., Kunz, M., Elmer, F., Muhr, B. & Merz, B., 2013. What made the June 2013
 Flood in Germany an exceptional event? A hydro-meteorological evaluation. Hydro.Earth
 Syst. Sci., 19, 309-327, 2015. doi: 10.5194/hess-19-309-2015. Available at:
 https://www.hydrol-earth-syst-sci.net/19/309/2015/hess-19-309-2015.html
- Scott, R. K. & Jaggard, K. W., 2000. Impact of weather, agronomy and breeding on yields of sugarbeet grown in the UK since 1970. *The Journal of Agricultural Science*, 134(4), pp.341–352.
- Scott, R. K. & Jaggard, K. W., 1993. Crop physiology and agronomy. In *The Sugar Beet Crop*, Cooke, D. A. & Scott, R. K. (eds). Chapman and Hall: London; 179-237.
- Scott, R. & Jaggard, K., 1978. Theoretical criteria for maximum yield. In *41th IIRB Congress*. Brussels, pp. 179–198.
- Sedgwick P. 2012. Pearson's correlation coefficient. *BMJ* 2012; 345: e4483. doi:https//doi.org/10.1136/bmj.e4483
- Semenov, V. & Bengtsson, L., 2002. Secular trends in daily precipitation characteristics: Greenhouse gas simulation with a coupled AOGCM. *Climate Dynamics*, 19(2), pp.123– 140.
- Seneviratne, S., Nicholls, N., Easterling, D., Goodess, C., Kanae, S., Kossin, J., Luo, Y., Marengo, J., McInnes, K., Rahimi, M., Reichstein, M., Sorteberg, A., Vera, C. & Zhang, X., 2012. Changes in climate extremes and their impacts on the natural physical environment. *Managing the Risk of Extreme Events and Disasters to Advance Climate Change Adaptation.*, pp.109–230.
- Sexton, D. M. H., Parker, D. E. & Folland, C. K., 2004. Natural and human influences on Central England temperature. Hadley Centre technical note 46, Met Office, Bracknell, UK. Available at:

https://www.researchgate.net/profile/Chris_Folland/publication/246922333_Natural_and_ human_influences_on_Central_England_Temperature/links/55b8ab4408ae9289a08db3e1. pdf

- Shrestha, N., Geerts, S., Raes, D., Horemans, S., Soentjens, S., Maupas, F. & Clouet, P., 2010. Yield response of sugar beets to water stress under Western European conditions. *Agricultural Water Management*, 97(2), pp.346–350.
- Siegenthaler, U. & Oeschger, H., 1978. Predicting Future Atmospheric Carbon Dioxide Levels. *Science*, 199(4327), pp.388–395.

- Sillmann, J., Kharin, V. V., Zwiers, F. W., Zhang, X. & Bronaugh, D., 2013. Climate extremes indices in the CMIP5 multimodel ensemble: Part 1. Model evaluation in the present climate. *Journal of Geophysical Research Atmospheres*, 118(4), pp.1716–1733.
- Simpson, I. R. & Jones, P. D., 2012. Updated precipitation series for the UK derived from Met Office gridded data. *International Journal of Climatology*, 32(15), pp.2271–2282.
- Sivakumar, M. V. K., 2006. Climate prediction and agriculture: current status and future challenges. *Climate Research*, 33, pp.3–17.
- Smit, A. L., 1993. The Influence of Sowing Date and Plant-Density on the Decision to Resow Sugar Beet. *Field Crops Research*, 34, pp.159–173.
- Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Avery, K. B., Tignor, M. & Miller, H. L., 2007. Summary for Policymakers. In: Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. *New York Cambridge University Press*, p.996. Available at: http://www.ipcc.ch/publications_and_data/ar4/wg1/en/ch5s5-es.html.
- Sperber, K. R., Annamalai, H., Kang, I-S., Kitoh, A., Moise, A., Turner, A., Wang, B. & Zhou, T., 2013. The Asian summer monsoon: An intercomparison of CMIP5 vs. CMIP3 simulations of the late 20th century. *Climate Dynamics*, 41(9–10), pp.2711–2744.
- Spintali, R. C. 2011. Basic statistical analysis, 9th ed.: Pearson Education: Boston, MA, USA.
- Stocker, T. F., Qin, D., Plattner, G-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y., Bex, V. & Midgley, P. M., (eds. ., 2015. Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. *CEUR Workshop Proceedings*, 1542, pp.33–36.
- Sun, Y., Solomon, S., Dai, A. & Portman, R. W., 2006. How often does it rain? *Journal of Climate*, 19(6), pp.916–934.
- Tao, F., Yokozawa, M., Xu, Y., Hayashi, Y. & Zhang, Z., 2006. Climate changes and trends in phenology and yields of field crops in China, 1981-2000. Agricultural and Forest Meteorology, 138(1–4), pp.82–92.
- Tao, F., Yokozawa, M., Hayashi, Y. & Lin, E., 2003. Changes in agricultural water demands and soil moisture in China over the last half-century and their effects on agricultural production. *Agricultural and Forest Meteorology*, 118(3–4), pp.251–261.
- Taylor, R. G., Scanlon, B.[...] Treidel, H., 2013. Ground water and climate change. *Nature Clim. Change*, 3(4), pp.322–329. Available at: http://dx.doi.org/10.1038/nclimate1744.

- Taylor, K.E., Stouffer, R. J. & Meehl, G. A., 2012. An overview of CMIP5 and the experiment design. *Bulletin of the American Meteorological Society*, 93(4), pp.485–498.
- Tebaldi, C., Arblaster, J. M. & Knutti, R., 2011. Mapping model agreement on future climate projections. *Geophysical Research Letters*, 38(23).
- Tebaldi, C. & Knutti, R., 2007. The use of the multi-model ensemble in probabilistic climate projections. *Philosophical Transactions of the Royal Society A: Mathematical, Physical* and Engineering Sciences, 365(1857), pp.2053–2075.
- Tebaldi, C., Hayhoe, K., Arblaster, J. M. & Meehl, G. A., 2006. Going to the extremes: An intercomparison of model-simulated historical and future changes in extreme events. *Climatic Change*, 79(3–4), pp.185–211.
- The World Bank, 2013. Understanding poverty: Overview. *The World Bank Group*. Available at: http://www.worldbank.org/en/topic/poverty/overview [Accessed June 25, 2017].
- Thornton, P. K., Ericksen, P. J., Herroro, M. & Challinor, A. J., 2014. Climate variability and vulnerability to climate change: A review. *Global Change Biology*, 20(11), pp.3313–3328.
- Thornton, P. K. & Herrero, M., 2010. The Inter-linkages between Rapid Growth in Livestock Production, Climate Change, and the Impacts on Water Resources, Land Use, and Deforestation. *World Development*, (January), p.84.
- Timbal, B., Mahfouf, J-F., Royer, J-F. & Cariolle, D., 1995. Sennsitivity to prescribed changes in seas surface temperature and sea ice in doubled carbon dioxide experiments. *Climate dynamics*, 12(1), pp.1–20.
- Trenberth, K., 2011. Changes in precipitation with climate change. *Climate Research*, 47(1), pp.123–138.
- Trenberth, K. E., Jones, P. D., Ambenje, P., Bojariu, R., Easterling, D. & Klein Tank, A., 2007. Observations: Surface and Atmospheric Climate Change. In: Climate Change 2007: The Physical Science Basis., Cambridge.
- Trenberth, K. E., 1999. Conceptual framework for changes of extremes of the hydrological cycle with climate change. *Climatic Change*, 42(1), pp.327–339.
- Trenberth, K. E. & Josey, S. A., 2007. Observations: surface and atmospheric climate change. *Changes*, 164(236–336), pp.235–336. Available at: http://eprints.soton.ac.uk/50395/.
- Tubiello, F. N. & Ewert, F., 2002. Simulating the effects of elevated CO2 on crops: Approaches and applications for climate change. In *European Journal of Agronomy*. pp. 57–74.
- Turner, A. G. & Annamalai, H., 2012. Climate change and the South Asian summer monsoon.

Nature Climate Change, 2(8), pp.587–595.

- UNFCCC, 2015. United Nations Framework Convention on Climate Change Paris. Background on the UNFCCC: The international response. Paris, France. Available at: http://unfccc.int/essential_background/items/6031.php.
- Urban, D., Roberts, M. J., Schlenker, W. & Lobell, D. B., 2012. Projected temperature changes indicate significant increase in interannual variability of U.S. maize yields: A Letter. *Climatic Change*, 112(2), pp.525–533.
- van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., Hurtt, G. C., Kram, T., Krey, V., Lamarque, J-F., Masui, T., Meinshausen, M. & Nakicenovic, N., 2011. The representative concentration pathways: An overview. *Climatic Change*, 109(1), pp.5–31.
- van Weverberg, K., Vogelmann, A. M., Lin, W., Luke, E. P., Cialella, A., Minnis, P., Khaiyer, M., Boer, E. R. & Jensen, M. P., 2013. The Role of Cloud Microphysics Parameterization in the Simulation of Mesoscale Convective System Clouds and Precipitation in the Tropical Western Pacific. *Journal of the Atmospheric Sciences*, 70(4), pp.1104–1128.
- von Braun, J., 2016. Toward Climate Resilient Food Systems and Europe 's Roles Overview of Issues, Available at:

http://ec.europa.eu/research/conferences/2016/food2030/pdf/03.presentation_joachim_von
_braun_panel2_food2030.pdf.Vermeulen, S. J., Campbell, B. M. & Ingram, J. S. I., 2012.
Climate Change and Food Systems. *Annual Review of Environment and Resources*, 37(1),
pp.195–222. Available at: http://www.annualreviews.org/doi/10.1146/annurev-environ-020411-130608.

- Waha, K., van Bussel, L. G. J., Muller, C. & Bondeau, A., 2012. Climate-driven simulation of global crop sowing dates. *Global Ecology and Biogeography*, 21(2), pp.247–259.
- Wang, X., Tang, J., Niu, X. & Wang, S., 2016. An assessment of precipitation and surface air temperature over China by regional climate models. *Front Earth Sci.*
- Wang, G., 2005. Agricultural drought in a future climate: Results from 15 global climate models participating in the IPCC 4th assessment. *Climate Dynamics*, 25(7–8), pp.739– 753.
- Watson, R.T., 2001. In: CGIAR Annual Report 2000, The Challenge of Climate Change: Poor Farmers at Risk. Consultative Group on International Agricultural Research Secretariat, The World Bank, Washington, DC, pp. 8-12.Watterson, I. G. & Dix, M. R., 2003.
 Simulated changes due to global warming in daily precipitation means and extremes and

their interpretation using the gamma distribution. *Journal of Geophysical Research-Atmospheres*, 108(D13).

- WFP, 2017. Climate change impacts on food security. World Food Programme. Available at: http://www.wfp.org/climate-change/climate-impacts
- Wehner, M. F., Smith, R. L., Bala, G. & Duffy, P., 2010. The effect of horizontal resolution on simulation of very extreme US precipitation events in a global atmosphere model. *Climate Dynamics*, 34(2), pp.241–247.
- Westra, S., Alexander, L. V. & Zwiers, F. W., 2013. Global increasing trends in annual maximum daily precipitation. *Journal of Climate*, 26(11), pp.3904–3918.
- Wheeler, T. & von Braun, J., 2013. Climate change impacts on global food security. *Science* (*New York, N.Y.*), 341(6145), pp.508–13.
- Whetton, P. H., England, M. H., O'Farrell, S. P., Watterson, I. G. & Barrie-Pittock, A., 1996. Global comparison of the regional rainfall results of enhanced greenhouse coupled and mixed layer ocean experiments: Implications for climate change scenario development. *Climatic Change*, 33(4), pp.497–519.
- Wigley, T. & Jones, P., 1987. England and Wales precipitation: A discussion of recent changes in variability and an update to 1985. *Journal of Climatology*, 7, pp.231–246.
- Williams, P. D., Guilyardi, E., Sutton, R., Gregory, J. & Madec, G., 2007. A new feedback on climate change from the hydrological cycle. *Geophysical Research Letters*, 34(8).
- Yonetani, T. & Gordon, H. B., 2001. Simulated changes in the frequency of extremes and regional features of seasonal/annual temperature and precipitation when atmospheric CO2 is doubled. *Journal of Climate*, 14(8), pp.1765–1779.
- Zhang, T. & Huang, Y., 2012. Impacts of climate change and inter-annual variability on cereal crops in China from 1980 to 2008. *Journal of the Science of Food and Agriculture*, 92(8), pp.1643–1652.
- Zhang, X., Zwiers, F. & Li, G., 2004. Monte Carlo experiments on the detection of trends in extreme values. *Journal of climate* (2004), 17 (10), 1945-1952.

Appendix

Appendix 1: Results of the number of watering days in October for each category.

	October	^r Watering Regime					
Historical	RCP45	RCP85		Historical	RCP45	RCP85	
2796	2500	2515	Mean ppt event:	2.730181	2.707843	2.929116	0.23
0.776667	0.694444	0.698611	Surface area:	70685.83	70685.83	70685.83	
24.07667	21.52778	21.65694	Water Vol	192985.1	191406.1	207047	
			Water vol(L)	0.192985	0.191406	0.207047	1.191802
			Scale to 0.23	0.23	0.228118	0.246759	
24.07667	21.52778	21.65694	By # of days	7.13	7.071663	7.64953	
			# of watering days	24	22	22	
			# of no watering days	7	9	9	
	Historical 2796 0.776667 24.07667 24.07667	October Historical RCP45 2796 2500 0.776667 0.694444 24.07667 21.52778 24.07667 21.52778	October Watering Regime Historical RCP45 RCP85 2796 2500 2515 0.776667 0.694444 0.698611 24.07667 21.52778 21.65694 24.07667 21.52778 21.65694	October Watering Regime Historical RCP45 RCP85 2796 2500 2515 Mean ppt event: 0.776667 0.694444 0.698611 Surface area: 24.07667 21.52778 21.65694 Water Vol Vater vol(L) Scale to 0.23 Scale to 0.23 24.07667 21.52778 21.65694 W of watering days	Historical RCP45 RCP85 Historical 2796 2500 2515 Mean ppt event: 2.730181 0.776667 0.694444 0.698611 Surface area: 70685.83 24.07667 21.52778 21.65694 Water Vol 192985.1 Water vol(L) 0.192985 Scale to 0.23 0.23 24.07667 21.52778 21.65694 By # of days 7.13	Historical RCP45 RCP85 Historical RCP45 2796 2500 2515 Mean ppt event: 2.730181 2.707843 0.776667 0.694444 0.698611 Surface area: 70685.83 70685.83 24.07667 21.52778 21.65694 Water Vol 192985.1 191406.1 Water vol(L) 0.192985 0.191406 Scale to 0.23 0.23 0.228118 24.07667 21.52778 21.65694 By # of days 7.13 7.071663	Historical RCP45 RCP85 Historical RCP45 RCP85 2796 2500 2515 Mean ppt event: 2.730181 2.707843 2.929116 0.776667 0.694444 0.698611 Surface area: 70685.83 70685.83 70685.83 70685.83 70685.83 2020147 24.07667 21.52778 21.65694 Water Vol 192985.1 191406.1 207047 24.07667 21.52778 21.65694 Water vol 0.192985 0.191406 0.207047 24.07667 21.52778 21.65694 Water vol 0.192985 0.218118 0.246759 24.07667 21.52778 21.65694 By # of days 7.13 7.071663 7.64953

Appendix 2: Distribution of precipitation events for the historical period (1971-2000).

Historical	Month					
6.927464	October	October	October	October	October	October
0.176093	October	October	October	October	0.176093	October
0.215507	October	October	October	October	October	0.215507
1.082262	October	October	October	October	October	October
4.527269	October	October	October	October	October	October
0.071467	October	0.071467	October	October	October	October
1.395238	October	October	October	October	October	October
2.527432	October	October	October	October	October	October
0.347652	October	October	October	October	October	October

RCP45	Month					
0.105834	October	October	October	0.105834	October	October
0.790525	October	October	October	October	October	October
0.31787	October	October	October	October	October	October
1.07962	October	October	October	October	October	October
2.475367	October	October	October	October	October	October
10.70222	October	October	October	October	October	October
0.184534	October	October	October	October	October	0.184534
1.230237	October	October	October	October	October	October
0.703488	October	October	October	October	October	October

Appendix 3: Distribution of precipitation events for the RCP45scenario (2021-2050).

Appendix 4: Distribution of precipitation events for the RCP85scenario (2021-2050).

RCP85	Month					
0.393498	October	October	October	October	October	October
0.152488	October	October	October	October	0.152488	October
0.295782	October	October	October	October	October	October
1.952672	October	October	October	October	October	October
2.232069	October	October	October	October	October	October
5.446183	October	October	October	October	October	October
0.237385	October	October	October	October	October	October
0.587408	October	October	October	October	October	October
0.818718	October	October	October	October	October	October

Appendix 5: Final watering rates used in the month of October for the historical, RCP45 and RCP85.

October Average Mean Precipitation Event (mm)

0.9055

1.1705

(mm)			
	Surface		Precipitation
Historical (mm)	area	Scale	(L)
0.0635	70685.83	1.191801851	0.0053
0.0938	70685.83	1.191802	0.0079
0.1274	70685.83	1.191802149	0.0107
0.1640	70685.83	1.191802298	0.0138
0.2117	70685.83	1.191802448	0.0178
0.2572	70685.83	1.191802597	0.0217
0.3168	70685.83	1.191802746	0.0267
0.3872	70685.83	1.191802895	0.0326
0.4688	70685.83	1.191803045	0.0395
0.5609	70685.83	1.191803194	0.0472
0.6833	70685.83	1.191803343	0.0576
0.8201	70685.83	1.191803492	0.0691
1.0002	70685.83	1.191803642	0.0843
1.2541	70685.83	1.191803791	0.1056
1.5963	70685.83	1.19180394	0.1345
2.0174	70685.83	1.191804089	0.1700
2.4944	70685.83	1.191804238	0.2101
3.0709	70685.83	1.191804388	0.2587
3.8808	70685.83	1.191804537	0.3269
4.8895	70685.83	1.191804686	0.4119
5.9333	70685.83	1.191804835	0.4998
7.4598	70685.83	1.191804985	0.6284
10.0669	70685.83	1.191805134	0.8481
17.8498	70685.83	1.191805134	1.5037
	Surface		Precipitation
RCP45 (mm)	Area	Scale	(L)
0.0592	70685.83	1.191801851	0.0050
0.0795	70685.83	1.191802	0.0067
0.1065	70685.83	1.191802149	0.0090
0.1342	70685.83	1.191802298	0.0113
0.1698	70685.83	1.191802448	0.0143
0.2223	70685.83	1.191802597	0.0187
0.2794	70685.83	1.191802746	0.0235
0.3584	70685.83	1.191802895	0.0302
0.4656	70685.83	1.191803045	0.0392
0.5817	70685.83	1.191803194	0.0490
0.7217	70685.83	1.191803343	0.0608

70685.83 1.191803492

1.191803642

70685.83

0.0763

0.0986

1.4733	70685.83	1.191803791	0.1241
1.8952	70685.83	1.19180394	0.1597
2.4369	70685.83	1.191804089	0.2053
3.2031	70685.83	1.191804238	0.2698
4.1658	70685.83	1.191804388	0.3509
5.6018	70685.83	1.191804537	0.4719
7.5824	70685.83	1.191804686	0.6388
10.2355	70685.83	1.191804835	0.8623
17.8453	70685.83	1.191804835	1.5034

	Surface		Precipitation
RCP85 (mm)	Area	Scale	(L)
0.0616	70685.83	1.191801851	0.0052
0.0859	70685.83	1.191802	0.0072
0.1116	70685.83	1.191802149	0.0094
0.1455	70685.83	1.191802298	0.0123
0.1889	70685.83	1.191802448	0.0159
0.2367	70685.83	1.191802597	0.0199
0.2943	70685.83	1.191802746	0.0248
0.3680	70685.83	1.191802895	0.0310
0.4625	70685.83	1.191803045	0.0390
0.5977	70685.83	1.191803194	0.0504
0.7387	70685.83	1.191803343	0.0622
0.9280	70685.83	1.191803492	0.0782
1.2089	70685.83	1.191803642	0.1018
1.5544	70685.83	1.191803791	0.1309
2.0428	70685.83	1.19180394	0.1721
2.6565	70685.83	1.191804089	0.2238
3.4051	70685.83	1.191804238	0.2869
4.4483	70685.83	1.191804388	0.3747
5.8789	70685.83	1.191804537	0.4953
7.9063	70685.83	1.191804686	0.6661
11.2292	70685.83	1.191804835	0.9460
19.9690	70685.83	1.191804835	1.6823

`